ZONGULDAK BÜLENT ECEVİT UNIVERSITY GRADUATE SCHOOL OF NATURAL AND APPLIED SCIENCES

DYNAMICAL ANALYSIS OF SOME SYSTEMS OF NONLINEAR DIFFERENCE EQUATIONS

DEPARTMENT OF MATHEMATICS

DOCTOR OF PHILOSOPHY THESIS

İNCİ OKUMUŞ

JULY 2019



ZONGULDAK BÜLENT ECEVİT UNIVERSITY GRADUATE SCHOOL OF NATURAL AND APPLIED SCIENCES

DYNAMICAL ANALYSIS OF SOME SYSTEMS OF NONLINEAR DIFFERENCE EQUATIONS

DEPARTMENT OF MATHEMATICS

DOCTOR OF PHILOSOPHY THESIS

İnci OKUMUŞ

ADVISOR: Prof. Yüksel SOYKAN

ZONGULDAK July 2019



APPROVAL OF THE THESIS:

The thesis entitled "Dynamical Analysis of Some Systems of Nonlinear Difference Equations" and submitted by Inci OKUMUŞ has been examined and accepted by the jury as a Doctor of Philosophy thesis in Department of Mathematics, Graduate School of Natural and Applied Sciences, Zonguldak Bülent Ecevit University. 05/07/2019

Advisor:	Prof. Yüksel SOYKAN Zonguldak Bülent Ecevit University, Faculty of Arts and Science, Department of Mathematics
Member:	Prof. Yusuf KAYA Zonguldak Bülent Ecevit University, Faculty of Arts and Science, Department of Mathematics
Member:	Prof. Esra ERKUŞ DUMAN Gazi University, Faculty of Science, Department of Mathematics
Member:	Assist. Prof. Mustafa AKINCI
Member:	Assist. Prof. Nejla ÖZMEN Düzce University, Faculty of Arts and Science, Department of Mathematics

Approved by the Graduate School of Natural and Applied Sciences.

..../..../2019

Prof. Ahmet ÖZARSLAN Director





"With this thesis it is declared that all the information in this thesis is obtained and presented according to academic rules and ethical principles. Also as required by the academic rules and ethical principles all works that are not result of this study are cited properly."

İnci OKUMUŞ



ABSTRACT

Doctor of Philosophy Thesis

DYNAMICAL ANALYSIS OF SOME SYSTEMS OF NONLINEAR DIFFERENCE EQUATIONS

İnci OKUMUŞ

Zonguldak Bülent Ecevit University Graduate School of Natural and Applied Sciences Department of Mathematics

Thesis Advisor: Prof. Yüksel SOYKAN July 2019, 143 pages

In this thesis, we present a systematic study of dynamical behavior of solutions of some specific non-linear difference equations and systems of difference equations. Especially, we investigate the stability character of equilibrium points, the exact forms, the periodicity, the oscillation and the boundedness of solutions of these equations and systems.

The organization of this thesis is as follows:

Chapter 1 is a concise overview of what this thesis is about and also is a literature summary of difference equation theory.

Chapter 2 consists of some basic important definitions and some significant theorems used throughout the thesis.

Chapter 3 includes some results about the stability, boundedness character and periodic nature of positive solutions of the system of difference equations

ABSTRACT (continued)

 $x_{n+1} = A + x_{n-1} / z_n$, $y_{n+1} = A + y_{n-1} / z_n$, $z_{n+1} = A + z_{n-1} / y_n$, for n = 0, 1, ... where the parameter *A* and the initial conditions $x_{-1}, x_0, y_{-1}, y_0, z_{-1}, z_0$ are arbitrary positive real numbers.

Chapter 4 presents the local asymptotic stability of the equilibrium points, boundedness character, oscillatory, and global asymptotic behavior of positive solutions of the system of rational difference equations $x_{n+1} = A + x_{n-m} / z_n$, $y_{n+1} = A + y_{n-m} / z_n$, $z_{n+1} = A + z_{n-m} / y_n$, for n = 0,1,... where the parameter A and the initial values x_{-i}, y_{-i}, z_{-i} , for i = 0,1,...,m are positive real numbers and *m* is positive integer.

Chapter 5 contains some results about the local asymptotic stability of the equilibrium points and oscillation behaviour of positive solutions of the following system of rational difference equations $x_{n+1} = A + x_{n-1}^p / z_n^p$, $y_{n+1} = A + y_{n-1}^p / z_n^p$, $z_{n+1} = A + z_{n-1}^p / y_n^p$, for n = 0,1,... where the parameters $A \in (0,\infty)$, $p \in [1,\infty)$ and the initial values $x_{-i}, y_{-i}, z_{-i} \in (0,\infty)$, i = -1,0.

Chapter 6 states the stability character of equilibrium points and the form of solutions and asymptotic behavior of positive solutions of the following four rational difference equations $x_{n+1} = 1/(x_n(x_{n-1}\pm 1)\pm 1), x_{n+1} = -1/(x_n(x_{n-1}\pm 1)\mp 1))$, such that their solutions are associated with Tribonacci numbers.

Chapter 7 acquaints about the stability character of equilibrium points, the periodic nature of solutions and the global behavior of solutions of the following four rational difference equations $x_{n+1} = \pm 1/(x_n(x_{n-1}\pm 1)-1), x_{n+1} = \pm 1/(x_n(x_{n-1}\mp 1)+1).$

Chapter 8 presents some results about the stability character of equilibrium points of and the explicit form and global behavior of positive solutions of the following two systems of rational difference equations $x_{n+1} = \pm 1/(y_n(x_{n-1}\pm 1)+1)$, $y_{n+1} = \pm 1/(x_n(y_{n-1}\pm 1)+1)$, such that their solutions are associated with Tribonacci numbers.

ABSTRACT (continued)

Chapter 9 expresses the stability character of equilibrium points of and the explicit form and asymptotic behavior of solutions of the following nonlinear difference equation $x_{n+1} = \gamma / (x_n (x_{n-1} + \alpha) + \beta)$, such that their solutions are associated with generalized Tribonacci numbers.

Keywords: Difference equation, equilibrium point, asymptotic behaviour, global asymptotic stability, oscillation, periodicity, unbounded solutions, boundedness, recursive sequence, Tribonacci numbers.

Science Code: 403.06.01



ÖZET

Doktora Tezi

LİNEER OLMAYAN BAZI FARK DENKLEM SİSTEMLERİNİN DİNAMİK ANALİZİ

İnci OKUMUŞ

Zonguldak Bülent Ecevit Üniversitesi Fen Bilimleri Enstitüsü Matematik Anabilim Dalı

Tez Danışmanı: Prof. Dr. Yüksel SOYKAN Temmuz 2019, 143 sayfa

Bu tezde, bazı özel lineer olmayan fark denklemlerinin ve fark denklem sistemlerinin çözümlerinin dinamik davranışlarının sistematik bir çalışması verilmiştir. Özellikle, bu fark denklemlerinin ve fark denklem sistemlerinin çözümlerinin sınırlılığı, salınımlılığı, periyodikliği, kesin çözüm formları ve denge noktalarının kararlılık karakteri araştırılmıştır.

Bu tezin organizasyonu aşağıdaki gibidir:

Bölüm 1, bu tezin ne ile ilgili olduğuna dair kısa bir tanıtım ve ayrıca fark denklem teorisinin literatür özetidir.

Bölüm 2, tez boyunca kullanılan bazı temel önemli tanımlardan ve teoremlerden oluşmaktadır.

ÖZET (devam ediyor)

Bölüm 3, *A* parametresi ve $x_{-1}, x_0, y_{-1}, y_0, z_{-1}, z_0$ başlangıç koşulları keyfi pozitif gerçel sayılar olmak üzere n = 0, 1, ... için $x_{n+1} = A + x_{n-1} / z_n$, $y_{n+1} = A + y_{n-1} / z_n$, $z_{n+1} = A + z_{n-1} / y_n$ fark denklem sisteminin pozitif çözümlerinin kararlılığı, sınırlılık karakteri ve periyodik niteliği hakkında bazı sonuçlar içerir.

Bölüm 4, *m* pozitif tamsayı ve *A* parametresi ve i = 0, 1, ..., m için x_{-i}, y_{-i}, z_{-i} , başlangıç koşulları pozitif gerçel sayılar olmak üzere $x_{n+1} = A + x_{n-m} / z_n$, $y_{n+1} = A + y_{n-m} / z_n$, $z_{n+1} = A + z_{n-m} / y_n$ rasyonel fark denklem sisteminin pozitif çözümlerinin sınırlılık karakterini, salınımlığını ve global asimptotik davranışını ve denge noktalarının yerel (lokal) asimptotik kararlılığını verir.

Bölüm 5, $A \in (0,\infty)$, $p \in [1,\infty)$ ve $x_{-i}, y_{-i}, z_{-i} \in (0,\infty)$, i = -1,0 olmak üzere n = 0,1,... için $x_{n+1} = A + x_{n-1}^p / z_n^p$, $y_{n+1} = A + y_{n-1}^p / z_n^p$, $z_{n+1} = A + z_{n-1}^p / y_n^p$, rasyonel fark denklem sisteminin pozitif çözümlerinin salınım davranışı ve denge noktalarının lokal asimptotik kararlılığı hakkında bazı sonuçlar içermektedir.

Bölüm 6, çözümleri Tribonacci sayılarıyla ilişkili olan aşağıdaki $x_{n+1} = 1/(x_n(x_{n-1}\pm 1)\pm 1)$, $x_{n+1} = -1/(x_n(x_{n-1}\pm 1)\mp 1)$, dört rasyonel fark denkleminin pozitif çözümlerinin kesin çözüm formlarını ve asimptotik davranışlarını ve denge noktalarının kararlılık karakterlerini ifade eder.

Bölüm 7, aşağıdaki $x_{n+1} = \pm 1/(x_n(x_{n-1}\pm 1)-1), x_{n+1} = \pm 1/(x_n(x_{n-1}\mp 1)+1)$ dört rasyonel fark denkleminin çözümlerinin periyodik doğası ve global davranışları ve denge noktalarının kararlılık yapıları hakkında bilgi verir.

Bölüm 8, çözümleri Tribonacci sayılarıyla ilişkili olan aşağıdaki $x_{n+1} = \pm 1/(y_n(x_{n-1}\pm 1)+1)$, $y_{n+1} = \pm 1/(x_n(y_{n-1}\pm 1)+1)$, iki rasyonel fark denklem sisteminin pozitif çözümlerinin kesin (açık) çözüm formları ve global davranışları ve denge noktalarının kararlılık karakterleri hakkında bazı sonuçlar sunmaktadır.

ÖZET (devam ediyor)

Bölüm 9, çözümleri genelleştirilmiş Tribonacci sayılarıyla ilişkili olan aşağıdaki $x_{n+1} = \gamma / (x_n(x_{n-1} + \alpha) + \beta)$, lineer olmayan fark denkleminin çözümlerinin asimptotik davranışlarını ve kesin çözüm formlarını ve denge noktalarının kararlılık karakterlerini ifade eder.

Anahtar Kelimeler: Fark denklem, denge noktası, asimptotik davranış, global asimptotik kararlı, salınımlılık, periyodiklik, sınırsız çözümler, sınırlılık, tekrarlı dizi, Tribonacci sayıları.

Bilim Kodu: 403.06.01



ACKNOWLEDGEMENTS

I would like to express the deepest appreciation to my advisor Prof. Yüksel Soykan who has the attitude and substance of a genius: he continually and convincingly conveyed a spirit of adventure in regard to research and scholarship, and an excitement in regard to teaching. I would like to thank to him for supporting me during my doctorate process.

Besides my advisor, I would like to thank to Prof. Yusuf Kaya and Assist. Prof. Mustafa Akıncı for their helping, suggestions and helpful career advice in general. The good advice, support and friendship of them has been invaluable on both an academic and a personal level for which I am extremely grateful.

I also appreciate to Assist. Prof. Melih Göcen, Assist. Prof. Can Murat Dikmen, Assist. Prof. Mehmet Gümüş and Assist. Prof. Erkan Taşdemir to give their support both in my studies and in my daily life adoptively.

Besides them, I thank to all my relatives, especially my aunt Elif Ümsür, for their supports during my graduate education.

Last, but by no means least, I would like to thank my parents Sadullah Okumuş and Vildane Okumuş who have a big role in my success for giving birth to me at the first place and supporting me spiritually throughout my life. Furthermore, I would like to thank to my sisters Demet Okumuş, Tuğba Okumuş, and Kübra Okumuş and my brother Metin Okumuş for their support and encouragement throughout. I owe them everything and wish I could show them just how much I love and appreciate them.



TABLE OF CONTENTS

	Page
APPROVAL OF THE THESIS	ii
ABSTRACT	iii
ÖZET	vii
ACKNOWLEDGMENTS	xi
TABLE OF CONTENTS	xiii
LIST OF TABLES	xvii
CHAPTER 1 INTRODUCTION AND LITERATURE REVIEWS	1
1.1 LITERATURE REVIEW FOR SYSTEMS OF NONLINEAR DIFFERENCE EQUATION	ONS3
1.1.1 Review On Two-Dimensional Systems Of Nonlinear Difference Equations	4
1.1.2 Review On Three-Dimensional Systems Of Nonlinear Difference Equations	19
1.1.3 Review On Multi-Dimensional Systems Of Nonlinear Difference Equations	20
1.2 LITERATURE SUMMARY FOR DISCRETE DYNAMICAL SYSTEMS	21
CHAPTER 2 THE GENERAL DEFINITIONS AND THEOREMS	
2.1 DIFFERENCE EQUATIONS	31
2.2 INTEGER SEQUENCES	
2.2.1 Fibonacci Numbers	
2.2.2 Padovan Numbers	40
2.2.3 Horadam Numbers	40
2.2.4 Generalized Padovan Numbers	41
2.2.5 Tribonacci Numbers	42
2.2.6 Generalized Tribonacci Numbers	43
CHAPTER 3 GLOBAL BEHAVIOR OF SOLUTIONS OF A SYSTEM OF THREE-	
DIMENSIONAL NONLINEAR DIFFERENCE EQUATIONS	45
3.1 MAIN RESULTS	45

TABLE OF CONTENTS (continued)

Page

CHAPTER 4 GLOBAL ANALYSIS OF A SYSTEM OF HIGHER ORDER NONLINEAR	
DIFFERENCE EQUATIONS	57
4.1 MAIN RESULTS	57
CHAPTER 5 THE NATURE OF SOLUTIONS OF A SYSTEM OF SECOND ORDER	
DIFFERENCE EQUATION	71
5.1 MAIN RESULTS	71
CHAPTER 6 THE EXACT SOLUTIONS OF FOUR RATIONAL DIFFERENCE EQUATION	S
ASSOCIATED TO TRIBONACCI NUMBERS	77
6.1 MAIN RESULTS	78
6.1.1 The Difference Equation (6.1)	78
6.1.2 The Difference Equation (6.2)	81
6.1.3 The Difference Equation (6.3)	85
6.1.4 The Difference Equation (6.4)	88
CHAPTER 7 THE DYNAMICS OF EXACT SOLUTIONS OF SECOND ORDER NONLINE	AR
FOUR RATIONAL DIFFERENCE EQUATIONS	
7.1 MAIN RESULTS	93
7.1.1 The Difference Equation (7.1)	93
7.1.2 The Difference Equation (7.2)	97
7.1.3 The Difference Equation (7.3)	101
7.1.4 The Difference Equation (7.4)	103
CHAPTER 8 GLOBAL ANALYSIS OF SOLUTIONS OF SYSTEMS OF DIFFERENCE	
EQUATIONS VIA TRIBONACCI NUMBERS	107
8.1 MAIN RESULTS	107
8.1.1 The System (8.1)	107
8.1.2 The System (8.2)	113

TABLE OF CONTENTS (continued)

Page

CHAPTER 9 THE DYNAMICS OF SOLUTIONS OF A RATIONAL DIFFER	ENCE EQUATION
VIA GENERALIZED TRIBONACCI NUMBERS	119
9.1 INTRODUCTION	
9.2 MAIN RESULTS	
REFERENCES.	
CURRICULUM VITAE	



LIST OF TABLES

No	Page
Table 2.1 A few values of generalized Tribonacci sequences.	43





CURRICULUM VITAE

PERSONAL INFORMATION

Name, Surname: İNCİ OKUMUŞ Nationality: Turkish (T.C.) Date of Birth: 08.01.1990 Date of Place: GEBZE E-mail: <u>inci_okumus_90@hotmail.com</u>

Address: Kazım Karabekir Mah. Vefa Sok. Okumuşlar Apt. No: 6/6 Darıca/ KOCAELİ Tel: (0 535) 613 73 85

EDUCATION

Degree	Institution	Year of Graduation
MSc	Bülent Ecevit University, Mathematics	2012-2014
BS	Bülent Ecevit University, Mathematics	2008-2012
High School	Darıca Yabancı Dil Ağırlıklı Lisesi	2003-2007

FOREIGN LANGUAGE

English

FIELD OF STUDY

Major Field: Difference and Functional Equations: (Stability and Asymptotics of Difference Equations; Oscillatory and Periodic Solutions, etc.)

CHAPTER 1

INTRODUCTION AND LITERATURE REVIEWS

Difference equation or discrete dynamical system is a diverse field which impact almost every branch of pure and applied mathematics. Every dynamical system

 $x_{n+1} = f(x_n, x_{n-1}, \dots, x_{n-k})$

determines a difference equation and vice versa.

Recently, there has been great interest in studying the difference equations and systems of difference equations. One of the reasons for this is a necessity for some techniques which can be used in investigating equations arising in mathematical models describing real life situations in population biology, economics, probability theory, genetics, psychology and so forth.

The theory of difference equations occupies a central position in applicable analysis. There is no doubt that the theory of difference equations will continue to play an important role in mathematics as a whole. Nonlinear difference equations of order greater than one are of paramount importance in applications. Such equations also appear naturally as discrete analogues and as numerical solutions of differential and delay differential equations which model various diverse phenomena in biology, ecology, psychology, physics, engineering, and economics (see [1-14]). It is very interesting to investigate the behavior of solutions of a system of nonlinear difference equations and to discuss the local asymptotic stability of their equilibrium points. Though difference equations are very simple in their form, it is quite hard to understand throughly the global behavior of their solutions. There are many papers in which systems of difference equations have been studied, see [29-42].

Moreover, there has been a growing interest in the study of finding closed-form solutions of difference equations and systems of difference equations. Some of the forms of solutions of these equations are representable via well-known integer sequences such as Fibonacci numbers, Lucas numbers, Pell numbers and Padovan numbers, see also [74-100].

The purpose in this thesis is to investigate a systematic study of dynamical behavior of so-

lutions of some specific nonlinear difference equations and systems of difference equations. Especially, we research the exact forms, periodicity, stability character and boundedness of solutions of difference equations and systems of nonlinear difference equations. This thesis consist of nine chapters.

The first chapter is a concise overview of what this thesis is about and also is a literature summary of difference equation theory.

The second chapter consists of some basic important definitions and some significant theorems used throughout the thesis.

The third chapter includes some results about the stability, boundedness character and periodicity of positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{x_{n-1}}{z_n}, \quad y_{n+1} = A + \frac{y_{n-1}}{z_n}, \quad z_{n+1} = A + \frac{z_{n-1}}{y_n}, \quad n = 0, 1, \dots,$$

where the parameter A and the initial conditions

$$x_{-1}, x_0, y_{-1}, y_0, z_{-1}, z_0$$

are arbitrary positive real numbers.

The fourth chapter presents the local asymptotic stability of the equilibrium points, boundedness character, oscillatory, and global asymptotic behavior of solutions of the system of difference equations

$$x_{n+1} = A + \frac{x_{n-m}}{z_n}, \ y_{n+1} = A + \frac{y_{n-m}}{z_n}, \ z_{n+1} = A + \frac{z_{n-m}}{y_n}, \ n = 0, 1, \dots,$$

where the parameter A and the initial values

$$x_{-i}, y_{-i}, z_{-i}, \text{ for } i = 0, 1, ..., m,$$

are positive real numbers and m is positive integer.

The fifth chapter contains some results about the local asymptotic stability of the equilibrium points and oscillation behaviour of positive solutions of the following system of rational difference equations

$$x_{n+1} = A + \frac{x_{n-1}^p}{z_n^p}, \ y_{n+1} = A + \frac{y_{n-1}^p}{z_n^p}, \ z_{n+1} = A + \frac{z_{n-1}^p}{y_n^p}, \ n = 0, 1, \dots$$

where the parameters

$$A \in (0,\infty), \quad p \in [1,\infty)$$

and the initial values

$$x_i, y_i, z_i \in (0, \infty), \quad i = -1, 0.$$

The sixth chapter introduces the form of solutions, stability character and asymptotic behavior of the following four rational difference equations

$$x_{n+1} = \frac{1}{x_n (x_{n-1} \pm 1) \pm 1},$$

$$x_{n+1} = \frac{-1}{x_n (x_{n-1} \pm 1) \mp 1},$$

such that their solutions are associated with Tribonacci numbers.

The seventh chapter acquaints about the stability character, the periodicity and the global behavior of solutions of the following four rational difference equations

$$x_{n+1} = \frac{\pm 1}{x_n (x_{n-1} \pm 1) - 1}$$

$$x_{n+1} = \frac{\pm 1}{x_n (x_{n-1} \mp 1) + 1}.$$

The eighth chapter presents some results about the explicit form, stability character and global behavior of solutions of the following two systems of rational difference equations

$$x_{n+1} = \frac{\pm 1}{y_n \left(x_{n-1} \pm 1\right) + 1}, \quad y_{n+1} = \frac{\pm 1}{x_n \left(y_{n-1} \pm 1\right) + 1}, \quad n = 0, 1, \dots$$

such that their solutions are associated with Tribonacci numbers.

The ninth chapter expresses the dynamical behavior of solutions of the following nonlinear difference equation

$$x_{n+1} = \frac{\gamma}{x_n (x_{n-1} + \alpha) + \beta}, \quad n = 0, 1, ...,$$

such that their solutions are associated with generalized Tribonacci numbers.

The following two sections are important summaries about difference equations and systems of difference equations which have shed light on our studies in this thesis.

1.1 LITERATURE REVIEW FOR SYSTEMS OF NONLINEAR DIFFER-ENCE EQUATIONS

In this section, we have divided the studies which we examine in literature into three subsections as two-dimensional, three-dimensional and multi-dimensional systems of nonlinear difference equations.

1.1.1 Review on Two-Dimensional Systems of Nonlinear Difference Equations

This subsection is concerned with the review of dynamical behavior of solutions of the systems of two-dimensional nonlinear difference equations. Then, we have classified these studies into three subsubsections as systems of rational-type difference equations, systems of exponential-type difference equations and systems of max-type difference equations.

Systems of Rational-Type Difference Equations

In [15], Papaschinopoulos and Schinas considered the system of difference equations

$$x_{n+1} = A + \frac{y_n}{x_{n-p}}, \quad y_{n+1} = A + \frac{x_n}{y_{n-q}}, \quad n = 0, 1, ...,$$
 (1.1)

where $A \in (0, \infty)$, p, q are positive integers and $x_{-p}, ..., x_0, y_{-q}, ..., y_0$ are positive numbers. They investigated the oscillatory behavior, the boundedness of the solutions, and the global asymptotic stability of the positive equilibrium of the system (1.1). As a result, they prove that:

- Every positive nontrivial solution $\{(x_n, y_n)\}$ of system (1.1) oscillates about the positive equilibrium of system (1.1).
- If A > 0 and one at least of p, q is an odd number (resp. A > 1 and p, q are both even numbers), then any positive solution of system (1.1) is bounded away from zero and infinity.
- If A > 1, then the positive equilibrium (c, c) of system (1.1) is globally asymptotically stable.

In [16], Papaschinopoulos and Schinas studied the oscillatory behavior, the periodicity and the asymptotic behavior of the positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{x_{n-1}}{y_n}, \quad y_{n+1} = A + \frac{y_{n-1}}{x_n}, \quad n = 0, 1, \dots,$$
 (1.2)

where A is a positive constants and initial conditions are positive numbers.

They established conditions so that a positive solution (x_n, y_n) of system (1.2) oscillates about positive equilibrium of the system (1.2). Moreover, they found

- For the case $0 \le A < 1$,
 - The unique positive equilibrium (c, c) of (1.2) is not stable.
 - The system (1.2) has unbounded solutions.
- For the case A = 1,
 - For every $\mu \in (1, \infty)$, there exist positive solutions (x_n, y_n) of system (1.2) which tend to the positive equilibrium $\left(\mu, \frac{\mu}{\mu-1}\right)$.
 - Every positive solution of system (1.2) tends to a period 2 solution as $n \to \infty$.
- For the case A > 1,
 - The unique positive equilibrium (c, c) of (1.2) is locally asymptotically stable.
 - The positive equilibrium (c, c) of system (1.2) is globally asymptotically stable.

In [17], Camouzis and Papaschinopoulos studied the boundedness, persistence, and the global asymptotic behavior of the positive solutions of the system of difference equations

$$x_{n+1} = 1 + \frac{x_n}{y_{n-m}}, \quad y_{n+1} = 1 + \frac{y_n}{x_{n-m}}, \quad n = 0, 1, ...,$$
 (1.3)

where $x_i, y_i, i = -m, -m + 1, ..., 0$ are positive numbers and m is a positive integer. Then the following results were exhibited in their paper:

- Every positive solution of system (1.3) is bounded and persists,
- System (1.3) has an infinite number of positive equilibrium solutions (x, y) with x, $y \in (1, \infty)$ that satisfy equation xy = x + y,
- Every positive solutions of system (1.3) converges to a positive equilibrium solution of system (1.3) as $n \to \infty$.

In [18], Yang studied the behavior of positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{y_{n-1}}{x_{n-p}y_{n-q}}, \quad y_{n+1} = A + \frac{x_{n-1}}{x_{n-r}y_{n-s}}, \quad n = 1, 2, \dots,$$
(1.4)

where $p \ge 2$, $q \ge 2$, $r \ge 2$, $s \ge 2$, A is a positive constant, and $x_{1-\max\{p,r\}}$, $x_{2-\max\{p,r\}}$, ..., x_0 , $y_{1-\max\{q,s\}}$, $y_{2-\max\{q,s\}}$, ..., y_0 are positive real numbers. He demonstrated that:

• The system (1.4) has the unique positive equilibrium

$$(c,c) = \left(\frac{A + \sqrt{A^2 + 4}}{2}, \frac{A + \sqrt{A^2 + 4}}{2}\right),$$

- When A > 1, every positive solution of system (1.4) is bounded,
- When $A > 2/\sqrt{3}$, (c, c) is locally asymptotically stable,
- When $A > \sqrt{2}$, every positive solution of system (1.4) approaches (c, c),
- When $A > \sqrt{2}$, the positive equilibrium (c, c) of (1.4) is globally asymptotically stable for all positive solutions.

In [19], Zhang et al. considered the behavior of positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{1}{y_{n-p}}, \quad y_{n+1} = A + \frac{1}{x_{n-r}y_{n-s}}, \quad n = 1, 2, ...,$$
 (1.5)

where $p \ge 1$, $r \ge 1$, $s \ge 1$, $A \ge 0$, and x_{1-r} , x_{2-r} , ..., x_0 , $y_{1-\max\{p,s\}}$, $y_{2-\max\{p,s\}}$, ..., y_0 are positive real numbers.

They obtained the following results:

- If A > 0, every positive solution of system (1.5) is bounded,
- If A = 0, all positive solutions of system (1.5) are periodic,
- If $A > 2/\sqrt{3}$ and $\max\{p, r, s\} \ge 2$, the positive equilibrium (c, c) of (1.5) is locally asymptotically stable where $(c, c) = \left(\frac{A+\sqrt{A^2+4}}{2}, \frac{A+\sqrt{A^2+4}}{2}\right)$,
- If $A > \sqrt{2}$, every positive solution of system (1.5) approaches (c, c),
- If $A > \sqrt{2}$ and $\max\{p, r, s\} \ge 2$, the positive equilibrium (c, c) of (1.5) is globally asymptotically stable for all positive solutions.

In [20], Zhang et al. studied the system of rational difference equations

$$x_{n+1} = A + \frac{y_{n-m}}{x_n}, \quad y_{n+1} = A + \frac{x_{n-m}}{y_n}, \quad n = 0, 1, \dots$$
 (1.6)

They investigated the dynamic behavior of positive solutions of system (1.6) for the cases of A < 1, A = 1, and A > 1.

For the case A < 1, they obtained that the system (1.6) has unbounded solutions.

For the case A = 1, they proved that every positive solution of the system (1.6) is bounded and persists with interval $\left[L, \frac{L}{L-1}\right]$ and has prime two periodic solutions.

For the case A > 1, the global asymptotic stability of the unique equilibrium point of the system (1.6) is established. For this case, they proved that:

- Every positive solution of the system (1.6) is bounded and persists by interval $\left[L, \frac{L}{L-A}\right]$,
- The positive equilibrium point (c, c) of system (1.6) is locally asymptotically stable where c = A + 1,
- Every positive solution of system (1.6) converges to (c, c).

In [21], Zhang et al. considered the behavior of the symmetrical system of rational difference equations

$$x_{n+1} = A + \frac{y_{n-k}}{y_n}, \quad y_{n+1} = A + \frac{x_{n-k}}{x_n}, \quad n = 0, 1, \dots$$
 (1.7)

where A > 0 and $x_i, y_i \in (0, \infty)$, for i = -k, -k + 1, ..., 0.

They investigated the dynamic behavior of positive solutions of system (1.7) for the cases of 0 < A < 1, A = 1, and A > 1.

In the case 0 < A < 1, they obtained similar results as in above Theorem 1 for k is odd. However, they said that they <u>can't</u> get some useful results for k is even.

In the case A = 1, the results which are obtained are similar to results in [20].

In the case A > 1, the following results were established:

- Every positive solution of the system (1.7) is bounded and persists by interval $[L, \frac{L}{L-A}],$
- Every positive solution of the system (1.7) converges to the equilibrium as $n \to \infty$.

In [22], Kurbanli, Çinar and Yalcinkaya studied the behavior of the positive solutions of the system of difference equations

$$x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} + 1}, \quad y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} + 1},$$
(1.8)

where the initial conditions are arbitrary non-negative real numbers.

They found the equilibrium point and all solutions of the system (1.8). Also, they obtained the followings where $y_0 = a$, $y_{-1} = b$, $x_0 = c$ and $x_{-1} = d$ are arbitrary non-negative real numbers:

- If $b \neq 0$ and c = 0, $x_{2n} = 0$ and $y_{2n-1} = b$,
- If b = 0 and $c \neq 0$, $x_{2n} = c$ and $y_{2n-1} = 0$,
- If d = 0 and $a \neq 0$, $y_{2n} = a$ and $x_{2n-1} = 0$,
- If a = 0 and $d \neq 0$, $y_{2n} = 0$ and $x_{2n-1} = d$.

In [23], Kurbanli et al. investigated the periodicity of the solutions of the system of difference equations

$$x_{n+1} = \frac{x_{n-1} + y_n}{y_n x_{n-1} - 1}, \quad y_{n+1} = \frac{y_{n-1} + x_n}{x_n y_{n-1} - 1},$$
(1.9)

where $x_0, x_{-1}, y_0, y_{-1} \in \mathbb{R}$.

They proved that the solutions of x_n and y_n are six periodic under the special conditions. In [24], Wang, Zhang and Fu considered the system of difference equations

$$x_{n+1} = \frac{x_{n-2k+1}}{Ay_{n-k+1}x_{n-2k+1} + \alpha}, \quad y_{n+1} = \frac{y_{n-2k+1}}{Bx_{n-k+1}y_{n-2k+1} + \beta}, \quad n \ge 0,$$
(1.10)

where k is a positive integer, A, B, α , β and the initial conditions are positive real numbers.

Under the specific conditions, they established the convergence of the positive solutions of the system (1.10) and showed that the system (1.10) has unbounded solutions.

In [25], Zhang et al. corcerned with the dynamical behavior of positive solutions of the system of two rational difference equations

$$x_{n+1} = A + \frac{x_n}{\sum_{i=1}^k y_{n-i}}, \quad y_{n+1} = B + \frac{y_n}{\sum_{i=1}^k x_{n-i}}, \quad n = 0, 1, ...,$$
(1.11)

where A, B are positive constants and the initial conditions $x_{-i}, y_{-i} \in (0, \infty), i = 0, 1, ..., k$.

They proved that under the case A > 1/k, B > 1/k and assuming that

$$\frac{k^2AB - 1}{kA - 1} + \frac{k^2AB - 1}{kB - 1} < 1:$$

- Every positive solution of system (1.11) is persistent and bounded,
- The system (1.11) has a unique positive equilibrium given by

$$x = \frac{k^2 A B - 1}{k (kB - 1)}, \ y = \frac{k^2 A B - 1}{k (kA - 1)},$$

- Every positive solution of the system (1.11) tends to the positive equilibrium of system (1.11) as $n \to \infty$,
- The unique positive equilibrium of the system (1.11) is locally asymptotically stable,
- The unique positive equilibrium of the system (1.11) is globally asymptotically stable.

In [26], Zhang and Zhang investigated the solutions, stability character and asymptotic behavior of the system of high-order nonlinear difference equations

$$x_{n+1} = \frac{x_{n-k}}{q + \prod_{i=0}^{k} y_{n-i}}, \quad y_{n+1} = \frac{y_{n-k}}{p + \prod_{i=0}^{k} x_{n-i}}, \quad k \in \mathbb{N}^+, \quad n = 0, 1, ...,$$
(1.12)

where $p, q \in (0, \infty), x_{-i} \in (0, \infty), y_{-i} \in (0, \infty)$ and i = 0, 1, ..., k. First, they obtained the equilibrium points of system (1.12) as follows:

- (0,0) and $\left(\sqrt[k+1]{1-p}, \sqrt[k+1]{1-q}\right)$ are equilibrium points if p < 1 and q < 1,
- Every point on the x-axis is an equilibrium point if q = 1,
- Every point on the *y*-axis is an equilibrium point if p = 1,
- (0,0) is the unique equilibrium point if p > 1 and q > 1.

Then, they proved the following results:

- If p > 1 and q > 1, then the unique equilibrium point (0,0) of system (1.12) is locally asymptotically stable,
- If p < 1 and q < 1, then the unique equilibrium point (0,0) of system (1.12) is unstable,
- If p < 1 and q < 1, then the positive equilibrium point $\left(\sqrt[k+1]{1-p}, \sqrt[k+1]{1-q} \right)$ of system (1.12) is unstable,
- Every solutions of system (1.12) is bounded,
- If p > 1 and q > 1, then the unique equilibrium point (0,0) of system (1.12) is globally asymptotically stable.

In [27], Zhang et al. studied the behavior of solutions of the following system

$$x_{n+1} = A + \frac{x_n}{y_{n-1}y_{n-2}}, \quad y_{n+1} = A + \frac{y_n}{x_{n-1}x_{n-2}}, \quad n = 0, 1, ...,$$
 (1.13)

where A is positive constant and $x_{-i}, y_{-i} \in (0, \infty), i = 0, 1, 2$. They obtained the results which are listed below:

- If A > 1, every positive solution of system (1.13) is bounded,
- If $A > 2/\sqrt{3}$, (c, c) is locally asymptotically stable,
- If $A > \sqrt{3}$, every positive solution of system (1.13) approaches (c, c),
- If $1 < A < 2/\sqrt{3}$, (a_1, b_1) and (a_2, b_2) are locally asymptotically stable.

In [28], Stevic et al. considered the following system of difference equations

$$x_{n+1} = A + \frac{y_n^p}{x_{n-1}^q}, \quad y_{n+1} = A + \frac{x_n^p}{y_{n-1}^q}, \quad n \in \mathbb{N}_0$$
(1.14)

where parameters A, p and q are positive and investigated the boundedness character of positive solutions of system (1.14).

They proved the following results:

• If $p^2 \ge 4q > 4$, or $p \ge 1 + q$, $q \le 1$, then system (1.14) has positive unbounded solutions where A > 0,

• If $p^2 < 4q$, or $2\sqrt{q} \le p < 1+q$, $q \in (0,1)$, then all postive solutions of system (1.14) are bounded.

In [29], Bao investigated the local stability, oscillation and boundedness character of positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{x_{n-1}^p}{y_n^p}, \quad y_{n+1} = A + \frac{y_{n-1}^p}{x_n^p}, \quad n = 0, 1, ...,$$
(1.15)

where $A \in (0, \infty)$, $p \in [1, \infty)$ and initial conditions $x_i, y_i \in (0, \infty)$, i = -1, 0.

He proved that the system (1.15) has a positive equilibrium point $(\bar{x}, \bar{y}) = (A + 1, A + 1)$ and the equilibrium point of system (1.15) is locally asymptotically stable if A > 2p-1, is unstable if 0 < A < 2p-1 and is a sink or an attracting equilibrium if $p/(A+1) < \sqrt{2}-1$. Also, he indicated that the positive solution of system (1.15) which consists of at least two semicycles is oscillatory and the system (1.15) has unbounded solutions.

In [30], Gümüş and Soykan considered the dynamical behavior of positive solutions for a system of rational difference equations of the following form

$$u_{n+1} = \frac{\alpha u_{n-1}}{\beta + \gamma v_{n-2}^p}, \quad v_{n+1} = \frac{\alpha_1 v_{n-1}}{\beta_1 + \gamma_1 u_{n-2}^p}, \quad n = 0, 1, ...,$$
(1.16)

where the parameters α , β , γ , α_1 , β_1 , γ_1 , p and the initial values u_{-i} , v_{-i} for i = 0, 1, 2 are positive real numbers.

First, they reduced the system (1.16) to the following system of difference equations

$$x_{n+1} = \frac{rx_{n-1}}{1+y_{n-2}^p}, \ y_{n+1} = \frac{sy_{n-1}}{1+x_{n-2}^p}, \ n = 0, 1, \dots,$$
(1.17)

by the change of variables $u_n = (\beta_1/\gamma_1)^{1/p} x_n$ and $v_n = (\beta/\gamma)^{1/p} y_n$ with $r = \alpha/\beta$ and $s = \alpha_1/\beta_1$.

Then, they found the equilibrium points of the system (1.17) under the certain conditions and investigated their local asymptotical behavior. Also, they proved that

- If r < 1 and s < 1, the zero equilibrium point of system (1.17) is globally asymptotically stable,
- For $r, s \in (1, \infty)$, the system (1.17) has unbounded solutions,
- If r = s = 1, the system (1.17) possesses the prime period two solution.
In [31], Din studied the qualitative behavior of positive solutions of following second-order system of rational difference equations

$$x_{n+1} = \frac{\alpha_1 + \beta_1 y_{n-1}}{a_1 + b_1 x_n}, \quad y_{n+1} = \frac{\alpha_2 + \beta_2 x_{n-1}}{a_2 + b_2 y_n}, \tag{1.18}$$

where the parameters α_i , β_i , a_i , b_i for $i \in \{1, 2\}$ and initial conditions are positive real numbers.

He determined the following results:

- Every positive solution of system (1.18) is bounded and persists when $\beta_1\beta_2 < a_1a_2$,
- The unique positive equilibrium point of system (1.18) is global attractor when $a_1a_2 \neq \beta_1\beta_2$,
- Under the some specific conditions the unique positive equilibrium point of system (1.18) is globally asymptotically stable.
- The system (1.18) has no prime period-two solutions when $a_1a_2 \neq \beta_1\beta_2$.

In [32], Mansour et al. got the exact form of the solutions and the periodic nature of the following systems of difference equations

$$x_{n+1} = \frac{x_{n-5}}{-1 + x_{n-5}y_{n-2}}, \quad y_{n+1} = \frac{y_{n-5}}{\pm 1 \pm y_{n-5}x_{n-2}}, \tag{1.19}$$

where the initial conditions are real numbers.

In [33], Elsayed and El-Metwally had the periodic nature and the form of the solutions of some systems of difference equations

$$x_{n+1} = \frac{x_n y_{n-2}}{y_{n-1} \left(\pm 1 \pm x_n y_{n-2}\right)}, \quad y_{n+1} = \frac{y_n x_{n-2}}{x_{n-1} \left(\pm 1 \pm y_n x_{n-2}\right)}, \tag{1.20}$$

where the initial conditions are nonzero real numbers.

In [34], Elsayed obtained the form of the solutions and the periodicity of the following systems of second-order rational difference equations

$$x_{n+1} = \frac{x_n y_{n-1}}{y_n \left(\pm 1 \pm x_n y_{n-1}\right)}, \quad y_{n+1} = \frac{y_n x_{n-1}}{x_n \left(\pm 1 \pm y_n x_{n-1}\right)}, \tag{1.21}$$

with the initial conditions are nonzero real numbers.

In [35], Clark and Kulenovic investigated the asymptotic and global stability behavior of solutions of the following systems of difference equations

$$x_{n+1} = \frac{x_n}{a + cy_n}, \quad y_{n+1} = \frac{y_n}{b + dx_n}, \quad n = 0, 1, \dots,$$
(1.22)

where the parameters are positive numbers and the initial conditions are arbitrary nonnegative numbers.

Then, in [36], Clark et al. completed the investigation studied in [35] of the global behavior of system (1.22).

In [37], Kulenovic and Nurkanovic studied the system of difference equations

$$x_{n+1} = Ax_n \frac{y_n}{1+y_n}, \quad y_{n+1} = By_n \frac{x_n}{1+x_n}, \quad n = 0, 1, ...,$$
 (1.23)

where the parameters A and B are in $(0, \infty)$ and the initial conditions are arbitrary nonnegative numbers. Under the special circumstances of parameters, they established the global asymptotic stability of the equilibrium points of the system (1.23). Also, there are many similar woks, see [38, 39, 40, 41, 42].

Systems of Exponential-Type Difference Equations

In this subsubsection, we review on some papers studied related to system of difference equations of exponential form.

In [43], Papaschinopoulos, Radin and Schinas studied the boundedness, the persistence and the asymptotic behavior of the positive solutions of the system of two difference equations of exponential form

$$x_{n+1} = a + bx_{n-1}e^{-y_n}, \quad y_{n+1} = c + dy_{n-1}e^{-x_n}$$
(1.24)

where a, b, c, d are positive constants, and the initial values x_{-1} , x_0 , y_{-1} , y_0 are positive real values.

They investigated the boundedness character and the existence of invariant intervals of system (1.24). Then, they found the following results. Under the conditions that $be^{-c} < 1$ and $de^{-a} < 1$, every positive solution of system (1.24) is bounded and persists. Also, they proved that the unique positive equilibrium ($\overline{x}, \overline{y}$) of system (1.24) is globally asymptotically stable under appropriate conditions. In [44], Papaschinopoulos and Schinas considered the following systems of difference equations

$$x_{n+1} = a + by_{n-1}e^{-x_n}, \ y_{n+1} = c + dx_{n-1}e^{-y_n}$$
(1.25)

$$x_{n+1} = a + by_{n-1}e^{-y_n}, \ y_{n+1} = c + dx_{n-1}e^{-x_n}$$
(1.26)

where the constants are positive real numbers and the initial values x_{-1} , x_0 , y_{-1} , y_0 are positive real numbers.

They investigated the boundedness and the persistence of the positive solutions, the existence of a unique positive equilibrium and the global asymptotic stability of the above mentioned systems. As a result, they established that every solution of the systems (1.25) and (1.26) is positive, bounded and persists if $p = bde^{-a-c} < 1$. Also, under the specific conditions, they indicated that the systems (1.25) and (1.26) have a unique positive equilibrium and every solution of these systems tends to the unique positive equilibrium of their as $n \to \infty$, each one positive equilibrium of these systems is globally asymptotically stable and finally, these systems have unbounded solutions.

In [45], Papaschinopoulos et al. investigated the boundedness, the persistence and the asymptotic behavior of the positive solutions of the following systems of difference equations

$$x_{n+1} = \frac{\alpha + \beta e^{-y_n}}{\gamma + y_{n-1}}, \quad y_{n+1} = \frac{\delta + \epsilon e^{-x_n}}{\zeta + x_{n-1}},$$
(1.27)

$$x_{n+1} = \frac{\alpha + \beta e^{-y_n}}{\gamma + x_{n-1}}, \quad y_{n+1} = \frac{\delta + \epsilon e^{-x_n}}{\zeta + y_{n-1}},$$
(1.28)

$$x_{n+1} = \frac{\alpha + \beta e^{-x_n}}{\gamma + y_{n-1}}, \quad y_{n+1} = \frac{\delta + \epsilon e^{-y_n}}{\zeta + x_{n-1}},$$
(1.29)

where α , β , γ , δ , ϵ , ζ are positive constant and the initial values x_{-1} , x_0 , y_{-1} , y_0 are positive constant.

They got the results are given below:

- For the system (1.27)
 - Every positive solution of the system (1.27) is bounded and persists,
 - If $\epsilon < \gamma$ and $\beta < \zeta$, the system (1.27) has a unique positive equilibrium and every solution of the system (1.27) tends to the unique positive equilibrium of the system (1.27) as $n \to \infty$,

- If $\frac{\beta\epsilon + (\beta+\epsilon)e 1}{\gamma\zeta} + \frac{(\alpha+\beta)(\delta+\epsilon)}{\gamma^2\zeta^2} < 1$, the unique positive equilibrium of the system (1.27) is globally asymptotically stable.
- For the system (1.28)
 - Every positive solution of the system (1.28) is bounded and persists,
 - If $\beta \epsilon < \gamma \zeta$, the system (1.28) has a unique positive equilibrium and every solution of the system (1.28) tends to the unique positive equilibrium of the system (1.28) as $n \to \infty$,
 - If $\frac{\alpha+\beta}{\gamma^2} + \frac{\delta+\epsilon}{\zeta^2} + \frac{\beta\epsilon}{\gamma\zeta} + \frac{(\alpha+\beta)(\delta+\epsilon)}{\gamma^2\zeta^2} < 1$, the unique positive equilibrium of the system (1.28) is globally asymptotically stable.
- For the system (1.29)
 - Every positive solution of the system (1.29) is bounded and persists,
 - If $\beta < \gamma$ and $\epsilon < \zeta$, the system (1.29) has a unique positive equilibrium and every solution of the system (1.29) tends to the unique positive equilibrium of the system (1.29) as $n \to \infty$,
 - If $\frac{\beta}{\gamma} + \frac{\epsilon}{\zeta} + \frac{\beta\epsilon}{\gamma\zeta} + \frac{(\alpha+\beta)(\delta+\epsilon)}{\gamma^2\zeta^2} < 1$, the unique positive equilibrium of the system (1.29) is globally asymptotically stable.

In [46], Elettreby and El-Metwally considered the system of difference equations, which describes an economic model,

$$x_{n+1} = (1-\alpha) x_n + \beta x_n (1-x_n) e^{-(x_n+y_n)}, \qquad (1.30)$$

$$y_{n+1} = (1-\alpha) y_n + \beta y_n (1-y_n) e^{-(x_n+y_n)}, \quad n = 0, 1, ...,$$
(1.31)

where α and $\beta \in (0, \infty)$ with the initial conditions x_0 and $y_0 \in (0, \infty)$.

They studied the boundedness and the invariant of the solutions of system (1.30) and also investigated global convergence for the solutions of system (1.30). Then, they obtained the following main results:

• Every positive solution $\{(x_n, y_n)\}_{n=0}^{\infty}$ of system (1.30) is bounded. Moreover,

$$\lim_{n \to \infty} \sup x_n \le \frac{\beta}{\alpha e}, \ \lim_{n \to \infty} \sup y_n \le \frac{\beta}{\alpha e}.$$

- When α ≥ β, the zero equilibrium (0,0) is a global attractor of all positive solutions of system (1.30).
- When $\alpha + \beta e^{-2} < 1$, the unique positive equilibrium point $(\overline{x}, \overline{x})$ of system (1.30) is a global attractor of all positive solutions of system (1.30).
- When $\beta (\alpha e \beta) \ge \alpha^2 e^3$, the unique positive equilibrium point $(\overline{x}, \overline{x})$ of system (1.30) is a global attractor of all positive solutions of system (1.30).
- When one of the following conditions hold

i)
$$5\beta \le 4e^2(1-\alpha)$$

ii) $\alpha + \beta < 1$

the unique positive equilibrium point $(\overline{x}, \overline{x})$ of system (1.30) is a global attractor of all positive solutions of system (1.30).

In [47], Papaschinopoulos et al. studied the asymptotic behavior of the positive solutions of the system of difference equations

$$x_{n+1} = ay_n + bx_{n-1}e^{-y_n}, \ y_{n+1} = cy_n + dy_{n-1}e^{-x_n}, \ n = 0, 1, \dots,$$
(1.32)

where a, b, c, d are positive constants and the initial values x_{-1} , x_0 , y_{-1} , y_0 are positive numbers.

Then, they prove that under the condition that $a, b, c, d \in (0, 1), a + b > 1, c + d > 1$;

- Every positive solution of system (1.32) is bounded and persists.
- Every positive solution of system (1.32) tends to the unique positive equilibrium $(\overline{x}, \overline{y})$ of system (1.32) as $n \to \infty$, when suppose that either relations

 $c\leq a, \ b\leq c, \ d\leq c \text{ or } a\leq c, \ b\leq a, \ d\leq a.$

Under the condition that $a + b \le 1$, $c + d \le 1$;

Every positive solution of system (1.32) tends to the zero equilibrium (0,0) of system (1.32) as n → ∞.

Finally, they established that where a, b, c, d are positive constants such that either

$$a+b < 1$$
, $c+d < 1$ or $a+b = 1$, $c+d = 1$,

the zero equilibrium (0,0) of system (1.32) is globally asymptotically stable.

In [48], Khan investigated the qualitative behavior of positive solutions of the following two systems of exponential rational difference equations

$$x_{n+1} = \frac{\alpha e^{-y_n} + \beta e^{-y_{n-1}}}{\gamma + \alpha y_n + \beta y_{n-1}}, \quad y_{n+1} = \frac{\alpha_1 e^{-x_n} + \beta_1 e^{-x_{n-1}}}{\gamma_1 + \alpha_1 x_n + \beta_1 x_{n-1}}, \quad n = 0, 1, ...,$$
(1.33)

$$x_{n+1} = \frac{\alpha e^{-x_n} + \beta e^{-x_{n-1}}}{\gamma + \alpha y_n + \beta y_{n-1}}, \quad y_{n+1} = \frac{\alpha_1 e^{-y_n} + \beta_1 e^{-y_{n-1}}}{\gamma_1 + \alpha_1 x_n + \beta_1 x_{n-1}}, \quad n = 0, 1, ...,$$
(1.34)

where α , β , γ , α_1 , β_1 , γ_1 and the initial conditions are positive real numbers. They obtained the results are given below:

- For the system (1.33)
 - Every positive solution of the system (1.33) is bounded and persists,

- If
$$(\alpha + \beta) e^{-L_2} < \overline{x} (\gamma + (\alpha + \beta) L_2)$$
 and
 $(\alpha_1 + \beta_1) e^{-L_1} < \overline{y} (\gamma_1 + (\alpha_1 + \beta_1) L_1),$

the unique positive equilibrium point of the system (1.33) is globally asymptotically stable.

- For the system (1.34)
 - Every positive solution of the system (1.34) is bounded and persists,
 - If $(\alpha + \beta) e^{-L_1} < \overline{x} (\gamma + (\alpha + \beta) L_2)$ and $(\alpha_1 + \beta_1) e^{-L_2} < \overline{y} (\gamma_1 + (\alpha_1 + \beta_1) L_1)$, the unique positive equilibrium point of the system (1.34) is globally asymptotically stable.

Systems of Max-Type Difference Equations

In [49], Simsek, Demir and Cinar considered the behavior of the solutions of the following system of difference equations

$$x_{n+1} = \max\left\{\frac{A}{x_n}, \frac{y_n}{x_n}\right\}, \quad y_{n+1} = \max\left\{\frac{A}{y_n}, \frac{x_n}{y_n}\right\},\tag{1.35}$$

where the constant A and the initial conditions are positive real numbers. They proved that the system (1.35) has unbounded solutions for special cases.

But then, in [50], Stevic corrected the results given in [49] and showed that the general solution to the max-type system of difference equations (1.35) for the case

$$y_0, x_0 \ge A > 0, y_0/x_0 \ge \max\{A, 1/A\},\$$

is given by:

$$x_n = \left(\frac{A^{f_{k(n)-1}-\alpha(n)}x_0^{f_{k(n)}}}{y_0^{f_{k(n)}}}\right)^{(-1)^n}, \ n \in \mathbb{N},$$

and

$$y_n = \left(\frac{y_0^{f_{k(n-1)+1}}}{A^{f_{k(n-1)}+\alpha(n)-1}x_0^{f_{k(n-1)+1}}}\right)^{(-1)^n}, \ n \ge 2.$$

In [51], Fotiades and Papaschinopoulos studied the periodic character of the solutions of the system of the difference equations

$$x_{n+1} = \max\left\{A, \frac{y_n}{x_{n-1}}\right\}, \quad y_{n+1} = \max\left\{B, \frac{x_n}{y_{n-1}}\right\},$$
 (1.36)

where A, B are positive constants and the initial values x_{-1} , x_0 , y_{-1} , y_0 are positive numbers.

The authors established that every solution of system (1.36) is eventually periodic for the cases:

$$1 \le A \le B, \ A < 1 \le B, \ A \le B < 1, \ 1 \le B \le A, \ B < 1 \le A, \ B \le A < 1.$$

In [52], Stevic studied behavior of positive solutions of the max-type system of difference equations

$$x_{n+1} = \max\left\{c, \frac{y_n^p}{x_{n-1}^p}\right\}, \ y_{n+1} = \max\left\{c, \frac{x_n^p}{y_{n-1}^p}\right\}, \ n \in \mathbb{N}_0$$
(1.37)

where $p,c \in (0,\infty)$. In his work, boundedness character and global attractivity are investigated for some special cases.

For the case $p \in (0, 4)$ and c > 0, boundedness of all positive solutions of system (1.37) is determined. Also, for $p \in (0, 4)$ and $c \ge 1$, it is given that every positive solution $(x_n, y_n)_{n\ge 1}$ of system (1.37) is eventually equal to (c, c). Besides, the system (1.37) has positive unbounded solutions when $p \ge 4$ and c > 0. Finally, every positive solution of system (1.37) converges to (1, 1) when $p \in (0, 1)$ and $c \in (0, 1)$.

In [53], Stevic et al. studied the boundedness character of positive solutions of system of difference equations

$$x_{n+1} = \max\left\{A, \frac{y_n^p}{x_{n-1}^q}\right\}, \quad y_{n+1} = \max\left\{A, \frac{x_n^p}{y_{n-1}^q}\right\}, \quad n \in \mathbb{N}_0$$
(1.38)

with $\min \{A, p, q\} > 0$.

Consequently, the following statements are obtained:

- All positive solutions of system (1.38) are bounded when A > 0, $2\sqrt{q} \le p < 1 + q$ and $q \in (0, 1)$.
- All positive solutions of system (1.38) are bounded when A > 0, p > 0 and $p^2 < 4q$.
- All positive solutions of system (1.38) are bounded when A > 0, p = 1 + q and $q \in (0, 1)$.
- The system (1.38) has positive unbounded solutions if A > 0, $p^2 \ge 4q \ge 4$, or p > 1 + q and $q \in (0, 1)$.

1.1.2 Review on Three-Dimensional Systems of Nonlinear Difference Equations

This subsection is concerned with review of dynamical behavior of solutions of the systems of three-dimensional nonlinear difference equations.

Systems of Rational-Type Difference Equations

In [54], Kulenovic and Nurkanovic studied the global behavior of solutions of the system of difference equations

$$x_{n+1} = \frac{a+x_n}{b+y_n}, \quad y_{n+1} = \frac{c+y_n}{d+z_n}, \quad z_{n+1} = \frac{e+z_n}{f+x_n}, \quad n = 0, 1, \dots,$$
(1.39)

where the parameters a, b, c, d, e and f are in $(0, \infty)$ and the initial conditions are arbitrary non-negative numbers.

They indicated that the equilibrium of system (1.39) is locally asymptotically stable if $b \ge 1$, $d \ge 1$, $f \ge 1$, and obtained the global asymptotic stability of the unique positive equilibrium for several cases depending of some special values of the parameters.

In [55] Kurbanli, in [56] Kurbanli and in [57] Kurbanli et al. investigated the behavior of the solutions of the difference equations systems

$$x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} - 1}, \quad y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} - 1}, \quad z_{n+1} = \frac{z_{n-1}}{y_n z_{n-1} - 1}$$
(1.40)

$$x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} - 1}, \quad y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} - 1}, \quad z_{n+1} = \frac{1}{y_n z_n}, \tag{1.41}$$

$$x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} - 1}, \quad y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} - 1}, \quad z_{n+1} = \frac{x_n}{y_n z_{n-1}}, \tag{1.42}$$

where the initial conditions are arbitrary real numbers, respectively.

They found all exact solutions of systems (1.40), (1.41), and (1.42) under special conditions and showed that the systems have unbounded solutions.

In [58], Özkan and Kurbanli studied the periodical solutions of the systems of difference equations

$$x_{n+1} = \frac{y_{n-2}}{-1 \pm y_{n-2} x_{n-1} y_n}, \ y_{n+1} = \frac{x_{n-2}}{-1 \pm x_{n-2} y_{n-1} x_n}, \ z_{n+1} = \frac{x_{n-2} + y_{n-2}}{-1 \pm x_{n-2} y_{n-1} x_n}, \ n \in \mathbb{N}_0,$$

where the initial conditions are arbitrary real numbers. They obtained all six-period solutions of given systems under special conditions.

In [59], Stevic showed that the following system of difference equations

$$x_{n+1} = \frac{a_1 x_{n-2}}{b_1 y_n z_{n-1} x_{n-2} + c_1}, \ y_{n+1} = \frac{a_2 y_{n-2}}{b_2 z_n x_{n-1} y_{n-2} + c_2}, \ z_{n+1} = \frac{a_3 z_{n-2}}{b_3 x_n y_{n-1} z_{n-2} + c_3}, \ n \in \mathbb{N}_0$$

where the parameters and the initial conditions are real numbers, can be solved.

1.1.3 Review on Multi-Dimensional Systems of Nonlinear Difference Equations

This subsection is concerned with review of dynamical behavior of solutions of the systems of multi-dimensional nonlinear difference equations.

Systems of Max-Type Difference Equations

In [60], Stevic studied the system of max-type difference equations

$$\begin{aligned}
x_{n}^{(1)} &= \max_{1 \leq i \leq m_{1}} \left\{ f_{1i} \left(x_{n-k_{i,1}^{(1)}}^{(1)}, x_{n-k_{i,2}^{(2)}}^{(2)}, \dots, x_{n-k_{i,l}^{(1)}}^{(l)}, n \right), x_{n-s}^{(1)} \right\}, \\
x_{n}^{(2)} &= \max_{1 \leq i \leq m_{2}} \left\{ f_{2i} \left(x_{n-k_{i,1}^{(2)}}^{(1)}, x_{n-k_{i,2}^{(2)}}^{(2)}, \dots, x_{n-k_{i,l}^{(2)}}^{(l)}, n \right), x_{n-s}^{(2)} \right\}, \\
&\vdots \end{aligned}$$
(1.43)

$$x_n^{(l)} = \max_{1 \le i \le m_l} \left\{ f_{li} \left(x_{n-k_{i,1}^{(l)}}^{(1)}, x_{n-k_{i,2}^{(l)}}^{(2)}, \dots, x_{n-k_{i,l}^{(l)}}^{(l)}, n \right), x_{n-s}^{(l)} \right\},$$

 $n \in \mathbb{N}_0$, where $s, l, m_j, k_{i,t}^{(j)} \in \mathbb{N}, j, t \in \{1, ..., l\}$ and for a fixed $j, i \in \{1, ..., m_j\}$, and where the functions $f_{ij}: (0, \infty)^l \times \mathbb{N}_0 \to (0, \infty), j \in \{1, ..., l\}, i \in \{1, ..., m_j\}$.

He proved that every positive solution to system (1.43) is eventually periodic with period s under some conditions. Also, he proved some related results for the corresponding system of min-type difference equations.

In [61], Stevic and Iricanin investigated the long-term behavior of positive solutions of the cyclic system of difference equations

$$x_{n+1}^{(i)} = \max\left\{\alpha, \frac{\left(x_n^{(i+1)}\right)^p}{\left(x_{n-1}^{(i+2)}\right)^q}\right\}, \quad i = 1, \dots, k, \ n \in \mathbb{N}_0,\tag{1.44}$$

where $k \in \mathbb{N}$, $\min \{\alpha, p, q\} > 0$.

They showed that the system (1.44) has bounded and unbounded solutions depending on the status of the parameters and gave some sufficient conditions which guaranty the global attractivity of all positive solutions of system (1.44).

1.2 LITERATURE REVIEWS FOR DIFFERENCE EQUATIONS AND DIS-CRETE SYSTEMS VIA INTEGER SEQUENCES

In this section, we study the recent investigations on the forms of solutions of systems difference equations and difference equations in terms of well-known integer sequences such as Fibonacci numbers, Horadam numbers, Padovan numbers. We focus on the papers given some interesting relationships both between the exact solutions of difference equations and the integer sequences and between the equilibrium points of difference equations and golden ratio, plastic number. In [74], Tollu et al. considered the following difference equations

$$x_{n+1} = \frac{1}{1+x_n}, \quad y_{n+1} = \frac{1}{-1+y_n}, \quad n = 0, 1, ...,$$
 (1.45)

such that their solutions are associated with Fibonacci numbers, where initial conditions are $x_0 \in \mathbb{R} - \left\{-\frac{F_{m+1}}{F_m}\right\}_{m=1}^{\infty}$ and $y_0 \in \mathbb{R} - \left\{-\frac{F_{m+1}}{F_m}\right\}_{m=1}^{\infty}$ and F_m is the *m*th Fibonacci number.

They investigated the some relationships both between Fibonacci numbers and solutions of equations (1.45) and between the golden ratio and equilibrium points of equations (1.45). Then, they proved that: the solutions of equations (1.45) are given by

$$x_n = \frac{F_n + F_{n-1}x_0}{F_{n+1} + F_n x_0}, \quad y_n = \frac{F_{-n} + F_{-(n-1)}y_0}{F_{-(n+1)} + F_{-n}y_0},$$

where F_n is the *n*th Fibonacci number, and the nontrival solutions of equations (1.45) converge to $-\beta$ and β , so that β is conjugate to the golden ratio.

Next, Rabago [75] presented a theoretical explanation in deriving the closed-form solution of Eq.(1.45) which Tollu et al. studied in [74] and provided another approach in proving Sroysang's conjecture (2013).

Then, in [76], Yazlik et al. studied the following rational difference equation systems

$$x_{n+1} = \frac{x_{n-1} \pm 1}{y_n x_{n-1}}, \quad y_{n+1} = \frac{y_{n-1} \pm 1}{x_n y_{n-1}}, \quad n = 0, 1, \dots,$$
(1.46)

such that their solutions associated with Padovan numbers. In their study, they obtained that the forms of solutions of system (1.46) are as follows

$$x_n = \begin{cases} \mp \frac{P_n x_{-1} y_0 \mp P_{n+1} x_{-1} + P_{n-1}}{P_{n-1} x_{-1} y_0 \mp P_n x_{-1} + P_{n-2}}, & \text{if } n \text{ is odd} \\ \\ \mp \frac{P_n y_{-1} x_0 \mp P_{n+1} y_{-1} + P_{n-1}}{P_{n-1} y_{-1} x_0 \mp P_n y_{-1} + P_{n-2}}, & \text{if } n \text{ is even} \end{cases}$$
$$y_n = \begin{cases} \mp \frac{P_n y_{-1} x_0 \mp P_{n+1} y_{-1} + P_{n-1}}{P_{n-1} y_{-1} x_0 \mp P_n y_{-1} + P_{n-2}}, & \text{if } n \text{ is odd} \\ \\ \mp \frac{P_n x_{-1} y_0 \mp P_{n+1} x_{-1} + P_{n-2}}{P_{n-1} x_{-1} y_0 \mp P_n x_{-1} + P_{n-2}}, & \text{if } n \text{ is even} \end{cases}$$

where P_n is the *n*th Padovan number. Also, they demonstrated that every solutions of the systems (1.46) converge to point (p, p) and (-p, -p), where *p* is the plastic number. Tollu et al. [77] considered the following four Riccati difference equations

$$x_{n+1} = \frac{1+x_n}{x_n}, \quad y_{n+1} = \frac{1-y_n}{y_n}, \quad u_{n+1} = \frac{1}{u_n+1}, \quad v_{n+1} = \frac{1}{v_n-1}, \tag{1.47}$$

in which the initial conditions are real numbers. They derived the formulae for the solutions of equations (1.47) are given by

$$x_n = \frac{F_{n+1}x_0 + F_n}{F_n x_0 + F_{n-1}},$$

$$y_n = \frac{F_{-(n+1)}y_0 + F_{-n}}{F_{-n}y_0 + F_{-(n-1)}},$$

$$u_n = \frac{F_n + F_{n-1}u_0}{\overline{----}},$$

$$v_n = \frac{F_{n+1} + F_n u_0}{F_{-n} + F_{-(n-1)} v_0},$$

where F_n is *n*th Fibonacci number, F_{-n} is *n*th negative Fibonacci number. In addition to, they stated the asymptotic behaviors of the solutions of these equations and introduced that every solutions of these equations converge to their positive or negative equilibrium points.

Also, they in [78] studied the systems of difference equations

$$x_{n+1} = \frac{1+p_n}{q_n}, \quad y_{n+1} = \frac{1+r_n}{s_n}, \quad n \in \mathbb{N}_0,$$

where each of the sequences p_n , q_n , r_n and s_n is some of the sequences x_n or y_n by their own. They solved fourteen systems out of sixteen possible systems. In particularly, the representation formulae of solutions of twelve systems were stated via Fibonacci numbers. Also, for ten systems, they expressed that the solutions of these systems tend to the unique point (α, α) where α is the golden ratio.

In [79], Halim concerned with the following systems of rational difference equations

$$x_{n+1} = \frac{1}{1+y_n}, \quad y_{n+1} = \frac{1}{1+x_n}, \quad n = 0, 1, ...,$$
 (1.48)

and

$$x_{n+1} = \frac{1}{1 - y_n}, \quad y_{n+1} = \frac{1}{1 - x_n}, \quad n = 0, 1, \dots,$$
(1.49)

initial conditions are arbitrary nonzero real numbers. He determined the form of solutions of system (1.48) as given below

$$\begin{aligned} x_{2n-1} &= \frac{F_{2n-1} + F_{2n-2}y_0}{F_{2n} + F_{2n-1}y_0}, \quad x_{2n} &= \frac{F_{2n} + F_{2n-1}x_0}{F_{2n+1} + F_{2n}x_0}, \\ y_{2n-1} &= \frac{F_{2n-1} + F_{2n-2}x_0}{F_{2n} + F_{2n-1}x_0}, \quad y_{2n} &= \frac{F_{2n} + F_{2n-1}y_0}{F_{2n+1} + F_{2n}y_0}, \end{aligned}$$

and proved that the equilibrium point E of system (1.48) is globally asymptotically stable, where $E = \left(\frac{-1+\sqrt{5}}{2}, \frac{-1+\sqrt{5}}{2}\right) = \left(\frac{1}{\alpha}, \frac{1}{\alpha}\right)$, where α is the golden ratio. Furthermore, he established the solutions of system (1.49) are periodic with period six and are unstable.

In [80], Bacani and Rabago studied the behavior of solutions of the following nonlinear difference equations

$$x_{n+1} = \frac{q}{p+x_n^v}$$
 and $y_{n+1} = \frac{q}{-p+y_n^v}$, (1.50)

where $p, q \in \mathbb{R}^+$ and $v \in \mathbb{N}$. They proved that the solutions of equations (1.50) are as follows

$$x_n = \frac{qW_n + x_0 qW_{n-1}}{W_{n+1} + x_0 W_n},$$

$$y_n = \frac{qW_{-n} + y_0 qW_{-(n-1)}}{W_{-(n+1)} + y_0 W_{-n}},$$

where W_n is the *n*th Horadam number.

In [81], Halim and Bayram investigated the solutions, stability character, and asymptotic behavior of the difference equation

$$x_{n+1} = \frac{\alpha}{\beta + \gamma x_{n-k}}, \quad n \in \mathbb{N}_0, \tag{1.51}$$

where the initial conditions $x_{-k}, x_{-k+1}, ..., x_0$ are nonzero real numbers, such that its solutions are associated to Horadam numbers, which are generalized Fibonacci numbers. Firstly, they had the difference equation

$$x_{n+1} = \frac{q}{p + x_{n-k}},\tag{1.52}$$

by putting $q = \frac{\alpha}{\gamma}$ and $p = \frac{\beta}{\gamma}$. Then, they proved that the forms of the solutions of difference equation (1.52) are as follows

$$x_{(k+1)n+i} = \frac{W_{n+1} + W_n x_{i-(k+1)}}{W_{n+2} + W_{n+1} x_{i-(k+1)}} q, \quad i = 1, 2, ..., k+1,$$

where W_n is the *n*th Horadam number. Also, they obtained that the equilibrium point E of difference equation (1.52) is globally asymptotically stable, where $E = \frac{-p + \sqrt{p^2 + 4q}}{2}$. Then, in [82] Halim considered the system of difference equations

$$x_{n+1} = \frac{1}{1+y_{n-2}}, \quad y_{n+1} = \frac{1}{1+x_{n-2}}, \quad n = 0, 1, ...,$$
 (1.53)

where $\mathbb{N}_0 = \mathbb{N} \cup \{0\}$ and the initial conditions x_{-2} , x_{-1} , x_0 , y_{-2} , y_{-1} , and y_0 are real numbers. He presented the relationship between Fibonacci numbers and the solutions of system (1.53), i.e., the form of the solutions of system (1.53) are given by

$$\begin{aligned} x_{6n+i} &= \frac{F_{2n+1} + F_{2n} y_{i-3}}{F_{2n+2} + F_{2n+1} y_{i-3}}, \quad i = 1, 2, 3, \\ y_{6n+i} &= \frac{F_{2n+1} + F_{2n} x_{i-3}}{F_{2n+2} + F_{2n+1} x_{i-3}}, \quad i = 1, 2, 3, \\ x_{6n+i} &= \frac{F_{2n+2} + F_{2n+1} x_{i-6}}{F_{2n+3} + F_{2n+2} x_{i-6}}, \quad i = 4, 5, 6, \\ y_{6n+i} &= \frac{F_{2n+2} + F_{2n+1} y_{i-6}}{F_{2n+3} + F_{2n+2} y_{i-6}}, \quad i = 4, 5, 6, \end{aligned}$$

where F_n is the *n*th Fibonacci number. Furthermore, he shown that the equilibrium point E of system (1.53) is globally asymptotically stable, where $E = \left(\frac{-1+\sqrt{5}}{2}, \frac{-1+\sqrt{5}}{2}\right)$. El-Dessoky in [83] dealt with the following difference equation

$$x_{n+1} = ax_n + \frac{\alpha x_n x_{n-l}}{\beta x_n + \gamma x_{n-k}}, \quad n = 0, 1, ...,$$
(1.54)

where the parameters α , β , γ and a and the initial conditions x_{-t} , x_{-t+1} , x_{-1} and x_0 where $t = \max\{l, k\}$ are positive real numbers. He introduced the explicit formula of solutions of some special cases of Eq.(1.54) via Fibonacci numbers and also, discussed the global behavior of solutions of Eq.(1.54).

In [84], Halim and Rabago studied the systems of difference equaions

$$x_{n+1} = \frac{1}{\pm 1 \pm y_{n-k}}, \quad y_{n+1} = \frac{1}{\pm 1 \pm x_{n-k}}, \quad n, k \in \mathbb{N}_0,$$
(1.55)

where the initial conditions x_{-k} , x_{-k+1} , ..., x_0 , y_{-k} , y_{-k+1} , ..., y_0 are nonzero real numbers. Initially, they examined the form and behavior of solutions of system of difference equations

$$x_{n+1} = \frac{1}{1+y_{n-k}}, \quad y_{n+1} = \frac{1}{1+x_{n-k}}.$$
 (1.56)

Therefore, they determined that the exact solutions of system (1.56) are as follows

$$\begin{aligned} x_{2(k+1)n+i} &= \frac{F_{2n+1}+F_{2n}y_{i-(k+1)}}{F_{2n+2}+F_{2n+1}y_{i-(k+1)}}, & i = 1, 2, ..., k+1, \\ y_{2(k+1)n+i} &= \frac{F_{2n+1}+F_{2n}x_{i-(k+1)}}{F_{2n+2}+F_{2n+1}x_{i-(k+1)}}, & i = 1, 2, ..., k+1, \\ x_{2(k+1)n+i} &= \frac{F_{2n+2}+F_{2n+1}x_{i-(2k+2)}}{F_{2n+3}+F_{2n+2}x_{i-(2k+2)}}, & i = k+2, ..., 2k+2, \\ y_{2(k+1)n+i} &= \frac{F_{2n+2}+F_{2n+1}y_{i-(2k+2)}}{F_{2n+3}+F_{2n+2}y_{i-(2k+2)}}, & i = k+2, ..., 2k+2, \end{aligned}$$

and the equilibrium point of system (1.56) is globally asymptotically stable. In addition, the authors given some results for other systems.

Then, in [85], the authors studied the rational difference equation

$$x_{n+1} = \frac{\alpha x_{n-1} + \beta}{\gamma x_n x_{n-1}}, \quad n \in \mathbb{N}_0, \tag{1.57}$$

where $\mathbb{N}_0 = \mathbb{N} \cup \{0\}$, $\alpha, \beta, \gamma \in \mathbb{R}^+$ and the initial conditions nonzero real numbers and also investigated the two-dimensional case of the this equation given by

$$x_{n+1} = \frac{\alpha x_{n-1} + \beta}{\gamma y_n x_{n-1}}, \quad y_{n+1} = \frac{\alpha y_{n-1} + \beta}{\gamma x_n y_{n-1}}, \quad n \in \mathbb{N}_0.$$
(1.58)

Firstly, they reduced the difference equation (1.57) to the difference equation

$$x_{n+1} = \frac{px_{n-1} + q}{x_n x_{n-1}} \tag{1.59}$$

by using changes variables $p = \frac{\alpha}{\gamma}$ and $q = \frac{\beta}{\gamma}$. Then, they presented that the closed-form solution of difference equation (1.59) is given by

$$x_n = \frac{S_{n+1}x_{-1} + S_n x_0 x_{-1} + q S_{n-1}}{S_n x_{-1} + S_{n-1} x_0 x_{-1} + q S_{n-2}},$$

where S_n is the *n*th generalized Padovan number and the equilibrium point of Eq.(1.59) is globally asymptotically stable.

Later, they reduced the system of difference equation (1.58) to the system

$$x_{n+1} = \frac{px_{n-1} + q}{y_n x_{n-1}}, \quad y_{n+1} = \frac{py_{n-1} + q}{x_n y_{n-1}}$$
(1.60)

by using changes variables $p = \frac{\alpha}{\gamma}$ and $q = \frac{\beta}{\gamma}$. Then, they presented that the closed-form solutions of system (1.60) are given by

$$x_n = \begin{cases} \frac{S_{n+1}y_{-1}+S_nx_0y_{-1}+qS_{n-1}}{S_ny_{-1}+S_{n-1}x_0y_{-1}+qS_{n-2}}, & \text{if } n \text{ is even,} \\ \frac{S_{n+1}x_{-1}+S_ny_0x_{-1}+qS_{n-1}}{S_nx_{-1}+S_{n-1}y_0x_{-1}+qS_{n-2}}, & \text{if } n \text{ is odd,} \end{cases}$$
$$y_n = \begin{cases} \frac{S_{n+1}x_{-1}+S_ny_0x_{-1}+qS_{n-1}}{S_nx_{-1}+S_{n-1}y_0x_{-1}+qS_{n-2}}, & \text{if } n \text{ is even,} \\ \frac{S_{n+1}y_{-1}+S_nx_0y_{-1}+qS_{n-2}}{S_ny_{-1}+S_{n-1}x_0y_{-1}+qS_{n-2}}, & \text{if } n \text{ is odd,} \end{cases}$$

and the equilibrium point of the system (1.60) is global attractor.

Then, in [86], Stevic et al. the following nonlinear second-order difference equation

$$x_{n+1} = a + \frac{b}{x_n} + \frac{c}{x_n x_{n-1}}, \ n \in \mathbb{N}_0,$$
(1.61)

in which parameters a, b, c and the initial values x_{-1} and x_0 are complex numbers such that $c \neq 0$. Next, they used the following change of variables

$$x_n = \frac{y_n}{y_{n-1}},$$

and obtained the following third-order linear difference equation with constant coefficients

$$y_{n+1} = ay_n + by_{n-1} + cy_{n-2}.$$

After, they introduced that the representation formula of every solution of Eq.(1.61) is

$$x_n = \frac{(s_{n+1} - as_n)x_{-1} + s_n x_0 x_{-1} + cs_{n-1}}{(s_n - as_{n-1})x_{-1} + s_{n-1} x_0 x_{-1} + cs_{n-2}},$$

where s_n is the *n*th generalized Padovan number. Note that, Eq.(1.57) is a special case of Eq.(1.61) such that a = 0.

Alotaibi et al. in [87] considered the following systems of difference equations

$$x_{n+1} = \frac{y_n y_{n-2}}{x_{n-1} + y_{n-2}}, \quad y_{n+1} = \frac{x_n x_{n-2}}{\pm y_{n-1} \pm x_{n-2}}, \quad n = 0, 1, \dots,$$
(1.62)

where the initial conditions x_{-2} , x_{-1} , x_0 , y_{-2} , y_{-1} , y_0 are arbitrary positive real numbers. They analyzed the solutions of the systems (1.62) such that their solutions are associated with Fibonacci numbers.

In [88], El-Dessoky et al. examined the following difference equation

$$y_{n+1} = \frac{\beta y_n y_{n-3}}{A y_{n-4} + B y_{n-3}}, \quad n = 0, 1, \dots,$$
(1.63)

where α , β , A, and B are real numbers and the initial values y_{-4} , y_{-3} , y_{-2} , y_{-1} and y_0 are positive real numbers. They presented the solutions of Eq.(1.63) in terms of Fibonacci numbers according to some special cases of the parameters α , β , A, and B.

Then, in [89], Matsunaga and Suzuki studied the following system of rational difference equations

$$x_{n+1} = \frac{ay_n + b}{cy_n + d}, \quad y_{n+1} = \frac{ax_{n-1} + b}{cx_n + d}, \quad n = 0, 1, \dots,$$
(1.64)

where the parameters a, b, c, d and the initial values x_0 , y_0 are real numbers. They obtained that the explicit solutions of system (1.64) are as follows

$$\begin{aligned} x_{2n-1} &= \frac{(ay_0+b) G_{2n-1} + (bc-ad) y_0 G_{2n-2}}{G_{2n} + (cy_0-a) G_{2n-1}}, \ x_{2n} &= \frac{(ax_0+b) G_{2n} + (bc-ad) x_0 G_{2n-1}}{G_{2n+1} + (cx_0-a) G_{2n}}, \\ y_{2n-1} &= \frac{(ax_0+b) G_{2n-1} + (bc-ad) x_0 G_{2n-2}}{G_{2n} + (cx_0-a) G_{2n-1}}, \ y_{2n} &= \frac{(ay_0+b) G_{2n} + (bc-ad) y_0 G_{2n-1}}{G_{2n+1} + (cy_0-a) G_{2n}}, \end{aligned}$$

where G_n is a generalized Fibonacci sequence defined by

$$G_{n+2} = (a+d) G_{n+1} + (bc - ad) G_n,$$

with $G_0 = 0$ and $G_1 = 1$. Moreover, they presented that every solution of system (1.64) converges to its equilibrium points.

In [90], Öcalan and Duman considered the following nonlinear recursive difference equation

$$x_{n+1} = \frac{x_{n-1}}{x_n}, \quad n = 0, 1, \dots,$$
(1.65)

with any nonzero initial values x_{-1} and x_0 . Then, they extended their all results to solutions of the following nonlinear recursive equations

$$x_{n+1} = \left(\frac{x_{n-1}}{x_n}\right)^p, \ p > 0 \text{ and } n = 0, 1, ...,$$
 (1.66)

with any nonzero initial values x_{-1} and x_0 . Later, they obtained that the exact solution of Eq.(1.65) is

$$x_n = \begin{cases} \frac{x_{-1}^{t_n-1}}{x_0^{f_n}} & \text{if } n = 1, 3, 5, ..., \\ \frac{x_0^{f_n}}{x_{-1}^{f_{n-1}}} & \text{if } n = 2, 4, 6, ..., \end{cases}$$

where f_n is the *n*th Fibonacci number. Under the special case of initial values, they determined that there exist non-oscillatory positive solutions of Eq.(1.65), which converge monotonically to the equilibrium point 1.

Furthermore, they given that the exact solution of Eq.(1.66) is

$$x_n = \begin{cases} \frac{x_{-1}^{g_{n-1}(p)}}{x_0^{f_n(p)}} & \text{if } n = 1, 3, 5, ..., \\ \frac{x_0^{f_n(p)}}{x_{-1}^{f_{n-1}(p)}} & \text{if } n = 2, 4, 6, ..., \end{cases}$$

where $f_n(p)$ and $g_n(p)$ are the *n*th Fibonacci-type number. And also, under the special case of initial values, they demonstrated that there exist non-oscillatory positive solutions of Eq.(1.66), which converge monotonically to the equilibrium point 1 and the Eq.(1.66) has unbounded solutions.

Next, Akrour et al. [91] studied the following system of difference equations

$$x_{n+1} = \frac{ay_n x_{n-1} + bx_{n-1} + c}{y_n x_{n-1}}, \ y_{n+1} = \frac{ax_n y_{n-1} + by_{n-1} + c}{x_n y_{n-1}}, \ n = 0, 1, \dots,$$

where the parameters a, b, c are arbitrary real numbers with $c \neq 0$ and the initial values x_{-1}, x_0, y_{-1} and y_0 are arbitrary nonzero real numbers. They examined that the explicit solutions of system (1.56) are given by

$$\begin{aligned} x_{2n+1} &= \frac{cJ_{2n+1} + (J_{2n+3} - aJ_{2n+2})x_{-1} + J_{2n+2}x_{-1}y_0}{cJ_{2n} + (J_{2n+2} - aJ_{2n+1})x_{-1} + J_{2n+1}x_{-1}y_0}, \\ x_{2n+2} &= \frac{cJ_{2n+2} + (J_{2n+4} - aJ_{2n+3})y_{-1} + J_{2n+3}x_0y_{-1}}{cJ_{2n+1} + (J_{2n+3} - aJ_{2n+2})y_{-1} + J_{2n+2}x_0y_{-1}}, \\ y_{2n+1} &= \frac{cJ_{2n+1} + (J_{2n+3} - aJ_{2n+2})y_{-1} + J_{2n+2}x_0y_{-1}}{cJ_{2n} + (J_{2n+2} - aJ_{2n+1})y_{-1} + J_{2n+1}x_0y_{-1}}, \\ y_{2n+2} &= \frac{cJ_{2n+2} + (J_{2n+4} - aJ_{2n+3})x_{-1} + J_{2n+3}x_{-1}y_0}{cJ_{2n+1} + (J_{2n+3} - aJ_{2n+2})x_{-1} + J_{2n+2}x_{-1}y_0}, \end{aligned}$$

where J_n is defined by the recurrent relation

$$J_{n+3} = aJ_{n+2} + bJ_{n+1} + cJ_n, \ n \in \mathbb{N},$$

such that $J_0 = 0$, $J_1 = 1$, $J_2 = a$.

For related studies on solving difference equations and systems of difference equations and investigating the asymptotic behavior of their solutions, see [92, 100].





CHAPTER 2

THE GENERAL DEFINITIONS AND THEOREMS

In this chapter, we state some definitions and theorems used in this thesis. For details, see [1-14, 105-120].

2.1 DIFFERENCE EQUATIONS

In this section, we give some important definitions and theorems about difference equations and systems of difference equations (discrete dynamical systems).

Let I be some interval of real numbers and let $f: I^{k+1} \to I$ be a continuously differentiable function. A difference equation of order (k+1) is an equation of the form

$$x_{n+1} = f(x_n, x_{n-1}, \dots, x_{n-k}), \quad n = 0, 1, \dots$$
(2.1)

A solution of Eq.(2.1) is a sequence $\{x_n\}_{n=-k}^{\infty}$ that satisfies Eq.(2.1) for all $n \ge -k$.

Definition 2.1 A solution of Eq.(2.1) that is constant for all $n \ge -k$ is called an equilibrium solution of Eq.(2.1). If

$$x_n = \overline{x}$$
, for all $n \ge -k$

is an equilibrium solution of Eq.(2.1), then \overline{x} is called an equilibrium point, or simply an equilibrium of Eq.(2.1).

Definition 2.2 (Stability) Let \overline{x} an equilibrium point of Eq.(2.1).

(a) An equilibrium point x̄ of Eq.(2.1) is called locally stable if, for every ε > 0; there exists δ > 0 such that if {x_n}_{n=-k}[∞] is a solution of Eq.(2.1) with

$$|x_{-k} - \overline{x}| + |x_{1-k} - \overline{x}| + \dots + |x_0 - \overline{x}| < \delta,$$

then

 $|x_n - \overline{x}| < \varepsilon$, for all $n \ge -k$.

(b) An equilibrium point x̄ of Eq.(2.1) is called locally asymptotically stable if, it is locally stable, and if in addition there exists γ > 0 such that if {x_n}_{n=-k}[∞] is a solution of Eq.(2.1) with

$$|x_{-k} - \overline{x}| + |x_{-k+1} - \overline{x}| + \dots + |x_0 - \overline{x}| < \gamma,$$

then we have

 $\lim_{n \to \infty} x_n = \overline{x}.$

(c) An equilibrium point \overline{x} of Eq.(2.1) is called a global attractor if, for every solution $\{x_n\}_{n=-k}^{\infty}$ of Eq.(2.1), we have

 $\lim_{n \to \infty} x_n = \overline{x}.$

- (d) An equilibrium point x̄ of Eq. (2.1) is called globally asymptotically stable if it is locally stable, and a global attractor.
- (e) An equilibrium point \overline{x} of Eq.(2.1) is called unstable if it is not locally stable.

Suppose that the function f is continuously differentiable in some open neighborhood of an equilibrium point \overline{x} . Let

$$q_i = \frac{\partial f}{\partial u_i}(\overline{x}, \overline{x}, ..., \overline{x}), \text{ for } i = 0, 1, ..., k$$

denote the partial derivative of $f(u_0, u_1, ..., u_k)$ with respect to u_i evaluated at the equilibrium point \overline{x} of Eq.(2.1).

Definition 2.3 The equation

$$y_{n+1} = q_0 y_n + q_1 y_{n-1} + \dots + q_k y_{n-k}, \ n = 0, 1, \dots$$
(2.2)

is called the linearized equation of Eq. (2.1) about the equilibrium point \overline{x} , and the equation

$$\lambda^{k+1} - q_0 \lambda^k - \dots - q_{k-1} \lambda - q_k = 0 \tag{2.3}$$

is called the characteristic equation of Eq. (2.2) about \overline{x} .

Theorem 2.1 (The Linearized Stability Theorem) Assume that the function f is a continuously differentiable function defined on some open neighborhood of an equilibrium point \overline{x} . Then the following statements are true:

- (a) When all the roots of characteristic equation (2.3) have absolute value less than one, then the equilibrium point x̄ of Eq.(2.1) is locally asymptotically stable.
- (b) If at least one root of characteristic equation (2.3) has absolute value greater than one, then the equilibrium point \overline{x} of Eq.(2.1) is unstable.
- (c) The equilibrium point x̄ of Eq.(2.1) is called hyperbolic if no root of characteristic equation (2.3) has absolute value equal to one. If there exists a root of characteristic equation (2.3) with absolute value equal to one, then the equilibrium x̄ is called nonhyperbolic.
- (d) An equilibrium point x̄ of Eq.(2.1) is called a repeller if all roots of characteristic equation (2.3) have absolute value greater than one.
- (e) An equilibrium point x̄ of Eq.(2.1) is called a saddle if one of the roots of characteristic equation (2.3) is greater and another is less than one in absolute value.

The following two theorems state necessary and sufficient conditions for all the roots of a real polynomial of degree two or three, respectively, to have modulus less than one.

Theorem 2.2 ([1], p.6) Assume that a_1 and a_0 are real numbers. Then a necessary and sufficient condition for all roots of the equation

$$\lambda^2 + a_1\lambda + a_0 = 0$$

to lie inside the unit disk is

 $|a_1| < 1 + a_0 < 2.$

Theorem 2.3 ([1], p.6) Assume that a_2 , a_1 , and a_0 are real numbers. Then a necessary and sufficient condition for all roots of the equation

$$\lambda^3 + a_2\lambda^2 + a_1\lambda + a_0 = 0$$

to lie inside the unit disk is

 $|a_2 + a_0| < 1 + a_1, \quad |a_2 - 3a_0| < 3 - a_1 \quad and \quad a_0^2 + a_1 - a_0a_2 < 1.$

Theorem 2.4 (Clark Theorem) ([1], p.6) Assume that $q_0, q_1, ..., q_k$ are real numbers such that

 $|q_0| + |q_1| + \dots + |q_k| < 1$

Then all roots of Eq.(2.3) lie inside the unit disk.

Let us introduce the discrete dynamical system:

$$\begin{aligned} x_{n+1} &= f_1 \left(x_n, x_{n-1}, \dots, x_{n-k}, y_n, y_{n-1}, \dots, y_{n-k}, z_n, z_{n-1}, \dots, z_{n-k} \right), \\ y_{n+1} &= f_2 \left(x_n, x_{n-1}, \dots, x_{n-k}, y_n, y_{n-1}, \dots, y_{n-k}, z_n, z_{n-1}, \dots, z_{n-k} \right), \\ z_{n+1} &= f_3 \left(x_n, x_{n-1}, \dots, x_{n-k}, y_n, y_{n-1}, \dots, y_{n-k}, z_n, z_{n-1}, \dots, z_{n-k} \right), \end{aligned}$$
(2.4)

 $n \in \mathbb{N}$, where $f_1 : I_1^{k+1} \times I_2^{k+1} \times I_3^{k+1} \to I_1$, $f_2 : I_1^{k+1} \times I_2^{k+1} \times I_3^{k+1} \to I_2$ and $f_3 : I_1^{k+1} \times I_2^{k+1} \times I_3^{k+1} \to I_3$ are continuously differentiable functions and I_1, I_2, I_3 are some intervals of real numbers. Also, a solution $\{(x_n, y_n, z_n)\}_{n=-k}^{\infty}$ of system (2.4) is uniquely determined by initial values $(x_{-i}, y_{-i}, z_{-i}) \in I_1 \times I_2 \times I_3$ for $i \in \{0, 1, ..., k\}$.

Definition 2.4 An equilibrium point of system (2.4) is a point $(\overline{x}, \overline{y}, \overline{z})$ that satisfies

$$\begin{split} \overline{x} &= f_1 \left(\overline{x}, \overline{x}, ..., \overline{x}, \overline{y}, \overline{y}, ..., \overline{y}, \overline{z}, \overline{z}, ..., \overline{z} \right), \\ \overline{y} &= f_2 \left(\overline{x}, \overline{x}, ..., \overline{x}, \overline{y}, \overline{y}, ..., \overline{y}, \overline{z}, \overline{z}, ..., \overline{z} \right), \\ \overline{z} &= f_3 \left(\overline{x}, \overline{x}, ..., \overline{x}, \overline{y}, \overline{y}, ..., \overline{y}, \overline{z}, \overline{z}, ..., \overline{z} \right). \end{split}$$

Together with system (2.4), if we consider the associated vector map

$$F = (f_1, x_n, x_{n-1}, \dots, x_{n-k}, f_2, y_n, y_{n-1}, \dots, y_{n-k}, f_3, z_{n-1}, \dots, z_{n-k}),$$

then the point $(\overline{x}, \overline{y}, \overline{z})$ is also called a fixed point of the vector map F.

Definition 2.5 Let $(\overline{x}, \overline{y}, \overline{z})$ be an equilibrium point of system (2.4).

(a) An equilibrium point (x̄, ȳ, z̄) is called stable if, for every ε > 0; there exists δ > 0 such that for every initial value (x_{-i}, y_{-i}, z_{-i}) ∈ I₁ × I₂ × I₃, with

$$\sum_{i=-k}^{0} |x_i - \overline{x}| < \delta, \quad \sum_{i=-k}^{0} |y_i - \overline{y}| < \delta, \quad \sum_{i=-k}^{0} |z_i - \overline{z}| < \delta$$

implying $|x_n - \overline{x}| < \varepsilon$, $|y_n - \overline{y}| < \varepsilon$, $|z_n - \overline{z}| < \varepsilon$ for $n \in \mathbb{N}$.

- (b) If an equilibrium point $(\overline{x}, \overline{y}, \overline{z})$ of system (2.4) is called unstable if it is not stable.
- (c) An equilibrium point (x̄, ȳ, z̄) of system (2.4) is called locally asymptotically stable if, it is stable, and if in addition there exists γ > 0 such that

$$\sum_{i=-k}^{0} |x_i - \overline{x}| < \gamma, \quad \sum_{i=-k}^{0} |y_i - \overline{y}| < \gamma, \quad \sum_{i=-k}^{0} |z_i - \overline{z}| < \gamma$$

and $(x_n, y_n, z_n) \to (\overline{x}, \overline{y}, \overline{z})$ as $n \to \infty$.

- (d) An equilibrium point $(\overline{x}, \overline{y}, \overline{z})$ of system (2.4) is called a global attractor if, $(x_n, y_n, z_n) \rightarrow (\overline{x}, \overline{y}, \overline{z})$ as $n \rightarrow \infty$.
- (e) An equilibrium point (\$\overline{x}\$, \$\overline{y}\$, \$\overline{z}\$) of system (2.4) is called globally asymptotically stable if it is stable, and a global attractor.

Definition 2.6 Let $(\overline{x}, \overline{y}, \overline{z})$ be an equilibrium point of the map F where f_1 , f_2 and f_3 are continuously differentiable functions at $(\overline{x}, \overline{y}, \overline{z})$. The linearized system of system (2.4) about the equilibrium point $(\overline{x}, \overline{y}, \overline{z})$ is

$$X_{n+1} = F\left(X_n\right) = BX_n,$$

where

$$X_{n} = (x_{n}, \dots, x_{n-k}, y_{n}, \dots, y_{n-k}, z_{n}, \dots, z_{n-k})^{T}$$

and B is a Jacobian matrix of system (2.4) about the equilibrium point $(\overline{x}, \overline{y}, \overline{z})$.

Theorem 2.5 (The Linearized Stability Theorem) Assume that

 $X_{n+1} = F(X_n), n = 0, 1, ...,$

be a system of difference equations such that \overline{X} is a fixed point of F.

- (a) If all eigenvalues of the Jacobian matrix B about X lie inside the open unit disk |λ| < 1, that is, if all of them have absolute value less than one, then X is locally asymptotically stable.
- (b) If at least one of them has a modulus greater than one, then \overline{X} is unstable.

Corollary 2.1 Assume that

$$X_{n+1} = F(X_n), n = 0, 1, \dots$$

be a system of difference equations such that \overline{X} is a fixed point of F. If no eigenvalues of the Jacobian matrix B about \overline{X} have absolute value equal to one, then \overline{X} is called hyperbolic. If there exists an eigenvalue of the Jacobian matrix B about \overline{X} with absolute value equal to one, then \overline{X} is called nonhyperbolic.

The so-called Schur-Cohn criterion provides necessary and sufficient conditions for all roots of the equation

$$P(\lambda) = a_0 \lambda^n + a_1 \lambda^{n-1} + \dots + a_{n-1} \lambda + a_n = 0$$
(2.5)

with real coefficients to lie in the open disk $|\lambda| < 1$.

Before we can explain the Schur-Cohn criterion, we need the so-called Routh-Hurwitz criterion.

Theorem 2.6 (Routh-Hurwitz criterion) Assume that

$$X_{n+1} = F(X_n), \ n = 0, 1, ...,$$

is a system of difference equations and \overline{X} is a fixed point of F, the characteristic polynomial of this system about the equilibrium point \overline{X} is given by (2.5) with real coefficients and $a_0 > 0$. Then all roots of the polynomial $P(\lambda)$ lie inside the open unit disk $|\lambda| < 1$ if and only if

 $\Delta_k > 0 \text{ for } k = 1, 2, ..., n$

where Δ_k is the principal minor of order k of the $n \times n$ matrix

$$\Delta_n = \begin{pmatrix} a_1 & a_3 & a_5 & \cdots & 0 \\ a_0 & a_2 & a_4 & \cdots & 0 \\ 0 & a_1 & a_3 & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \cdots & a_n \end{pmatrix}$$

Theorem 2.7 (Schur-Cohn criterion) The equation

$$P(\lambda) = \alpha_0 \lambda^n + \alpha_1 \lambda^{n-1} + \ldots + \alpha_{n-1} \lambda + \alpha_n = 0$$

has all its roots in the open unit disk $|\lambda| < 1$ if and only if the equation

$$P\left(\frac{z+1}{z-1}\right) = 0$$

has all its roots in the left-half plane

$$\operatorname{Re}(z) < 0.$$

Definition 2.7 A solution $\{(x_n, y_n, z_n)\}_{n=-k}^{\infty}$ of system (2.4) is bounded and persists if there exist positive constants M, N such that

 $M \le x_n, y_n, z_n \le N, \quad n = -m, -m + 1, \dots$

Definition 2.8 A solution $\{(x_n, y_n, z_n)\}_{n=-k}^{\infty}$ of system (2.4) is periodic with period p if

 $x_{n+p} = x_n, y_{n+p} = y_n, z_{n+p} = z_n, \text{ for all } n \ge -1.$

Definition 2.9 Let $(\overline{x}, \overline{y}, \overline{z})$ be an equilibrium point of system (2.4), and assume that $\{(x_n, y_n, z_n)\}_{n=-k}^{\infty}$ is a solution of the system (2.4).

A "string" of consecutive terms $\{x_{s}...,x_{m}\}$ (resp. $\{y_{s}...,y_{m}\}$, $\{z_{s}...,z_{m}\}$), $s \geq -1$, $m \leq \infty$ is said to be a positive semicycle if $x_{i} \geq \overline{x}$ (resp. $y_{i} \geq \overline{y}$, $z_{i} \geq \overline{z}$), $i \in \{s,...,m\}$, $x_{s-1} < \overline{x}$ (resp. $y_{s-1} < \overline{y}$, $z_{s-1} < \overline{z}$), and $x_{m+1} < \overline{x}$ (resp. $y_{m+1} < \overline{y}$, $z_{m+1} < \overline{z}$).

A "string" of consecutive terms $\{x_{s}..., x_{m}\}$ (resp. $\{y_{s}..., y_{m}\}$, $\{z_{s}..., z_{m}\}$), $s \geq -1$, $m \leq \infty$ is said to be a negative semicycle if $x_{i} < \overline{x}$ (resp. $y_{i} < \overline{y}, z_{i} < \overline{z}$), $i \in \{s, ..., m\}$, $x_{s-1} \geq \overline{x}$ (resp. $y_{s-1} \geq \overline{y}, z_{s-1} \geq \overline{z}$), and $x_{m+1} \geq \overline{x}$ (resp. $y_{m+1} \geq \overline{y}, z_{m+1} \geq \overline{z}$).

A "string" of consecutive terms $\{(x_s, y_s, z_s), ..., (x_m, y_m, z_m)\}$ is said to be a positive (resp. negative) semicycle if $\{x_s, ..., x_m\}$, $\{y_s, ..., y_m\}$, $\{z_s, ..., z_m\}$ are positive (resp. negative) semicycles.

Definition 2.10 A solution $\{(x_n, y_n, z_n)\}_{n=-k}^{\infty}$ of system (2.4) is called nonoscillatory about $(\overline{x}, \overline{y}, \overline{z})$, or simply nonoscillatory, if there exists $N \ge -k$ such that either

 $x_n \ge \overline{x}, y_n \ge \overline{y}, z_n \ge \overline{z}, \text{ for all } n \ge N$

or

 $x_n < \overline{x}, y_n < \overline{y}, z_n < \overline{z}, \text{ for all } n \ge N.$

Otherwise, the solution $\{(x_n, y_n, z_n)\}_{n=-k}^{\infty}$ is called oscillatory about $(\overline{x}, \overline{y}, \overline{z})$, or simply oscillatory.

Theorem 2.8 (Rouche's Theorem) ([105], **p.365**) Let C be a simple closed contour lying entirely with a domain D. Suppose f and g are analytic in D. If the strict inequality |f(z) - g(z)| < |f(z)| holds for all z on C, then f and g have the same number of zeros (counted according to their order or multiplicities) inside C.

Now, we give some notifications about centre manifold theorem see [106-113].

Centre manifold theory [106] may be utilized to refer to the stability of non-hyberbolic fixed points. A centre manifold is a set M_c in a lower dimensional space where the dynamics of the original systems can be derived by examining the dynamics on M_c . Regard the m-parameter map $F(m, u), F : \mathbb{R}^s \times \mathbb{R}^k \to \mathbb{R}^k$, where $m \in \mathbb{R}^s$ is a parameter and $u \in \mathbb{R}^k$. Let $F(m, u^*(m)) = u^*(m)$ be a fixed point of F. It is notice that the stability of the hyperbolic fixed points of F is established from the stability of the fixed points under the linear map $J = D_u F(m, u^*(m))$.

Centre manifold theory make use of when one of the eigenvalues lies on the unit circle and the other eigenvalues are inside the unit circle.

Assuming, without loss of generality that $u^* = 0_k = (0, 0, ..., 0)$ the k-dimensional zero vector, the map F can be written in the form

$$x \longmapsto Ax + f(x, y)$$

$$y \longmapsto Bx + g(x, y)$$

$$(2.6)$$

where J on (2.6) has the form

$$J = \left[\begin{array}{cc} A & 0 \\ 0 & B \end{array} \right].$$

Notice that all of the eigenvalues of A lie on the unit circle and all of the eigenvalues of B are off the unit circle. Hereby, A is a $t \times t$ matrix and B is an $s \times s$ matrix, with t + s = k. The following theorem alleges the existence of a (non-unique) centre manifold (a curve y = h(x)) on which the dynamics of system (2.6) is provided with the map on the centre manifold.

Theorem 2.9 There is a C^r centre manifold for system (2.6) that can be represented locally as

 $M_{c} = \left\{ (x, y) \in \mathbb{R}^{t} \times \mathbb{R}^{s} : y = h(x), \|x\| < \delta, h(0) = 0, Dh(0) = 0 \right\}.$

Furthermore, the dynamics restricted to M_c are given locally by the map

 $x \longmapsto Ax + f(x, h(x)), x \in \mathbb{R}^t.$

The following theorem indicates that the dynamics on the centre manifold M_c determines the dynamics on (2.6).

Theorem 2.10 Suppose that the t-dimensional zero vector $0_t = (0, ..., 0)$ is a fixed point for the map $x \mapsto Ax + f(x, h(x)), x \in \mathbb{R}^t$. If 0_t is stable, asymptotically stable, or unstable, then the fixed point 0_k of system (2.6) is stable, asymptotically stable, or unstable, respectively.

2.2 INTEGER SEQUENCES

Now, we give information about integer sequences that establish a large part of our study. Some properties of the above sequences were studied in various papers, see [114-120].

2.2.1 Fibonacci Numbers

The Fibonacci sequence is defined by

$$F_n = F_{n-1} + F_{n-2}, \quad n \ge 2 \tag{2.7}$$

with initial conditions $F_0 = 0$, $F_1 = 1$. Also, it is obtained to extend the Fibonacci sequence backward as

$$F_{-n} = (-1)^{n+1} F_n$$

The characteristic equation of (2.7) is $x^2 - x - 1 = 0$ such that the roots

$$\alpha = \frac{1+\sqrt{5}}{2}$$
 (golden ratio) and $\beta = \frac{1-\sqrt{5}}{2}$.

Also, there exists the following limit

$$\lim_{n \to \infty} \frac{F_{n+1}}{F_n} = \alpha,$$

where F_n is *n*th Fibonacci number.

2.2.2 Padovan Numbers

The Padovan sequence is defined by

$$P_n = P_{n-2} + P_{n-3}, \quad n \in \mathbb{N} \tag{2.8}$$

with initial conditions $P_{-2} = 0$, $P_{-1} = 0$, $P_0 = 1$.

The characteristic equation of (2.8) is $x^3 - x - 1 = 0$ such that the roots

$$p = \frac{r^{2} + 12}{6r}$$

$$q = -\frac{r^{2} + 12}{6r} - i\frac{\sqrt{3}}{2}\left(\frac{r}{6} - \frac{2}{3r}\right)$$

$$t = -\frac{r^{2} + 12}{6r} + i\frac{\sqrt{3}}{2}\left(\frac{r}{6} - \frac{2}{3r}\right)$$

where $r = \sqrt[3]{108 + 12\sqrt{69}}$ and the unique real root is p named as plastic number. Also, there exists the following limit

$$\lim_{n \to \infty} \frac{P_{n+1}}{P_n} = p$$

where P_n is *n*th Padovan number.

2.2.3 Horadam Numbers

Horadam sequence, a generalization of Fibonacci sequence, $(W_n(a, b; p, q))_{n \ge 0}$ or simply $(W_n)_{n \ge 0}$ is defined by

$$W_n = pW_{n-1} + qW_{n-2}, \quad W_0 = a, W_1 = b, \qquad n \ge 2,$$
(2.9)

where a, b, p and q are arbitrary real numbers.

The characteristic equation of (2.9) is $x^2 - px - q = 0$ such that the roots

$$\lambda = \frac{p + \sqrt{p^2 + 4q}}{2}$$
 and $\mu = \frac{p - \sqrt{p^2 + 4q}}{2}$.

Also, there exists the following limit

$$\lim_{n \to \infty} \frac{W_{n+1}}{W_n} = \lambda,$$

where W_n is *n*th Horadam number.

2.2.4 Generalized Padovan Numbers

The generalized Padovan sequence, an extension of the padovan sequence, is defined by

$$S_n = pS_{n-2} + qS_{n-3}, \quad n \in \mathbb{N}$$

$$(2.10)$$

with initial conditions $S_{-2} = 0$, $S_{-1} = 0$, $S_0 = 1$, where p and q are arbitrary real numbers.

The characteristic equation of (2.10) is $x^3 - px - q = 0$ such that the roots

$$\phi = \frac{R^2 + 12p}{6R}$$

$$\varphi = -\frac{R^2 + 12p}{12R} + i\frac{\sqrt{3}}{2}\left(\frac{R}{6} - \frac{2p}{R}\right)$$

$$\psi = -\frac{R^2 + 12p}{12R} - i\frac{\sqrt{3}}{2}\left(\frac{R}{6} - \frac{2p}{R}\right)$$

where $R = \sqrt[3]{108q + 12\sqrt{-12p^3 + 81q^2}}$. Also, there exists the following limit

$$\lim_{n \to \infty} \frac{S_{n+1}}{S_n} = \phi,$$

where S_n is *n*th generalized Padovan number. Also, the other integer sequences are as follows:

• Lucas sequence is defined by

$$L_n = L_{n-1} + L_{n-2}, \quad L_0 = 2, \ L_1 = 1,$$

• Pell sequence is defined by

$$P_n = 2P_{n-1} + P_{n-2}, \quad P_0 = 0, \ P_1 = 1,$$

• Pell-Lucas sequence is defined by

$$P_n = 2P_{n-1} + P_{n-2}, \quad P_0 = 2, \ P_1 = 2,$$

• Jacobsthal sequence is defined by

$$J_n = J_{n-1} + 2J_{n-2}, \quad J_0 = 0, \ J_1 = 1,$$

• Jacobsthal-Lucas sequence is defined by

$$j_n = j_{n-1} + 2j_{n-2}, \quad j_0 = 2, \ j_1 = 1,$$

• Perrin sequence is defined by

$$Q_n = Q_{n-2} + Q_{n-3}, \quad Q_0 = 3, Q_1 = 0, Q_2 = 2.$$

Now, we give information about Tribonacci numbers that we afterwards need in the paper.

2.2.5 Tribonacci Numbers

The Tribonacci sequence $\{T_n\}_{n=0}^{\infty}$ is defined by the third-order recurrence relations

$$T_{n+3} = T_{n+2} + T_{n+1} + T_n, (2.11)$$

with initial conditions $T_0 = 0$, $T_1 = 1$, $T_2 = 1$. Also, it can be extended the Tribonacci sequence backward (negative subscripts) as

$$T_{-n} = T_{-n+3} - T_{-n+2} - T_{-n+1}.$$
(2.12)

It can be clearly obtained that the characteristic equation of (2.11) has the form

$$x^3 - x^2 - x - 1 = 0 \tag{2.13}$$

such that the roots

$$\alpha = \frac{1 + \sqrt[3]{19 + 3\sqrt{33} + \sqrt[3]{19 - 3\sqrt{33}}}}{3}$$

$$\beta = \frac{1 + \omega\sqrt[3]{19 + 3\sqrt{33} + \omega^2\sqrt[3]{19 - 3\sqrt{33}}}}{3}$$

$$\gamma = \frac{1 + \omega^2\sqrt[3]{19 + 3\sqrt{33} + \omega\sqrt[3]{19 - 3\sqrt{33}}}}{3}$$

where α is called Tribonacci constant and

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. Therefore, Tribonacci sequence can be expressed using Binet formula

$$T_n = \frac{\alpha^{n+1}}{(\alpha - \beta)(\alpha - \gamma)} + \frac{\beta^{n+1}}{(\beta - \alpha)(\beta - \gamma)} + \frac{\gamma^{n+1}}{(\gamma - \alpha)(\gamma - \beta)}.$$

Furthermore, there exist the following limit

$$\lim_{n \to \infty} \frac{T_{n+r}}{T_n} = \alpha^r, \tag{2.14}$$

where $r \in \mathbb{Z}$ and T_n is the *n*th Tribonacci number.

2.2.6 Generalized Tribonacci Numbers

The generalized Tribonacci sequence in [121] $\{V_n\}_{n\geq 0}$ is defined as follows:

$$V_n = rV_{n-1} + sV_{n-2} + tV_{n-3}, \quad V_0 = a, V_1 = b, V_2 = c, \quad n \ge 3$$
(2.15)

where a, b, c are arbitrary integers and r, s, t are real numbers.

The sequence $\{V_n\}_{n\geq 0}$ can be expanded to negative subscripts by defining

$$V_{-n} = -\frac{s}{t}V_{-(n-1)} - \frac{r}{t}V_{-(n-2)} + \frac{1}{t}V_{-(n-3)}$$

for n = 1, 2, 3, ... when $t \neq 0$. Thus, recurrence (2.15) holds for all integer n.

If we set r = s = t = 1 and $V_0 = 0, V_1 = 1, V_2 = 1$ then $\{V_n\}_{n \ge 0}$ is the well-known Tribonacci sequence and if we set r = s = t = 1 and $V_0 = 3, V_1 = 1, V_2 = 3$ then $\{V_n\}_{n \ge 0}$ is the well-known Tribonacci-Lucas sequence.

Actually, the generalized Tribonacci sequence is the generalization of the renowned sequences like Tribonacci, Tribonacci-Lucas, Padovan (Cordonnier), Perrin, Padovan-Perrin, Narayana, third order Jacobsthal and third order Jacobsthal-Lucas. In literature, for instance, the following names and notations (see Table 1) are used for the special cases of r, s, t and initial values.

Sequences (Numbers)	Notation
Tribonacci	$\{T_n\} = \{V_n(0, 1, 1; 1, 1, 1)\}$
Tribonacci-Lucas	$\{K_n\} = \{V_n(3, 1, 3; 1, 1, 1)\}$
Padovan (Cordonnier)	$\{P_n\} = \{V_n(1, 1, 1; 0, 1, 1)\}$
Pell-Padovan	$\{R_n\} = \{V_n(1, 1, 1; 0, 2, 1)\}$
Jacobsthal-Padovan	${JP_n} = {V_n(1, 1, 1; 0, 1, 2)}$
Perrin	$\{Q_n\} = \{V_n(3,0,2;0,1,1)\}$
Pell-Perrin	$\{pQ_n\} = \{V_n(3,0,2;0,2,1)\}$
Jacobsthal-Perrin	${JQ_n} = {V_n(3, 0, 2; 0, 1, 2)}$
Padovan-Perrin	$\{S_n\} = \{V_n(0,0,1;0,1,1)\}$
Narayana	$\{N_n\} = \{V_n(0, 1, 1; 1, 0, 1)\}$
third order Jacobsthal	$\{J_n\} = \{V_n(0, 1, 1; 1, 1, 2)\}$
third order Jacobsthal-Lucas	${j_n} = {V_n(2, 1, 5; 1, 1, 2)}$

Table 2.1 A few values of generalized Tribonacci sequences.

As $\{V_n\}_{n\geq 0}$ is a third order recurrence sequence (difference equation), it's characteristic equation is $x^3 - rx^2 - sx - t = 0$, whose roots are

$$\alpha = \alpha(r, s, t) = \frac{r}{3} + A + B$$

$$\beta = \beta(r, s, t) = \frac{r}{3} + \omega A + \omega^2 B$$

$$\gamma = \gamma(r, s, t) = \frac{r}{3} + \omega^2 A + \omega B$$

where

$$A = \left(\frac{r^3}{27} + \frac{rs}{6} + \frac{t}{2} + \sqrt{\Delta}\right)^{1/3}, \ B = \left(\frac{r^3}{27} + \frac{rs}{6} + \frac{t}{2} - \sqrt{\Delta}\right)^{1/3}$$
$$\Delta = \Delta(r, s, t) = \frac{r^3 t}{27} - \frac{r^2 s^2}{108} + \frac{rst}{6} - \frac{s^3}{27} + \frac{t^2}{4},$$
$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp(2\pi i/3).$$

Note that we obtain the following identities

$$\begin{aligned} \alpha + \beta + \gamma &= r, \\ \alpha \beta + \alpha \gamma + \beta \gamma &= -s, \\ \alpha \beta \gamma &= t. \end{aligned}$$

From now on, we assume that $\Delta(r, s, t) > 0$, so that the Eq.(2.15) has one real (α) and two non-real solutions with the latter being conjugate complex. Therefore, in this case, it is known that generalized Tribonacci numbers can be declared, for all integers n, using Binet's formula

$$V_n = \frac{P\alpha^n}{(\alpha - \beta)(\alpha - \gamma)} + \frac{Q\beta^n}{(\beta - \alpha)(\beta - \gamma)} + \frac{R\gamma^n}{(\gamma - \alpha)(\gamma - \beta)}$$
(2.16)
where

where

$$P = V_2 - (\beta + \gamma)V_1 + \beta\gamma V_0, \ Q = V_2 - (\alpha + \gamma)V_1 + \alpha\gamma V_0, \ R = V_2 - (\alpha + \beta)V_1 + \alpha\beta V_0.$$

Notice that the Binet form of a sequence satisfying (2.15) for non-negative integers is valid for all integers n, for a proof of this result see [122]. This result of Howard and Saidak [122] is even true in the case of higher-order recurrence relations.

We can present Binet's formula of the generalized Tribonacci numbers for the negative subscripts: for $n = 1, 2, 3, \dots$ we get

$$V_{-n} = \frac{\alpha^2 - r\alpha - s}{t} \frac{P\alpha^{1-n}}{(\alpha - \beta)(\alpha - \gamma)} + \frac{\beta^2 - r\beta - s}{t} \frac{Q\beta^{1-n}}{(\beta - \alpha)(\beta - \gamma)} + \frac{\gamma^2 - r\gamma - s}{t} \frac{R\gamma^{1-n}}{(\gamma - \alpha)(\gamma - \beta)}.$$

CHAPTER 3

GLOBAL BEHAVIOR OF SOLUTIONS OF A SYSTEM OF THREE-DIMENSIONAL NONLINEAR DIFFERENCE EQUATIONS

Firstly, we state that the results of this chapter are cited from [65] which has been published by us.

In this chapter, we investigate the stability, boundedness character and periodicity of positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{x_{n-1}}{z_n}, \quad y_{n+1} = A + \frac{y_{n-1}}{z_n}, \quad z_{n+1} = A + \frac{z_{n-1}}{y_n}, \quad n = 0, 1, \dots,$$
(3.1)

where the parameter A and the initial values x_{-1} , x_0 , y_{-1} , y_0 , z_{-1} , z_0 are positive real numbers.

3.1 MAIN RESULTS

In this section, we prove our main results.

Theorem 3.1 The following statements are true:

(i) If $(\overline{x}, \overline{y}, \overline{z})$ is a positive equilibrium point of system (3.1), then

$$(\overline{x},\overline{y},\overline{z}) = \begin{cases} (A+1,A+1,A+1), & \text{if } A \neq 1, \\ \left(\mu,\mu,\frac{\mu}{\mu-1}\right), \ \mu \in (1,\infty) & \text{if } A = 1. \end{cases}$$

(ii) If A > 1, then the equilibrium point of system (3.1) is locally asymptotically stable.

- (iii) If 0 < A < 1, then the equilibrium point of system (3.1) is locally unstable.
- (iv) If A = 1, then for every $\mu \in (1, \infty)$ there exist positive solutions $\{(x_n, y_n, z_n)\}$ of system (3.1) which tend to the positive equilibrium point $(\mu, \mu, \frac{\mu}{\mu-1})$.

Proof. (i) It is easily seen from the definition of equilibrium point that the equilibrium points of system (3.1) are the nonnegative solution of the equations

$$\overline{x} = A + \frac{\overline{x}}{\overline{z}}, \ \overline{y} = A + \frac{\overline{y}}{\overline{z}}, \ \overline{z} = A + \frac{\overline{z}}{\overline{y}}$$

From this, we get

$$\begin{aligned} \overline{x}.\overline{z} &= A\overline{z} + \overline{x}, \ \overline{y}.\overline{z} = A\overline{z} + \overline{y}, \ \overline{z}.\overline{y} = A\overline{y} + \overline{z} \\ \Rightarrow \overline{x}.\overline{z} - \overline{x} = \overline{y}.\overline{z} - \overline{y}, \ A\overline{z} + \overline{y} = A\overline{y} + \overline{z} \\ \Rightarrow \overline{x}\left(\overline{z} - 1\right) = \overline{y}\left(\overline{z} - 1\right), \ \overline{z}\left(A - 1\right) = \overline{y}\left(A - 1\right). \end{aligned}$$

From which it follows that if $A \neq 1$,

$$\overline{x} = \overline{y} = \overline{z} = A + 1 \Rightarrow (\overline{x}, \overline{y}, \overline{z}) = (A + 1, A + 1, A + 1).$$

Also, we have

$$\begin{split} \overline{\overline{x}.\overline{z}-\overline{x}} &= A, \ \overline{\overline{y}.\overline{z}-\overline{y}} = A, \ \overline{\overline{z}}.\overline{\overline{y}}-\overline{\overline{z}} = A\\ \Rightarrow \ \overline{\overline{x}.\overline{z}}-\overline{\overline{x}} = \overline{\overline{y}.\overline{z}-\overline{y}}, \ \overline{\overline{y}}.\overline{\overline{z}}-\overline{\overline{y}} = \overline{\overline{z}}.\overline{\overline{y}}-\overline{\overline{z}}\\ \Rightarrow \ \overline{x}.\overline{z}-\overline{x} = \overline{y}.\overline{z}-\overline{y}, \ \overline{y}^2\overline{z}-\overline{y}^2 = \overline{z}^2\overline{y}-\overline{z}^2\\ \Rightarrow \ \overline{x}(\overline{z}-1) = \overline{y}(\overline{z}-1), \ \overline{y}.\overline{z}(\overline{y}-\overline{z}) = (\overline{y}-\overline{z})(\overline{y}+\overline{z}). \end{split}$$

From which it follows that if A = 1,

$$\overline{x} = \overline{y} \text{ and } \overline{y}.\overline{z} = \overline{y} + \overline{z} \Rightarrow (\overline{x},\overline{y},\overline{z}) = \left(\mu,\mu,\frac{\mu}{\mu-1}\right), \ \mu \in (1,\infty)$$

In that case, we have a continuous of positive equilibriums which lie on the hyperboloid

$$\overline{y}.\overline{z} = \overline{y} + \overline{z}.\tag{3.2}$$

(ii) We consider the following transformation to build the corresponding linearized form of system (3.1):

$$(x_n, x_{n-1}, y_n, y_{n-1}, z_n, z_{n-1}) \rightarrow (f, f_1, g, g_1, h, h_1)$$

where

$$f = A + \frac{x_{n-1}}{z_n}$$

$$f_1 = x_n$$

$$g = A + \frac{y_{n-1}}{z_n}$$

$$g_1 = y_n$$

$$h = A + \frac{z_{n-1}}{y_n}$$

$$h_1 = z_n.$$

The Jacobian matrix about the equilibrium point $(\overline{x}, \overline{y}, \overline{z})$ under the above transformation is given by

$$B\left(\overline{x}, \overline{y}, \overline{z}\right) = \begin{pmatrix} 0 & \frac{1}{\overline{z}} & 0 & 0 & -\frac{\overline{x}}{\overline{z}^2} & 0\\ 1 & 0 & 0 & 0 & 0\\ 0 & 0 & 0 & \frac{1}{\overline{z}} & -\frac{\overline{y}}{\overline{z}^2} & 0\\ 0 & 0 & 1 & 0 & 0 & 0\\ 0 & 0 & -\frac{\overline{z}}{\overline{y}^2} & 0 & 0 & \frac{1}{\overline{y}}\\ 0 & 0 & 0 & 0 & 1 & 0 \end{pmatrix}.$$
 (3.3)

Hence, the linearized system of system (3.1) about the equilibrium point

$$(\overline{x},\overline{y},\overline{z}) = (A+1,A+1,A+1)$$

is

$$X_{n+1} = B\left(\overline{x}, \overline{y}, \overline{z}\right) X_n,$$

where

$$X_n = ((x_n, x_{n-1}, y_n, y_{n-1}, z_n, z_{n-1}))^T$$

and

$$B\left(\overline{x},\overline{y},\overline{z}\right) = \begin{pmatrix} 0 & \frac{1}{A+1} & 0 & 0 & -\frac{1}{A+1} & 0\\ 1 & 0 & 0 & 0 & 0 & 0\\ 0 & 0 & 0 & \frac{1}{A+1} & -\frac{1}{A+1} & 0\\ 0 & 0 & 1 & 0 & 0 & 0\\ 0 & 0 & -\frac{1}{A+1} & 0 & 0 & \frac{1}{A+1}\\ 0 & 0 & 0 & 0 & 1 & 0 \end{pmatrix}$$

.
Then, the characteristic equation of $B\left(\overline{x}, \overline{y}, \overline{z}\right)$ about

$$(\overline{x},\overline{y},\overline{z}) = (A+1,A+1,A+1)$$

is

$$\lambda^{6} - \frac{(3A+4)}{(A+1)^{2}}\lambda^{4} + \frac{(3A+4)}{(A+1)^{3}}\lambda^{2} - \frac{1}{(A+1)^{3}} = 0.$$
(3.4)

From this, the roots of characteristic equation (3.4) are

$$\lambda_{1} = \frac{1}{\sqrt{A+1}},$$

$$\lambda_{2} = -\frac{1}{\sqrt{A+1}},$$

$$\lambda_{3} = \frac{1}{2} \frac{\sqrt{4A+5}-1}{A+1},$$

$$\lambda_{4} = -\frac{1}{2} \frac{\sqrt{4A+5}+1}{A+1},$$

$$\lambda_{5} = \frac{1}{2} \frac{\sqrt{4A+5}+1}{A+1},$$

$$\lambda_{6} = -\frac{1}{2} \frac{\sqrt{4A+5}-1}{A+1}.$$

From the Linearized Stability Theorem, since A > 1, all roots of the characterictic equation lie inside the open unit disk $|\lambda| < 1$. Therefore, the positive equilibrium point of system (3.1) is locally asymptotically stable.

- (iii) From the proof of (ii), it is true.
- (iv) From (3.3), the linearized system of system (3.1) about the equilibrium point

$$(\overline{x},\overline{y},\overline{z}) = \left(\mu,\mu,\frac{\mu}{\mu-1}\right)$$

is

$$X_{n+1} = B\left(\overline{x}, \overline{y}, \overline{z}\right) X_n,$$

where

$$X_n = ((x_n, x_{n-1}, y_n, y_{n-1}, z_n, z_{n-1}))^T$$

and

$$B\left(\overline{x},\overline{y},\overline{z}\right) = \begin{pmatrix} 0 & \frac{\mu-1}{\mu} & 0 & 0 & -\frac{(\mu-1)^2}{\mu} & 0\\ 1 & 0 & 0 & 0 & 0\\ 0 & 0 & 0 & \frac{\mu-1}{\mu} & -\frac{(\mu-1)^2}{\mu} & 0\\ 0 & 0 & 1 & 0 & 0 & 0\\ 0 & 0 & -\frac{1}{\mu(\mu-1)} & 0 & 0 & \frac{1}{\mu}\\ 0 & 0 & 0 & 0 & 1 & 0 \end{pmatrix}.$$

Hence, the characteristic equation of the matrix B is

$$\lambda^{6} - \left(\frac{2\mu^{2} - 1}{\mu^{2}}\right)\lambda^{4} + \left(\frac{\mu^{3} + \mu^{2} - 3\mu + 1}{\mu}\right)\lambda^{2} - \frac{(\mu - 1)^{2}}{\mu^{3}} = 0.$$
(3.5)

Therefore, the roots of the equation (3.5) are:

$$\lambda_{1} = -1,$$

$$\lambda_{2} = 1,$$

$$\lambda_{3} = \frac{\sqrt{\mu - 1}}{\mu},$$

$$\lambda_{4} = -\frac{\sqrt{\mu - 1}}{\mu},$$

$$\lambda_{5} = \sqrt{\frac{\mu - 1}{\mu}},$$

$$\lambda_{6} = -\sqrt{\frac{\mu - 1}{\mu}}.$$

Then, the modulus of four of the roots of (3.5) are less than 1. So, there exist positive solutions of system (3.1) which tend to the positive equilibrium point $\left(\mu, \mu, \frac{\mu}{\mu-1}\right)$ of system (3.1) (this follows from the following proposition). This completes the proof.

In the following proposition we find positive solutions of system (3.1) which tend to $(\overline{x}, \overline{y}, \overline{z})$ as $n \to \infty$.

Proposition 3.2 Let $\{(x_n, y_n, z_n)\}$ be a positive solution of system (3.1). Then, if there exists an $s \in \{-1, 0, ...\}$ such that for $n \ge s$, $x_n \ge \overline{x}$, $y_n \ge \overline{y}$, $z_n \ge \overline{z}$ (resp., $x_n < \overline{x}$, $y_n < \overline{y}$, $z_n < \overline{z}$), the solution $\{(x_n, y_n, z_n)\}$ tends to the positive equilibrium $(\overline{x}, \overline{y}, \overline{z})$ of system (3.1) as $n \to \infty$.

Proof. Let $\{(x_n, y_n, z_n)\}$ be a positive solution of system (3.1) such that

$$x_n \ge \overline{x}, \ y_n \ge \overline{y}, \ z_n \ge \overline{z}, \ n \ge s,$$

$$(3.6)$$

where $s \in \{-1, 0, ...\}$. Then from (3.1) and (3.6) we have

$$x_{n+1} = A + \frac{x_{n-1}}{z_n} \le A + \frac{x_{n-1}}{\overline{z}}, \quad n \ge 1.$$
(3.7)

 Set

$$u_{n+1} = A + \frac{u_{n-1}}{\bar{z}}, n \ge 1$$
(3.8)

such that

$$u_s = x_s, \, u_{s+1} = x_{s+1}, \, s \in \{-1, 0, 1, ...\}, \quad n \ge s.$$
(3.9)

Then, the solution u_n of the difference equation (3.8) is as follows:

$$u_n = c_1 \left(\frac{1}{\sqrt{\overline{z}}}\right)^n + c_1 \left(-\frac{1}{\sqrt{\overline{z}}}\right)^n + \frac{A\overline{z}}{A\overline{z} - 1} = c_1 \left(\frac{1}{\sqrt{\overline{z}}}\right)^n + c_1 \left(-\frac{1}{\sqrt{\overline{z}}}\right)^n + \overline{x},\tag{3.10}$$

where c_1 , c_2 depend on x_s , x_{s+1} . In addition, the relations (3.7) and (3.8) imply that

$$x_{n+1} - u_{n+1} \le \frac{x_{n-1} - u_{n-1}}{\overline{z}}, \quad n > s.$$
(3.11)

Then, by using (3.9) and (3.11) and induction, we have

$$x_n \le u_n, \quad n \ge s. \tag{3.12}$$

Therefore, from (3.6), (3.10), and (3.12), it is clear that

$$\lim_{n \to \infty} x_n = \overline{x}.$$
(3.13)

Similarly, we can prove that

$$\lim_{n \to \infty} y_n = \overline{y} \quad \text{and} \quad \lim_{n \to \infty} z_n = \overline{z}.$$
(3.14)

Thus, from (3.13) and (3.14), the solution $\{(x_n, y_n, z_n)\}$ tends to $(\overline{x}, \overline{y}, \overline{z})$ as $n \to \infty$. Arguing as above we can show that if $x_n < \overline{x}, y_n < \overline{y}, z_n < \overline{z}$ for $n \ge s$, then $\{(x_n, y_n, z_n)\}$ tends to $(\overline{x}, \overline{y}, \overline{z})$ as $n \to \infty$. The proof of the proposition is completed.

Theorem 3.3 Assume that 0 < A < 1 and $\{(x_n, y_n, z_n)\}$ is an arbitrary positive solution of system (3.1). Then, the following statements are true.

(i) If

$$x_{-1} < 1, \ y_{-1} < 1, \ z_{-1} < 1, \ x_0 > \frac{1}{1-A}, \ y_0 > \frac{1}{1-A}, \ z_0 > \frac{1}{1-A},$$
 (3.15) then

$$\lim_{n \to \infty} x_{2n+1} = A, \quad \lim_{n \to \infty} y_{2n+1} = A, \quad \lim_{n \to \infty} z_{2n+1} = A,$$
$$\lim_{n \to \infty} x_{2n} = \infty, \quad \lim_{n \to \infty} y_{2n} = \infty, \quad \lim_{n \to \infty} z_{2n} = \infty.$$

(ii) If

$$x_0 < 1, y_0 < 1, z_0 < 1, x_{-1} > \frac{1}{1-A}, y_{-1} > \frac{1}{1-A}, z_{-1} > \frac{1}{1-A}, (3.16)$$

then

 $\lim_{n \to \infty} x_{2n+1} = \infty, \lim_{n \to \infty} y_{2n+1} = \infty, \lim_{n \to \infty} z_{2n+1} = \infty,$ $\lim_{n \to \infty} x_{2n} = A, \lim_{n \to \infty} y_{2n} = A, \lim_{n \to \infty} z_{2n} = A.$

Proof. (i) From (3.1) and (3.15), we get

$$\begin{aligned} x_1 &= A + \frac{x_{-1}}{z_0} < A + \frac{1}{z_0} < A + (1 - A) = 1, \\ y_1 &= A + \frac{y_{-1}}{z_0} < A + \frac{1}{z_0} < A + (1 - A) = 1, \\ z_1 &= A + \frac{z_{-1}}{y_0} < A + \frac{1}{y_0} < A + (1 - A) = 1, \\ x_2 &= A + \frac{x_0}{z_1} > x_0 > \frac{1}{1 - A}, \\ y_2 &= A + \frac{y_0}{z_1} > y_0 > \frac{1}{1 - A}, \\ z_2 &= A + \frac{z_0}{y_1} > z_0 > \frac{1}{1 - A}. \end{aligned}$$

By induction for n = 0, 1, 2, ..., we obtain

$$\begin{aligned} x_{2n-1} &< 1, \ y_{2n-1} < 1, \ z_{2n-1} < 1, \\ x_{2n} &> \frac{1}{1-A}, \ y_{2n} > \frac{1}{1-A}, \ z_{2n} > \frac{1}{1-A}. \end{aligned}$$
(3.17)

Thus, relations (3.1) and (3.17) imply that

$$\begin{aligned} x_{2n} &= A + \frac{x_{2n-2}}{z_{2n-1}} > A + x_{2n-2} > 2A + \frac{x_{2n-4}}{z_{2n-3}} > 2A + x_{2n-4}, \\ y_{2n} &= A + \frac{y_{2n-2}}{z_{2n-1}} > A + y_{2n-2} > 2A + \frac{y_{2n-4}}{z_{2n-3}} > 2A + y_{2n-4}, \\ z_{2n} &= A + \frac{z_{2n-2}}{y_{2n-1}} > A + z_{2n-2} > 2A + \frac{z_{2n-4}}{y_{2n-3}} > 2A + z_{2n-4}. \end{aligned}$$

From which we get

 $\lim_{n \to \infty} x_{2n} = \infty, \quad \lim_{n \to \infty} y_{2n} = \infty, \quad \lim_{n \to \infty} z_{2n} = \infty.$

Noting that (3.17) and taking limits on both sides of three equations

$$x_{2n+1} = A + \frac{x_{2n-1}}{z_{2n}}, \ y_{2n+1} = A + \frac{y_{2n-1}}{z_{2n}}, \ \ z_{2n+1} = A + \frac{z_{2n-1}}{y_{2n}},$$

we have

 $\lim_{n \to \infty} x_{2n+1} = A, \quad \lim_{n \to \infty} y_{2n+1} = A, \quad \lim_{n \to \infty} z_{2n+1} = A.$

(ii) The proof is similar to the proof of (i), so we leave it to readers.

Theorem 3.4 Assume that A = 1. Then every positive solution of system (3.1) is bounded and persists.

Proof. Let $\{(x_n, y_n, z_n)\}$ be a positive solution of the system (3.1). Obviously, $x_n > 1$, $y_n > 1$, $z_n > 1$, for $n \ge 1$. So, we have

$$x_i, y_i, z_i \in \left[K, \frac{K}{K-1}\right], \ i = 1, 2, ..., m+1,$$

where

$$K = \min\left\{\alpha, \frac{\beta}{\beta - 1}\right\} > 1, \ \alpha = \min_{1 \le i \le m + 1} \left\{x_i, y_i, z_i\right\}, \ \beta = \max_{1 \le i \le m + 1} \left\{x_i, y_i, z_i\right\}.$$

Then, we obtain

$$K = 1 + \frac{K}{K/(K-1)} \le x_{m+2} = 1 + \frac{x_1}{z_{m+1}} \le 1 + \frac{K/(K-1)}{K} = \frac{K}{K-1},$$

$$K = 1 + \frac{K}{K/(K-1)} \le y_{m+2} = 1 + \frac{y_1}{z_{m+1}} \le 1 + \frac{K/(K-1)}{K} = \frac{K}{K-1},$$

$$K = 1 + \frac{K}{K/(K-1)} \le z_{m+2} = 1 + \frac{z_1}{y_{m+1}} \le 1 + \frac{K/(K-1)}{K} = \frac{K}{K-1}.$$

By induction, we get

$$x_i, y_i, z_i \in \left[K, \frac{K}{K-1}\right], \ i = 1, 2, \dots$$

Theorem 3.5 Assume A = 1. Then, every positive solution of system (3.1) is periodic of period 2.

Proof. From system (3.1), we have

$$\begin{aligned} x_{n+1} &= 1 + \frac{x_{n-1}}{z_n}, \quad y_{n+1} = 1 + \frac{y_{n-1}}{z_n}, \quad z_{n+1} = 1 + \frac{z_{n-1}}{y_n} \\ x_{n+2} &= 1 + \frac{x_n}{z_{n+1}}, \quad y_{n+2} = 1 + \frac{y_n}{z_{n+1}}, \quad z_{n+1} = 1 + \frac{z_{n-1}}{y_n} \\ &= 1 + \frac{x_n}{1 + \frac{z_{n-1}}{y_n}} &= 1 + \frac{y_n}{1 + \frac{z_{n-1}}{y_n}} &= 1 + \frac{z_n}{1 + \frac{y_{n-1}}{z_n}} \\ &= 1 + \frac{x_n y_n}{y_n + z_{n-1}} &= 1 + \frac{y_n^2}{y_n + z_{n-1}} &= 1 + \frac{z_n^2}{z_n + y_{n-1}} \\ &\text{(from 3.2)} \underbrace{z_{n-1} + x_n}_{z_{n-1}} & (from 3.2) \underbrace{z_{n-1} + y_n}_{z_{n-1}} & (from 3.2) \underbrace{y_{n-1} + z_n}_{y_{n-1}} \\ &= x_n &= y_n &= z_n \end{aligned}$$

Theorem 3.6 Assume A > 1. Then, every positive solution of system (3.1) is bounded.

Proof. Let $\{(x_n, y_n, z_n)\}$ is a positive solution of system (3.1). Clearly,

$$x_n, y_n, z_n > A > 1, \text{ for } n \ge 1.$$
 (3.18)

From (3.18), we have

$$x_{n+1} = A + \frac{x_{n-1}}{z_n} \le A + \frac{x_{n-1}}{A}, \quad n \ge 1.$$
(3.19)

 Set

$$u_{n+1} = A + \frac{u_{n-1}}{A}, \ n \ge 1$$
(3.20)

such that

$$u_s = x_s, \ u_{s+1} = x_{s+1}, \ s \in \{-1, 0, 1, ...\}, \ n \ge s.$$
 (3.21)

Then, the solution u_n of the difference equation (3.20) is as follows:

$$u_n = c_1 \left(\frac{1}{\sqrt{A}}\right)^n + c_1 \left(-\frac{1}{\sqrt{A}}\right)^n + \frac{A^2}{A-1}.$$
(3.22)

Indeed, from (3.20), we get

$$u_{n+1} - \frac{1}{A}u_{n-1} = 0 \Rightarrow \lambda^2 - \frac{1}{A} = 0$$
$$\Rightarrow \lambda_{1,2} = \pm \frac{1}{\sqrt{A}}.$$

The homogen solution of difference equation (3.20) is given by

$$u_h = c_1 \left(\frac{1}{\sqrt{A}}\right)^n + c_2 \left(-\frac{1}{\sqrt{A}}\right)^n.$$

Also, from (3.20), the equilibrium solution of difference equation (3.20) is following

$$\overline{x} - \frac{1}{A}\overline{x} = A \Rightarrow \overline{x} = \frac{A^2}{A-1}$$

In addition, the relations (3.19) and (3.22) imply that

$$x_{n+1} - u_{n+1} \le \frac{x_{n-1} - u_{n-1}}{A}, \quad n > s.$$
(3.23)

Then, by using (3.21) and (3.23) and induction, we have

$$x_n \le u_n, \ n \ge s. \tag{3.24}$$

Therefore, from (3.18), (3.22), and (3.24), we obtain

$$A < x_n \le c_1 \left(\frac{1}{\sqrt{A}}\right)^n + c_2 \left(-\frac{1}{\sqrt{A}}\right)^n + \frac{A^2}{A-1},$$

where

$$c_{1} = \frac{1}{2} \left(x_{0} + \sqrt{A}x_{1} - \frac{A^{2}}{A-1} \left(1 + \sqrt{A} \right) \right),$$

$$c_{2} = \frac{1}{2} \left(x_{0} - \sqrt{A}x_{1} - \frac{A^{2}}{A-1} \left(1 - \sqrt{A} \right) \right).$$

Similarly, we can prove that

$$A < y_n \le c_3 \left(\frac{1}{\sqrt{A}}\right)^n + c_4 \left(-\frac{1}{\sqrt{A}}\right)^n + \frac{A^2}{A-1},$$

$$A < z_n \le c_5 \left(\frac{1}{\sqrt{A}}\right)^n + c_6 \left(-\frac{1}{\sqrt{A}}\right)^n + \frac{A^2}{A-1},$$

where

$$c_{3} = \frac{1}{2} \left(y_{0} + \sqrt{A}y_{1} - \frac{A^{2}}{A - 1} \left(1 + \sqrt{A} \right) \right),$$

$$c_{4} = \frac{1}{2} \left(y_{0} - \sqrt{A}y_{1} - \frac{A^{2}}{A - 1} \left(1 - \sqrt{A} \right) \right),$$

$$c_{5} = \frac{1}{2} \left(z_{0} + \sqrt{A}z_{1} - \frac{A^{2}}{A - 1} \left(1 + \sqrt{A} \right) \right),$$

$$c_{6} = \frac{1}{2} \left(z_{0} - \sqrt{A}z_{1} - \frac{A^{2}}{A - 1} \left(1 - \sqrt{A} \right) \right).$$

Theorem 3.7 Suppose that A > 1. Then, the positive equilibrium point of system (3.1) is globally asymptotically stable.

Proof. By the means of Theorem 3.6, we set

$$L_{1} = \lim_{n \to \infty} \sup x_{n}, \quad L_{2} = \lim_{n \to \infty} \sup y_{n}, \quad L_{3} = \lim_{n \to \infty} \sup z_{n},$$

$$m_{1} = \lim_{n \to \infty} \inf x_{n}, \quad m_{2} = \lim_{n \to \infty} \inf y_{n}, \quad m_{3} = \lim_{n \to \infty} \inf z_{n}.$$
(3.25)

Then, from (3.1) and (4.4) we have

$$L_{1} \leq A + \frac{L_{1}}{m_{3}}, \quad L_{2} \leq A + \frac{L_{2}}{m_{3}}, \quad L_{3} \leq A + \frac{L_{3}}{m_{2}},$$

$$m_{1} \geq A + \frac{m_{1}}{L_{3}}, \quad m_{2} \geq A + \frac{m_{2}}{L_{3}}, \quad m_{3} \geq A + \frac{m_{3}}{L_{2}}.$$
(3.26)

Relations (4.5) imply that

 $AL_2 + m_3 \le m_3 L_2 \le Am_3 + L_2, \ AL_3 + m_2 \le m_2 L_3 \le Am_2 + L_3,$

from which we have

$$(A-1)(L_2-m_3) \le 0, \quad (A-1)(L_3-m_2) \le 0.$$

Since A > 1, we get

$$L_2 \le m_3 \le L_3, \ L_3 \le m_2 \le L_2,$$

from this it is obvious that

$$L_2 = L_3 = m_2 = m_3.$$

Moreover, from (4.5) it follows that

$$L_1 m_3 \leq A m_3 + L_1, \ m_1 L_3 \leq A L_3 + m_1,$$

from which

$$L_1(m_3 - 1) \le Am_3, \ AL_3 \le m_1(L_3 - 1).$$

Using (4.6), we have

$$L_1(L_3-1) \le m_1(L_3-1),$$

then

$$L_1 \leq m_1.$$

Since x_n is bounded, it implys that

$$L_1 = m_1.$$

Hence, every positive solution $\{(x_n, y_n, z_n)\}$ of system (3.1) tends to the positive equilibrium system (3.1). So, this completes the proof.

55

(3.27)



CHAPTER 4

GLOBAL ANALYSIS OF A SYSTEM OF HIGHER ORDER NONLINEAR DIFFERENCE EQUATIONS

We state that the results of this chapter are cited from [66] which has been published by us.

In this chapter, in the light of work in [65], we investigate the global asymptotic stability, boundedness character and oscillatory of positive solutions of the system of difference equations

$$x_{n+1} = A + \frac{x_{n-m}}{z_n}, \quad y_{n+1} = A + \frac{y_{n-m}}{z_n}, \quad z_{n+1} = A + \frac{z_{n-m}}{y_n}, \quad n = 0, 1, \dots,$$
(4.1)

where A and the initial values x_{-i} , y_{-i} , z_{-i} , for i = 0, 1, ..., m, are positive real numbers and m is positive integer.

4.1 MAIN RESULTS

In this section, we prove our main results. We deal with the following cases of 0 < A < 1, A = 1, and A > 1.

Theorem 4.1 If $(\overline{x}, \overline{y}, \overline{z})$ is a positive equilibrium point of system (4.1), then

$$(\overline{x}, \overline{y}, \overline{z}) = \begin{cases} (A+1, A+1, A+1), & \text{if } A \neq 1, \\ \left(\mu, \mu, \frac{\mu}{\mu-1}\right), \ \mu \in (1, \infty) & \text{if } A = 1. \end{cases}$$

Proof. It is easily seen from the definition of equilibrium point that the equilibrium points of system (4.1) are the nonnegative solution of the equations

$$\overline{x} = A + \frac{\overline{x}}{\overline{z}}, \, \overline{y} = A + \frac{\overline{y}}{\overline{z}}, \, \overline{z} = A + \frac{\overline{z}}{\overline{y}}.$$

From this, we get

$$\overline{xz} = A\overline{z} + \overline{x}, \ \overline{yz} = A\overline{z} + \overline{y}, \ \overline{zy} = A\overline{y} + \overline{z}$$

$$\Rightarrow \overline{xz} - \overline{x} = \overline{yz} - \overline{y}, \ A\overline{z} + \overline{y} = A\overline{y} + \overline{z}$$

$$\Rightarrow \overline{x}(\overline{z} - 1) = \overline{y}(\overline{z} - 1), \ \overline{z}(A - 1) = \overline{y}(A - 1).$$

From which it follows that if $A \neq 1$,

$$\overline{x} = \overline{y} = \overline{z} = A + 1 \Rightarrow (\overline{x}, \overline{y}, \overline{z}) = (A + 1, A + 1, A + 1).$$

Also, we have

$$\begin{aligned} \overline{\overline{z}z} &-\overline{\overline{x}} &= A, \ \overline{\overline{y}z} - \overline{\overline{y}} = A, \ \overline{\overline{z}y} - \overline{\overline{z}} = A \\ \Rightarrow & \frac{\overline{xz} - \overline{x}}{\overline{z}} = \frac{\overline{yz} - \overline{y}}{\overline{z}}, \ \overline{\overline{yz}} - \overline{y} = \frac{\overline{zy} - \overline{z}}{\overline{y}} \\ \Rightarrow & \overline{xz} - \overline{x} = \overline{yz} - \overline{y}, \ \overline{y}^2 \overline{z} - \overline{y}^2 = \overline{z}^2 \overline{y} - \overline{z}^2 \\ \Rightarrow & \overline{x} (\overline{z} - 1) = \overline{y} (\overline{z} - 1), \ \overline{yz} (\overline{y} - \overline{z}) = (\overline{y} - \overline{z}) (\overline{y} + \overline{z}). \end{aligned}$$

From which it follows that if A = 1,

$$\overline{x} = \overline{y} \text{ and } \overline{yz} = \overline{y} + \overline{z} \Rightarrow (\overline{x}, \overline{y}, \overline{z}) = \left(\mu, \mu, \frac{\mu}{\mu - 1}\right), \mu \in (1, \infty).$$

In that case, we have a continuous of positive equilibriums which lie on the hyperboloid $\overline{yz} = \overline{y} + \overline{z}$.

Theorem 4.2 Assume that 0 < A < 1. Let $\{(x_n, y_n, z_n)\}$ be an arbitrary positive solution of the system (4.1). Then, the following statements are true.

(i) If m is odd and $0 < x_{2k-1} < 1$, $0 < y_{2k-1} < 1$, $0 < z_{2k-1} < 1$, $x_{2k} > \frac{1}{1-A}$, $y_{2k} > \frac{1}{1-A}$, $z_{2k} > \frac{1}{1-A}$, $y_{2k} > \frac{1}{1-A}$, $z_{2k} > \frac{1}{1-A}$ for $k = \frac{1-m}{2}, \frac{3-m}{2}, ..., 0$, then

 $\lim_{n \to \infty} x_{2n} = \infty, \quad \lim_{n \to \infty} y_{2n} = \infty, \quad \lim_{n \to \infty} z_{2n} = \infty,$ $\lim_{n \to \infty} x_{2n+1} = A, \quad \lim_{n \to \infty} y_{2n+1} = A, \quad \lim_{n \to \infty} z_{2n+1} = A.$

(ii) If m is odd and $0 < x_{2k} < 1$, $0 < y_{2k} < 1$, $0 < z_{2k} < 1$, $x_{2k-1} > \frac{1}{1-A}$, $y_{2k-1} > \frac{1}{1-A}$, $z_{2k-1} > \frac{1}{1-A}$, $y_{2k-1} > \frac{1}{1-A}$, $z_{2k-1} > \frac{1}{1-A}$ for $k = \frac{1-m}{2}, \frac{3-m}{2}, \dots, 0$, then

$$\lim_{n \to \infty} x_{2n} = A, \quad \lim_{n \to \infty} y_{2n} = A, \quad \lim_{n \to \infty} z_{2n} = A,$$
$$\lim_{n \to \infty} x_{2n+1} = \infty, \quad \lim_{n \to \infty} y_{2n+1} = \infty, \quad \lim_{n \to \infty} z_{2n+1} = \infty.$$

(iii) If m is even, we can not get some useful results.

Proof.

(i) Clearly, we get

$$\begin{array}{rcl} 0 &< & x_1 = A + \frac{x_{-m}}{z_0} < A + \frac{1}{z_0} < A + (1 - A) = 1, \\ 0 &< & y_1 = A + \frac{y_{-m}}{z_0} < A + \frac{1}{z_0} < A + (1 - A) = 1, \\ 0 &< & z_1 = A + \frac{z_{-m}}{y_0} < A + \frac{1}{y_0} < A + (1 - A) = 1, \\ x_2 &= & A + \frac{x_{1-m}}{z_1} > x_{1-m} > \frac{1}{1 - A}, \\ y_2 &= & A + \frac{y_{1-m}}{z_1} > y_{1-m} > \frac{1}{1 - A}, \\ z_2 &= & A + \frac{z_{1-m}}{y_1} > z_{1-m} > \frac{1}{1 - A}. \end{array}$$

By induction for n = 1, 2, ..., we obtain

$$0 < x_{2n-1} < 1, \ 0 < y_{2n-1} < 1, \ 0 < z_{2n-1} < 1,$$

$$x_{2n} > \frac{1}{1-A}, \ y_{2n} > \frac{1}{1-A}, \ z_{2n} > \frac{1}{1-A}.$$
(4.2)

Thus, for $n \ge (m+2)/2$,

$$\begin{aligned} x_{2n} &= A + \frac{x_{2n-(m+1)}}{z_{2n-1}} > A + x_{2n-(m+1)} = 2A + \frac{x_{2n-(2m+2)}}{z_{2n-(m+2)}} \\ &> 2A + x_{2n-(2m+2)}, \\ y_{2n} &= A + \frac{y_{2n-(m+1)}}{z_{2n-1}} > A + y_{2n-(m+1)} = 2A + \frac{y_{2n-(2m+2)}}{z_{2n-(m+2)}} \\ &> 2A + y_{2n-(2m+2)}, \\ z_{2n} &= A + \frac{z_{2n-(m+1)}}{y_{2n-1}} > A + z_{2n-(m+1)} = 2A + \frac{z_{2n-(2m+2)}}{y_{2n-(m+2)}} \\ &> 2A + z_{2n-(2m+2)}, \end{aligned}$$

from which we get

$$\lim_{n \to \infty} x_{2n} = \infty, \quad \lim_{n \to \infty} y_{2n} = \infty, \quad \lim_{n \to \infty} z_{2n} = \infty.$$

Noting that (4.2) and taking limits on the both sides of three equations

$$x_{2n+1} = A + \frac{x_{2n-m}}{z_{2n}}, \ y_{2n+1} = A + \frac{y_{2n-m}}{z_{2n}}, \ z_{2n+1} = A + \frac{z_{2n-m}}{y_{2n}},$$

we have

$$\lim_{n \to \infty} x_{2n+1} = A, \quad \lim_{n \to \infty} y_{2n+1} = A, \quad \lim_{n \to \infty} z_{2n+1} = A.$$

(ii) Obviously, we have

$$\begin{aligned} x_1 &= A + \frac{x_{-m}}{z_0} > x_{-m} > \frac{1}{1 - A}, \\ y_1 &= A + \frac{y_{-m}}{z_0} > y_{-m} > \frac{1}{1 - A}, \\ z_1 &= A + \frac{z_{-m}}{y_0} > z_{-m} > \frac{1}{1 - A}, \\ 0 &< x_2 = A + \frac{x_{1-m}}{z_1} < A + \frac{1}{z_1} < A + (1 - A) = 1, \\ 0 &< y_2 = A + \frac{y_{1-m}}{z_1} < A + \frac{1}{z_1} < A + (1 - A) = 1, \\ 0 &< z_2 = A + \frac{z_{1-m}}{y_1} < A + \frac{1}{y_1} < A + (1 - A) = 1. \end{aligned}$$

By induction for n = 1, 2, ..., we obtain

$$\begin{aligned} x_{2n-1} > \frac{1}{1-A}, y_{2n-1} > \frac{1}{1-A}, z_{2n-1} > \frac{1}{1-A}, \\ 0 < x_{2n} < 1, \ 0 < y_{2n} < 1, \ 0 < z_{2n} < 1. \end{aligned} \tag{4.3}$$

So, for $n \ge (m+2)/2$,

$$\begin{aligned} x_{2n+1} &= A + \frac{x_{2n-m}}{z_{2n}} > A + x_{2n-m} = 2A + \frac{x_{(2n-2m)-1}}{z_{2n-(m+1)}} \\ &> 2A + x_{(2n-2m)-1}, \\ y_{2n+1} &= A + \frac{y_{2n-m}}{z_{2n}} > A + y_{2n-m} = 2A + \frac{y_{(2n-2m)-1}}{z_{2n-(m+1)}} \\ &> 2A + y_{(2n-2m)-1}, \\ z_{2n+1} &= A + \frac{z_{2n-m}}{y_{2n}} > A + z_{2n-m} = 2A + \frac{z_{(2n-2m)-1}}{y_{2n-(m+1)}} \\ &> 2A + z_{(2n-2m)-1}, \end{aligned}$$

from which we get

$$\lim_{n \to \infty} x_{2n+1} = \infty, \quad \lim_{n \to \infty} y_{2n+1} = \infty, \quad \lim_{n \to \infty} z_{2n+1} = \infty.$$

Noting that (4.3) and taking limits on the both sides of three equations

$$x_{2n} = A + \frac{x_{2n-(m+1)}}{z_{2n-1}}, \ y_{2n+1} = A + \frac{y_{2n-(m+1)}}{z_{2n-1}}, \ z_{2n+1} = A + \frac{z_{2n-(m+1)}}{y_{2n-1}},$$

we have

$$\lim_{n \to \infty} x_{2n} = A, \quad \lim_{n \to \infty} y_{2n} = A, \quad \lim_{n \to \infty} z_{2n+} = A.$$

Theorem 4.3 Suppose that A = 1. Then every positive solution of system (4.1) is bounded and persists.

Proof. Let $\{(x_n, y_n, z_n)\}$ be a positive solution of the system (4.1).

Obviously, $x_n > 1$, $y_n > 1$, $z_n > 1$, for $n \ge 1$. So, we have

$$x_i, y_i, z_i \in \left[M, \frac{M}{M-1}\right], \ i = 1, 2, ..., m+1,$$

where

$$M = \min\left\{\alpha, \frac{\beta}{\beta - 1}\right\} > 1, \ \alpha = \min_{1 \le i \le m + 1} \left\{x_i, y_i, z_i\right\}, \ \beta = \max_{1 \le i \le m + 1} \left\{x_i, y_i, z_i\right\}.$$

Then, we obtain

$$M = 1 + \frac{M}{M/(M-1)} \le x_{m+2} = 1 + \frac{x_1}{z_{m+1}} \le 1 + \frac{M/(M-1)}{M} = \frac{M}{M-1},$$

$$M = 1 + \frac{M}{M/(M-1)} \le y_{m+2} = 1 + \frac{y_1}{z_{m+1}} \le 1 + \frac{M/(M-1)}{M} = \frac{M}{M-1},$$

$$M = 1 + \frac{M}{M/(M-1)} \le z_{m+2} = 1 + \frac{z_1}{y_{m+1}} \le 1 + \frac{M/(M-1)}{M} = \frac{M}{M-1}.$$

By induction, we get

$$x_i, y_i, z_i \in \left[M, \frac{M}{M-1}\right], \ i = 1, 2, \dots$$

Theorem 4.4 Assume that A = 1. Let $\{(x_n, y_n, z_n)\}$ be a positive solution of the system (4.1). Then, either $\{(x_n, y_n, z_n)\}$ consists of a single semicycle or $\{(x_n, y_n, z_n)\}$ oscillates about the equilibrium point $(\overline{x}, \overline{y}, \overline{z}) = (\mu, \mu, \frac{\mu}{\mu-1})$ with semicycles having at most m terms.

Proof. Suppose that $\{(x_n, y_n, z_n)\}$ has at least two semicycles Then, there exists $N \ge -m$ such that either $x_N < \overline{x} \le x_{N+1}$ or $x_{N+1} < \overline{x} \le x_N$ ($y_N < \overline{y} \le y_{N+1}$ or $y_{N+1} < \overline{y} \le y_N$ and $z_N < \overline{z} \le z_{N+1}$ or $z_{N+1} < \overline{z} \le z_N$). Firstly, we assume that the case $x_N < \overline{x} \le x_{N+1}$, $y_N < \overline{y} \le y_{N+1}$ and $z_N < \overline{z} \le z_{N+1}$. Since the other case is similar, it will be omitted. Suppose that the positive semicycle beginning with the term ($x_{N+1}, y_{N+1}, z_{N+1}$) have m terms. Then we have

$$x_{N+1} < \overline{x} = \mu \le x_{N+m},$$

$$y_{N+1} < \overline{y} = \mu \le y_{N+m},$$

$$z_{N+1} < \overline{z} = \frac{\mu}{\mu - 1} \le z_{N+m}$$

Therefore, we get

$$\begin{aligned} x_{N+m+1} &= 1 + \frac{x_N}{z_{N+m}} < 1 + \frac{\overline{x}}{\overline{z}} = \mu, \\ y_{N+m+1} &= 1 + \frac{y_N}{z_{N+m}} < 1 + \frac{\overline{y}}{\overline{z}} = \mu, \\ z_{N+m+1} &= 1 + \frac{z_N}{y_{N+m}} < 1 + \frac{\overline{z}}{\overline{y}} = \frac{\mu}{\mu - 1} \end{aligned}$$

This completes the proof.

Theorem 4.5 Suppose that A > 1. Then every positive solution of system (4.1) is bounded and persists.

Proof. Let $\{(x_n, y_n, z_n)\}$ be a positive solution of the system (4.1). Obviously, $x_n > A > 1$, $y_n > A > 1$, $z_n > A > 1$, for $n \ge 1$. So, we have

$$x_i, y_i, z_i \in \left[M, \frac{M}{M-A}\right], \ i = 1, 2, ..., m+1$$

where

$$M = \min\left\{\alpha, \frac{\beta}{\beta - 1}\right\} > 1, \ \alpha = \min_{1 \le i \le m + 1} \left\{x_i, y_i, z_i\right\}, \ \beta = \max_{1 \le i \le m + 1} \left\{x_i, y_i, z_i\right\}.$$

Then, we obtain

$$M = A + \frac{M}{M/(M-A)} \le x_{m+2} = 1 + \frac{x_1}{z_{m+1}} \le 1 + \frac{M/(M-A)}{M} = \frac{M}{M-A}$$

$$M = A + \frac{M}{M/(M-A)} \le y_{m+2} = 1 + \frac{y_1}{z_{m+1}} \le 1 + \frac{M/(M-A)}{M} = \frac{M}{M-A},$$

$$M = A + \frac{M}{M/(M-A)} \le z_{m+2} = 1 + \frac{z_1}{y_{m+1}} \le 1 + \frac{M/(M-A)}{M} = \frac{M}{M-A}.$$

By induction, we get

$$x_i, y_i, z_i \in \left[M, \frac{M}{M-A}\right], \ i = 1, 2, \dots$$

The proof is completed.

Before we give the following theorems about the stability of the equilibrium points, we consider the following transformation to build the corresponding linearized form of system (4.1) :

$$(x_n, x_{n-1}, ..., x_{n-m}, y_n, y_{n-1}, ..., y_{n-m}, z_n, z_{n-1}, ..., z_{n-m})$$

$$\rightarrow (f, f_1, ..., f_m, g, g_1, ..., g_m, h, h_1, ..., h_m)$$

where

$$f = A + \frac{x_{n-m}}{z_n}$$

$$f_1 = x_n$$

$$\vdots$$

$$f_m = x_{n-m}$$

$$g = A + \frac{y_{n-m}}{z_n}$$

$$g_1 = y_n$$

$$\vdots$$

$$g_m = y_{n-m}$$

$$h = A + \frac{z_{n-m}}{y_n}$$

$$h_1 = z_n$$

$$\vdots$$

$$h_m = z_{n-m}.$$

The Jacobian matrix about the equilibrium point $(\overline{x}, \overline{y}, \overline{z})$ under the above transformation is given by

where $B = (b_{ij}), 1 \le i, j \le 3m + 3$ is an $(3m + 3) \times (3m + 3)$ matrix.

Theorem 4.6 If A = 1, then the equilibrium point of system (4.1) is locally asymptotically stable.

Proof. The linearized system of system (4.1) about the equilibrium point $\left(\mu, \mu, \frac{\mu}{\mu-1}\right)$ is

 $X_{n+1} = BX_n,$

where $X_n = (x_n, x_{n-1}, ..., x_{n-m}, y_n, y_{n-1}, ..., y_{n-m}, z_n, z_{n-1}, ..., z_{n-m})^T$ and

.

Let $\lambda_1, \lambda_2, ..., \lambda_{3m+3}$ denote the 3m + 3 eigenvalues of the matrix B and

 $D = diag(d_1, d_2, ..., d_{3m+3})$

be a diagonal matrix, where

$$d_1 = d_{m+2} = d_{2m+3} = 1, \ d_{1+k} = d_{m+2+k} = d_{2m+3+k} = 1 - k\varepsilon, \ 1 \le k \le m$$

and

$$0 < \varepsilon < \left\{ \frac{\mu^2 - 2\mu + 2}{m\mu}, \frac{\mu^2 - 2\mu + 2}{m\mu(\mu - 1)} \right\}.$$

Obviously, D is invertible. Computing matrix DBD^{-1} , we have that

where

$$p = \frac{\mu - 1}{\mu} \frac{d_1}{d_{m+1}},$$

$$q = -\frac{(\mu - 1)^2}{\mu} \frac{d_1}{d_{2m+3}},$$

$$r = \frac{\mu - 1}{\mu} \frac{d_{m+2}}{d_{2m+2}},$$

$$s = -\frac{(\mu - 1)^2}{\mu} \frac{d_{m+2}}{d_{2m+3}},$$

$$t = -\frac{1}{\mu} \frac{d_{2m+3}}{d_{2m+3}},$$

$$w = \frac{1}{\mu} \frac{d_{2m+3}}{d_{3m+3}}.$$

The three chains of inequalities

$$\begin{array}{rcl} 1 & = & d_1 > d_2 > \ldots > d_m > d_{m+1} > 0, \\ \\ 1 & = & d_{m+2} > d_{m+3} > \ldots > d_{2m+1} > d_{2m+2} > 0, \\ \\ 1 & = & d_{2m+3} > d_{2m+4} > \ldots > d_{3m+2} > d_{3m+3} > 0, \end{array}$$

imply that

$$d_2 d_1^{-1} < 1, d_3 d_2^{-1} < 1, \dots, d_{m+1} d_m^{-1} < 1,$$

$$d_{m+3} d_{m+2}^{-1} < 1, d_{m+4} d_{m+3}^{-1} < 1, \dots, d_{2m+2} d_{2m+1}^{-1} < 1,$$

$$d_{2m+4} d_{2m+3}^{-1} < 1, d_{2m+5} d_{2m+4}^{-1} < 1, \dots, d_{3m+3} d_{3m+2}^{-1} < 1.$$

Also,

$$\begin{pmatrix} \frac{\mu-1}{\mu} \end{pmatrix} d_1 d_{m+1}^{-1} + \left(-\frac{(\mu-1)^2}{\mu} \right) d_1 d_{2m+3}^{-1}$$

$$= \left(\frac{\mu-1}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right) - \frac{(\mu-1)^2}{\mu}$$

$$< \left(\frac{\mu-1}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right) - \left(\frac{(\mu-1)^2}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right)$$

$$= \left(\frac{-\mu^2 + 3\mu - 2}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right) < 1,$$

$$\left(\frac{\mu-1}{\mu} \right) d_{m+2} d_{2m+2}^{-1} + \left(-\frac{(\mu-1)^2}{\mu} \right) d_{m+2} d_{2m+3}^{-1}$$

$$= \left(\frac{\mu-1}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right) - \frac{(\mu-1)^2}{\mu}$$

$$= \left(\frac{-\mu^2 + 3\mu - 2}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right) < 1,$$

$$\left(-\frac{1}{\mu(\mu-1)^2} \right) d_{2m+3} d_{m+2}^{-1} + \left(\frac{1}{\mu} \right) d_{2m+3} d_{3m+3}^{-1}$$

$$= \left(-\frac{1}{\mu(\mu-1)^2} \right) + \left(\frac{1}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right)$$

$$< \left(-\frac{1}{\mu(\mu-1)^2} \right) \left(\frac{1}{1-m\varepsilon} \right) + \left(\frac{1}{\mu} \right) \left(\frac{1}{1-m\varepsilon} \right)$$

$$= \left(\frac{\mu-2}{\mu(\mu-1)} \right) \left(\frac{1}{1-m\varepsilon} \right) < 1.$$

Since B has the same eigenvalues as $DBD^{-1} = E = (e_{ij})$, we obtain that

$$\max_{1 \le i \le 3m+3} |\lambda_i| \le \|DBD^{-1}\|_{\infty} \\
= \max_{1 \le i \le 3m+3} \left\{ \sum_{j=1}^{3m+3} |e_{ij}| \right\} \\
= \max \left\{ \begin{array}{l} d_2 d_1^{-1}, d_3 d_2^{-1}, \dots, d_{m+1} d_m^{-1}, \\ d_{m+3} d_{m+2}^{-1}, d_{m+4} d_{m+3}^{-1}, \dots, d_{2m+2} d_{2m+1}^{-1}, \\ d_{2m+4} d_{2m+3}^{-1}, d_{2m+5} d_{2m+4}^{-1}, \dots, d_{3m+3} d_{3m+2}^{-1}, \\ \left(\frac{\mu-1}{\mu}\right) d_1 d_{m+1}^{-1} - \left(\frac{(\mu-1)^2}{\mu}\right) d_1 d_{2m+3}^{-1}, \\ \left(\frac{\mu-1}{\mu}\right) d_{m+2} d_{2m+2}^{-1} - \left(\frac{(\mu-1)^2}{\mu}\right) d_{m+2} d_{2m+3}^{-1}, \\ \left(-\frac{1}{\mu(\mu-1)^2}\right) d_{2m+3} d_{m+2}^{-1} + \left(\frac{1}{\mu}\right) d_{2m+3} d_{3m+3}^{-1} \\ < 1. \end{array} \right\}$$

This implies that the equilibrium point of system (4.1) is locally asymptotically stable.

Theorem 4.7 If A > 1, then the equilibrium point of system (4.1) is locally asymptotically stable.

Proof. The linearized system of system (4.1) about the equilibrium point $(\overline{x}, \overline{y}, \overline{z}) = (A+1, A+1, A+1)$ is

$$X_{n+1} = BX_n,$$

where $X_n = (x_n, x_{n-1}, ..., x_{n-m}, y_n, y_{n-1}, ..., y_{n-m}, z_n, z_{n-1}, ..., z_{n-m})^T$ and

where c = A + 1.

Let $\lambda_1, \lambda_2, ..., \lambda_{3m+3}$ denote the 3m+3 eigenvalues of the matrix B and

$$D = diag(d_1, d_2, ..., d_{3m+3})$$

be a diagonal matrix, where

$$d_1 = d_{m+2} = d_{2m+3} = 1, \ d_{1+k} = d_{m+2+k} = d_{2m+3+k} = 1 - k\varepsilon, \ 1 \le k \le m$$

and

$$0 < \varepsilon < \left\{\frac{1}{m}, \frac{c-2}{cm}\right\}.$$

Obviously, D is invertible. Computing matrix DBD^{-1} , we have that

 $DBD^{-1} =$

(0		0	$\frac{c^{-1}d_1}{d_{m+1}}$	0		0	0	$\frac{-c^{-1}d_1}{d_{2m+3}}$		0	0
	$\frac{d_2}{d_1}$		0	0	0		0	0	0		0	0
	÷	·	:	÷	÷	·	÷	:	:	·	÷	÷
	0		$\frac{d_{m+1}}{d_m}$	0	0		0	0	0		0	0
	0		0	0	0		0	$\frac{c^{-1}d_{m+2}}{d_{2m+2}}$	$\frac{-c^{-1}d_{m+2}}{d_{2m+3}}$		0	0
	0		0	0	$\frac{d_{m+3}}{d_{m+2}}$		0	0	0		0	0
	÷	·	÷	÷	÷	·	÷	÷	:	·	÷	÷
	0		0	0	0		$\frac{d_{2m+2}}{d_{2m+1}}$	0	0		0	0
	0		0	0	$\frac{-c^{-1}d_{2m+3}}{d_{m+2}}$		0	0	0		0	$\frac{c^{-1}d_{2m+3}}{d_{3m+3}}$
	0		0	0	0		0	0	$\frac{d_{2m+4}}{d_{2m+3}}$		0	0
	÷		1	÷	÷	•••	÷	÷	÷		÷	÷
	0		0	0	0		0	0	0		$\frac{d_{3m+3}}{d_{3m+2}}$	0)
ΤÌ	ne tl	hree o	chains	of ineq	ualities						0.0012	,

 $\begin{array}{rcl} 1 & = & d_1 > d_2 > \ldots > d_m > d_{m+1} > 0, \\ \\ 1 & = & d_{m+2} > d_{m+3} > \ldots > d_{2m+1} > d_{2m+2} > 0, \\ \\ 1 & = & d_{2m+3} > d_{2m+4} > \ldots > d_{3m+2} > d_{3m+3} > 0, \end{array}$

imply that

$$\begin{aligned} d_2 d_1^{-1} &< 1, \, d_3 d_2^{-1} < 1, \dots, \, d_{m+1} d_m^{-1} < 1, \\ d_{m+3} d_{m+2}^{-1} &< 1, \, d_{m+4} d_{m+3}^{-1} < 1, \dots, \, d_{2m+2} d_{2m+1}^{-1} < 1, \\ d_{2m+4} d_{2m+3}^{-1} &< 1, \, d_{2m+5} d_{2m+4}^{-1} < 1, \dots, \, d_{3m+3} d_{3m+2}^{-1} < 1. \end{aligned}$$

Also,

$$c^{-1}d_{1}d_{m+1}^{-1} + c^{-1}d_{1}d_{2m+3}^{-1} = c^{-1}\left(\frac{1}{1-m\varepsilon} + 1\right)$$

$$< c^{-1}\frac{2}{1-m\varepsilon} < 1,$$

$$c^{-1}d_{m+2}d_{2m+2}^{-1} + c^{-1}d_{m+2}d_{2m+3}^{-1} = c^{-1}\left(\frac{1}{1-m\varepsilon} + 1\right)$$

$$< c^{-1}\frac{2}{1-m\varepsilon} < 1,$$

$$c^{-1}d_{2m+3}d_{m+2}^{-1} + c^{-1}d_{2m+3}d_{3m+3}^{-1} = c^{-1}\left(1 + \frac{1}{1 - m\varepsilon}\right)$$
$$< c^{-1}\frac{2}{1 - m\varepsilon} < 1.$$

Since B has the same eigenvalues as $DBD^{-1} = E = (e_{ij})$, we obtain that

$$\max_{1 \le i \le 3m+3} |\lambda_i| \le \|DBD^{-1}\|_{\infty} \\
= \max_{1 \le i \le 3m+3} \left\{ \sum_{j=1}^{3m+3} |e_{ij}| \right\} \\
= \max \left\{ \begin{array}{c} d_2 d_1^{-1}, d_3 d_2^{-1}, \dots, d_{m+1} d_m^{-1}, \\ d_{m+3} d_{m+2}^{-1}, d_{m+4} d_{m+3}^{-1}, \dots, d_{2m+2} d_{2m+1}^{-1}, \\ d_{2m+4} d_{2m+3}^{-1}, d_{2m+5} d_{2m+4}^{-1}, \dots, d_{3m+3} d_{3m+2}^{-1}, \\ c^{-1} d_1 d_{m+1}^{-1} + c^{-1} d_1 d_{2m+3}^{-1}, \\ c^{-1} d_{m+2} d_{2m+2}^{-1} + c^{-1} d_{m+2} d_{2m+3}^{-1}, \\ c^{-1} d_{2m+3} d_{m+2}^{-1} + c^{-1} d_{2m+3} d_{3m+3}^{-1} \\ \end{array} \right\} \\ < 1.$$

This implies that the equilibrium point of system (4.1) is locally asymptotically stable.

Theorem 4.8 Assume that A > 1. Then, the positive equilibrium point of system (4.1) is globally asymptotically stable.

Proof. Using Theorem 4.5, we have

$$L_{1} = \lim_{n \to \infty} \sup x_{n}, \quad L_{2} = \lim_{n \to \infty} \sup y_{n}, \quad L_{3} = \lim_{n \to \infty} \sup z_{n},$$

$$m_{1} = \lim_{n \to \infty} \inf x_{n}, \quad m_{2} = \lim_{n \to \infty} \inf y_{n}, \quad m_{3} = \lim_{n \to \infty} \inf z_{n}.$$
(4.4)

Then, from (4.1) and (4.4) we have

$$L_{1} \leq A + \frac{L_{1}}{m_{3}}, \quad L_{2} \leq A + \frac{L_{2}}{m_{3}}, \quad L_{3} \leq A + \frac{L_{3}}{m_{2}},$$

$$m_{1} \geq A + \frac{m_{1}}{L_{3}}, \quad m_{2} \geq A + \frac{m_{2}}{L_{3}}, \quad m_{3} \geq A + \frac{m_{3}}{L_{2}}.$$

$$(4.5)$$

Relations (4.5) imply that

$$AL_2 + m_3 \le m_3 L_2 \le Am_3 + L_2, \ AL_3 + m_2 \le m_2 L_3 \le Am_2 + L_3,$$

from which we have

$$(A-1)(L_2-m_3) \le 0, (A-1)(L_3-m_2) \le 0.$$

Since A > 1, we get

$$L_2 \le m_3 \le L_3, \ L_3 \le m_2 \le L_2,$$

from which

$$L_2 = L_3 = m_2 = m_3. (4.6)$$

Moreover, from (4.5) it follows that $L_1m_3 \leq Am_3 + L_1$, $m_1L_3 \leq AL_3 + m_1$, from which

$$L_1(m_3-1) \le Am_3, \ AL_3 \le m_1(L_3-1).$$

Using (4.6), we have

$$L_1(L_3-1) \le m_1(L_3-1),$$

from which $L_1 \leq m_1$.

Since x_n is bounded, it implys that $L_1 = m_1$.

Hence, every positive solution $\{(x_n, y_n, z_n)\}$ of system (4.1) tends to the positive equilibrium system (4.1). So, the proof is completed.

CHAPTER 5

THE NATURE OF SOLUTIONS OF A SYSTEM OF SECOND ORDER RATIONAL DIFFERENCE EQUATIONS

We state that the results of this chapter are cited from [68] which was published by us. In the light of the works in [65] and [66], the aim of this chapter is to study local stability of the equilibrium points and oscillation behaviour of positive solutions of the following system of rational difference equations

$$x_{n+1} = A + \frac{x_{n-1}^p}{z_n^p}, \ y_{n+1} = A + \frac{y_{n-1}^p}{z_n^p}, \ z_{n+1} = A + \frac{z_{n-1}^p}{y_n^p}, \ n = 0, 1, ...,$$
(5.1)

where $A \in (0, \infty)$, $p \in [1, \infty)$ and the initial values $x_i, y_i, z_i \in (0, \infty)$, i = -1, 0.

5.1 MAIN RESULTS

In this section, we prove our main results.

Theorem 5.1 The following statements are true:

- (i) The system (5.1) has a positive equilibrium point $(\overline{x}, \overline{y}, \overline{z}) = (A+1, A+1, A+1)$.
- (ii) If A > 2p 1, then the equilibrium point of system (5.1) is locally asymptotically stable.
- (iii) If A < 2p 1, then the equilibrium point of system (5.1) is unstable.
- (iv) Also, when A = 2p 1 and p = 1, the results has been investigated in [65].

Proof.

(i) It is easily seen from the definition of equilibrium point.

(ii) We consider the following transformation to build the corresponding linearized form of system (5.1):

 $(x_n, x_{n-1}, y_n, y_{n-1}, z_n, z_{n-1}) \rightarrow (f, f_1, g, g_1, h, h_1)$

where

$$f = A + \frac{x_{n-1}^p}{z_n^p}$$

$$f_1 = x_n$$

$$g = A + \frac{y_{n-1}^p}{z_n^p}$$

$$g_1 = y_n$$

$$h = A + \frac{z_{n-1}^p}{y_n^p}$$

$$h_1 = z_n.$$

The Jacobian matrix about the equilibrium point $(\overline{x}, \overline{y}, \overline{z})$ under the above transformation is given by

$$B\left(\overline{x}, \overline{y}, \overline{z}\right) = \begin{pmatrix} 0 & \frac{p\overline{x}^{p-1}}{\overline{z}^{p}} & 0 & 0 & -\frac{p\overline{x}^{p}}{\overline{z}^{p+1}} & 0\\ 1 & 0 & 0 & 0 & 0 & 0\\ 0 & 0 & 0 & \frac{p\overline{y}^{p-1}}{\overline{z}^{p}} & -\frac{p\overline{y}^{p}}{\overline{z}^{p+1}} & 0\\ 0 & 0 & 1 & 0 & 0 & 0\\ 0 & 0 & -\frac{p\overline{z}^{p}}{\overline{y}^{p+1}} & 0 & 0 & \frac{p\overline{z}^{p-1}}{\overline{y}^{p}}\\ 0 & 0 & 0 & 0 & 1 & 0 \end{pmatrix}.$$
(5.2)

Hence, the linearized system of system (5.1) about the equilibrium point

$$(\overline{x},\overline{y},\overline{z}) = (A+1,A+1,A+1)$$

is

$$X_{n+1} = B\left(\overline{x}, \overline{y}, \overline{z}\right) X_n,$$

where

$$X_n = (x_n, x_{n-1}, y_n, y_{n-1}, z_n, z_{n-1})^T$$

and

$$B\left(\overline{x},\overline{y},\overline{z}\right) = \begin{pmatrix} 0 & \frac{p}{A+1} & 0 & 0 & -\frac{p}{A+1} & 0\\ 1 & 0 & 0 & 0 & 0 & 0\\ 0 & 0 & 0 & \frac{p}{A+1} & -\frac{p}{A+1} & 0\\ 0 & 0 & 1 & 0 & 0 & 0\\ 0 & 0 & -\frac{p}{A+1} & 0 & 0 & \frac{p}{A+1}\\ 0 & 0 & 0 & 0 & 1 & 0 \end{pmatrix}$$

Then, the characteristic equation of $B(\overline{x}, \overline{y}, \overline{z})$ about

$$(\overline{x}, \overline{y}, \overline{z}) = (A+1, A+1, A+1)$$

is

$$\lambda^{6} - \left(\frac{p^{2}}{\left(A+1\right)^{2}} + 3\frac{p}{A+1}\right)\lambda^{4} + \left(\frac{p^{3}}{\left(A+1\right)^{3}} + 3\frac{p^{2}}{\left(A+1\right)^{2}}\right)\lambda^{2} - \frac{p^{3}}{\left(A+1\right)^{3}} = 0.$$
(5.3)

From this, the roots of characteristic equation (5.3) are

$$\lambda_{1} = \sqrt{\frac{p}{A+1}},$$

$$\lambda_{2} = -\sqrt{\frac{p}{A+1}},$$

$$\lambda_{3} = -\frac{1}{2A+2} \left(p + \sqrt{p^{2} + 4Ap + 4p} \right),$$

$$\lambda_{4} = \frac{1}{2A+2} \left(-p + \sqrt{p^{2} + 4Ap + 4p} \right),$$

$$\lambda_{5} = \frac{1}{2A+2} \left(p + \sqrt{p^{2} + 4Ap + 4p} \right),$$

$$\lambda_{6} = \frac{1}{2A+2} \left(p - \sqrt{p^{2} + 4Ap + 4p} \right).$$

From the Linearized Stability Theorem, since A > 2p - 1, all roots of the characterictic equation lie inside the open unit disk $|\lambda| < 1$. Therefore, the positive equilibrium point of system (5.1) is locally asymptotically stable.

(iii) From the proof of (ii), it is true.

Theorem 5.2 Let 0 < A < 1 and $\{(x_n, y_n, z_n)\}$ be an arbitrary positive solution of system (5.1). Then, the following statements are true.

(i) If

$$0 < x_{-1} < 1, \ 0 < y_{-1} < 1, \ 0 < z_{-1} < 1,$$

$$x_{0} \geq \frac{1}{(1-A)^{1/p}}, \ y_{0} \geq \frac{1}{(1-A)^{1/p}}, \ z_{0} \geq \frac{1}{(1-A)^{1/p}},$$
(5.4)

then

$$\lim_{n \to \infty} x_{2n+1} = A, \quad \lim_{n \to \infty} y_{2n+1} = A, \quad \lim_{n \to \infty} z_{2n+1} = A,$$
$$\lim_{n \to \infty} x_{2n} = \infty, \quad \lim_{n \to \infty} y_{2n} = \infty, \quad \lim_{n \to \infty} z_{2n} = \infty.$$

(ii) If

$$0 < x_{0} < 1, 0 < y_{0} < 1, 0 < z_{0} < 1,$$

$$x_{-1} \geq \frac{1}{(1-A)^{1/p}}, y_{-1} \geq \frac{1}{(1-A)^{1/p}}, z_{-1} \geq \frac{1}{(1-A)^{1/p}},$$
(5.5)

then

$$\lim_{n \to \infty} x_{2n+1} = \infty, \quad \lim_{n \to \infty} y_{2n+1} = \infty, \quad \lim_{n \to \infty} z_{2n+1} = \infty,$$
$$\lim_{n \to \infty} x_{2n} = A, \quad \lim_{n \to \infty} y_{2n} = A, \quad \lim_{n \to \infty} z_{2n} = A.$$

Proof. (i) From system (5.1) and (5.4), we have

$$\begin{aligned} x_1 &= A + \frac{x_{-1}^p}{z_0^p} \le A + \frac{1}{z_0^p} \le A + (1 - A) = 1, \\ y_1 &= A + \frac{y_{-1}^p}{z_0^p} \le A + \frac{1}{z_0^p} \le A + (1 - A) = 1, \\ z_1 &= A + \frac{z_{-1}^p}{y_0^p} \le A + \frac{1}{y_0^p} \le A + (1 - A) = 1, \\ x_1 &= A + \frac{x_{-1}^p}{z_0^p} > A, \\ y_1 &= A + \frac{y_{-1}^p}{z_0^p} > A, \\ z_1 &= A + \frac{z_{-1}^p}{y_0^p} > A. \end{aligned}$$

Hence,

$$x_1 \in (A, 1], y_1 \in (A, 1], z_1 \in (A, 1].$$

Also,

$$x_{2} = A + \frac{x_{0}^{p}}{z_{1}^{p}} \ge A + x_{0}^{p},$$

$$y_{2} = A + \frac{y_{0}^{p}}{z_{1}^{p}} \ge A + y_{0}^{p},$$

$$z_{2} = A + \frac{z_{0}^{p}}{y_{1}^{p}} \ge A + z_{0}^{p}.$$

Similarly, we get

$$\begin{aligned} x_3 &= A + \frac{x_1^p}{z_2^p} \le A + \frac{1}{(A + z_0^p)^p} \le A + \frac{1}{A + z_0^p} \le A + \frac{1}{z_0^p} \le A + (1 - A) = 1, \\ y_3 &= A + \frac{y_1^p}{z_2^p} \le A + \frac{1}{(A + z_0^p)^p} \le A + \frac{1}{A + z_0^p} \le A + \frac{1}{z_0^p} \le A + (1 - A) = 1, \\ z_3 &= A + \frac{z_1^p}{y_2^p} \le A + \frac{1}{(A + y_0^p)^p} \le A + \frac{1}{A + y_0^p} \le A + \frac{1}{y_0^p} \le A + (1 - A) = 1. \end{aligned}$$

Thus,

 $x_3 \in (A, 1], y_3 \in (A, 1], z_3 \in (A, 1].$

Also,

$$\begin{aligned} x_4 &= A + \frac{x_2^p}{z_3^p} \ge A + x_2^p \ge A + (A + x_0^p)^p \ge A + (A + x_0^p) = 2A + x_0^p, \\ y_4 &= A + \frac{y_2^p}{z_3^p} \ge A + y_2^p \ge A + (A + y_0^p)^p \ge A + (A + y_0^p) = 2A + y_0^p, \\ z_4 &= A + \frac{z_2^p}{y_3^p} \ge A + z_2^p \ge A + (A + z_0^p)^p \ge A + (A + z_0^p) = 2A + z_0^p. \end{aligned}$$

By induction for n = 1, 2, ..., we obtain

$$A < x_{2n-1} < 1, \ A < y_{2n-1} < 1, \ A < z_{2n-1} < 1,$$

$$x_{2n} \geq nA + x_0^p, \ y_{2n} \geq nA + y_0^p, \ z_{2n} \geq nA + z_0^p.$$
(5.6)

From system (5.1) and (5.6), it follows that

$$\lim_{n \to \infty} x_{2n} = \infty, \quad \lim_{n \to \infty} y_{2n} = \infty, \quad \lim_{n \to \infty} z_{2n} = \infty,$$
$$\lim_{n \to \infty} x_{2n+1} = \lim_{n \to \infty} \left(A + \frac{x_{2n-1}^p}{z_{2n}^p} \right) = A,$$
$$\lim_{n \to \infty} y_{2n+1} = \lim_{n \to \infty} \left(A + \frac{y_{2n-1}^p}{z_{2n}^p} \right) = A,$$
$$\lim_{n \to \infty} z_{2n+1} = \lim_{n \to \infty} \left(A + \frac{z_{2n-1}^p}{y_{2n}^p} \right) = A.$$

(ii) The proof is similar to the proof of (i), so we omit it. The proof is completed.

Theorem 5.3 Let $\{(x_n, y_n, z_n)\}$ be a positive solution of system (5.1) which consists of at least two semicycles. Then $\{(x_n, y_n, z_n)\}_{n=-1}^{\infty}$ is oscillatory.

Proof. Since $\{(x_n, y_n, z_n)\}_{n=-1}^{\infty}$ has at least two semicycles, there exists $N \ge 0$ such that either

$$\begin{aligned}
 x_{N-1} &< A+1 \le x_N, \\
 y_{N-1} &< A+1 \le y_N, \\
 z_{N-1} &< A+1 \le z_N,
 \end{aligned}$$
(5.7)

or

$$x_N < A+1 \le x_{N-1},$$

 $y_N < A+1 \le y_{N-1},$
 $z_N < A+1 \le z_{N-1}.$

First, we suppose the case (5.7). Then

$$\begin{aligned} x_{N+1} &= A + \frac{x_{N-1}^p}{z_N^p} < A + 1, \\ y_{N+1} &= A + \frac{y_{N-1}^p}{z_N^p} < A + 1, \\ z_{N+1} &= A + \frac{z_{N-1}^p}{y_N^p} < A + 1, \\ x_{N+2} &= A + \frac{x_{N+1}^p}{z_{N+1}^p} > A + 1, \\ y_{N+2} &= A + \frac{y_N^p}{z_{N+1}^p} > A + 1, \\ z_{N+2} &= A + \frac{z_N^p}{y_{N+1}^p} > A + 1. \end{aligned}$$

So, we have

$$\begin{aligned} x_{N+1} &< A+1 < x_{N+2}, \\ y_{N+1} &< A+1 < y_{N+2}, \\ z_{N+1} &< A+1 < z_{N+2}. \end{aligned}$$

Last, we suppose the case (5.8). The case is similar to the first case, so we leave it to readers.

(5.8)

CHAPTER 6

THE EXACT SOLUTIONS OF FOUR RATIONAL DIFFERENCE EQUATIONS ASSOCIATED TO TRIBONACCI NUMBERS

Initially, we express that the results of this chapter are cited from [101] which has been published by us.

In Chapter 6 and Chapter 7, we discuss eight cases (eight distinct difference equations) of the following difference equation

$$x_{n+1} = \frac{\pm 1}{x_n \left(x_{n-1} \pm 1\right) \pm 1}.$$

We study the four cases of eight cases in this chapter and the remaining four cases in Chapter 7.

As far as we examine, there is no paper dealing with the following difference equations. Hence, in this chapter, we investigate the form of solutions, stability character and asymptotic behavior of the following four rational difference equations

$$x_{n+1} = \frac{1}{x_n \left(x_{n-1} - 1\right) - 1}, \quad n = 0, 1, ...,$$
(6.1)

$$x_{n+1} = \frac{1}{x_n \left(x_{n-1} + 1\right) + 1}, \quad n = 0, 1, ...,$$
(6.2)

$$x_{n+1} = \frac{-1}{x_n \left(x_{n-1} - 1\right) + 1}, \quad n = 0, 1, ...,$$
(6.3)

$$x_{n+1} = \frac{-1}{x_n \left(x_{n-1} + 1\right) - 1}, \quad n = 0, 1, ...,$$
(6.4)

such that their solutions are associated with Tribonacci numbers.

Our aim in this chapter is to investigate some relationships both between Tribonacci numbers and solutions of above mentioned difference equations and between the Tribonacci constant and the equilibrium points of these difference equations.

6.1 MAIN RESULTS

In this section, we present our main results considering above mentioned difference equations. Our aim is to investigate the general solutions in explicit form of difference equations and the asymptotic behavior of solutions of difference equations.

6.1.1 The Difference Equation (6.1)

In this subsection, we consider the Eq.(6.1), that is,

$$x_{n+1} = \frac{1}{x_n (x_{n-1} - 1) - 1}, \quad n = 0, 1, \dots$$

and investigate the dynamical behavior of solutions of Eq.(6.1).

Theorem 6.1 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (6.1). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n\}_{n=-1}^{\infty}$ is given by

$$x_{n} = \frac{T_{-n}x_{-1}x_{0} + (T_{-(n+1)} + T_{-(n+2)})x_{0} + T_{-(n+1)}}{T_{-(n+1)}x_{-1}x_{0} + (T_{-n} - T_{-(n+1)})x_{0} + T_{-(n+2)}},$$
(6.5)

where T_n is the nth Tribonacci number and the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_1$, with F_1 is the forbidden set of Eq.(6.1) given by

$$F_1 = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : T_{-(n+1)} x_{-1} x_0 + (T_{-n} - T_{-(n+1)}) x_0 + T_{-(n+2)} = 0 \right\}.$$

Proof. We will prove this theorem by induction on k. For k = 0, from Eq.(6.1),

$$x_{1} = \frac{1}{x_{0}(x_{-1}-1)-1} = \frac{1}{x_{-1}x_{0}-x_{0}-1} = \frac{T_{-1}x_{-1}x_{0} + (T_{-2}+T_{-3})x_{0} + T_{-2}}{T_{-2}x_{-1}x_{0} + (T_{-1}-T_{-2})x_{0} + T_{-3}}$$

Now, we assume that

$$x_{k} = \frac{T_{-k}x_{-1}x_{0} + (T_{-(k+1)} + T_{-(k+2)})x_{0} + T_{-(k+1)}}{T_{-(k+1)}x_{-1}x_{0} + (T_{-k} - T_{-(k+1)})x_{0} + T_{-(k+2)}}$$
(6.6)

is true for all $1 \le n \le k$. Hence, we have to prove that it is true for k + 1. Taking into account (2.12) and (6.6), we have

$$\begin{array}{ll} x_{k+1} \\ = & \frac{1}{x_k \left(x_{k-1} - 1\right) - 1} \\ = & \frac{1}{\left(\frac{T_{-k}x_{-1}x_0 + \left(T_{-(k+1)} + T_{-(k+2)}\right)x_0 + T_{-(k+1)}}{T_{-(k+1)}x_{-1}x_0 + \left(T_{-k} - T_{-(k+1)}\right)x_0 + T_{-(k+2)}}\right) \left(\frac{T_{-(k-1)}x_{-1}x_0 + \left(T_{-k} - T_{-(k+1)}\right)x_0 + T_{-k}}{T_{-k}x_{-1}x_0 + \left(T_{-k} - T_{-(k+1)}\right)x_0 + T_{-(k+2)}} - 1\right) - 1 \\ = & \frac{T_{-(k+1)}x_{-1}x_0 + \left(T_{-k} - T_{-(k+1)}\right)x_0 + T_{-(k+2)}}{\left(T_{-(k-1)} - T_{-k} - T_{-(k+1)}\right)x_{-1}x_0 + \left(T_{-(k+1)} - T_{-(k+2)}\right)x_0 + T_{-k} - T_{-(k+1)} - T_{-(k+2)}} \\ = & \frac{T_{-(k+1)}x_{-1}x_0 + \left(T_{-(k+2)} + T_{-(k+3)}\right)x_0 + T_{-(k+3)}}{T_{-(k+2)}x_{-1}x_0 + \left(T_{-(k+1)} - T_{-(k+3)}\right)x_0 + T_{-(k+3)}}, \end{array}$$

which ends the induction and the proof.

Theorem 6.2 Eq. (6.1) has unique positive equilibrium point $\overline{x} = \alpha$ and α is saddle point.

Proof. Equilibrium point of Eq.(6.1) satisfy the equation

$$\overline{x} = \frac{1}{\overline{x}\left(\overline{x} - 1\right) - 1}.$$

After simplification, we get the following cubic equation

$$\overline{x}^3 - \overline{x}^2 - \overline{x} - 1 = 0. \tag{6.7}$$

The cubic equation (6.7) is the characteristic equation of the recurrence relation of the Tribonacci numbers in (2.13) having the unique real root α . Therefore, the unique positive equilibrium point of Eq.(6.1) is $\overline{x} = \alpha$.

Now, we indicate that the equilibrium point of Eq.(6.1) is saddle point.

Let I be an interval of real numbers and

$$f: I^2 \to I$$

be a continuous function defined by

$$f(x,y) = \frac{1}{x(y-1) - 1}.$$

Therefore, it follows that

$$\frac{\partial f(x,y)}{\partial x} = \frac{-(y-1)}{(x(y-1)-1)^2},$$
$$\frac{\partial f(x,y)}{\partial y} = \frac{-x}{(x(y-1)-1)^2}.$$

Then, from (6.7)

$$\frac{\partial f\left(\overline{x},\overline{x}\right)}{\partial x} = \frac{-(\alpha-1)}{(\alpha(\alpha-1)-1)^2}$$

$$= \frac{1-\alpha}{(\alpha^2-\alpha-1)^2}$$

$$= \frac{1-\alpha}{\left(\frac{1}{\alpha}\right)^2}$$

$$= \alpha^2 - \alpha^3$$

$$= -(\alpha+1),$$

$$\frac{\partial f\left(\overline{x},\overline{x}\right)}{\partial y} = \frac{-\alpha}{(\alpha(\alpha-1)-1)^2}$$

$$= \frac{-\alpha}{(\alpha^2-\alpha-1)^2}$$

$$= \frac{-\alpha}{\left(\frac{1}{\alpha}\right)^2}$$

$$= -\alpha^3,$$

and the linearized equation of Eq.(6.1) about $\overline{x} = \alpha$ is

$$z_{n+1} = -(\alpha + 1) z_n + (-\alpha^3) z_{n-1}$$

or equivalently

$$z_{n+1} + (\alpha + 1) z_n + \alpha^3 z_{n-1} = 0.$$

Therefore, the corresponding characteristic polynomial is

$$\lambda^2 + (\alpha + 1)\lambda + \alpha^3 = 0.$$

Then, from Theorem (2.1), it is clearly seen that

$$\lambda_{1,2} = \frac{-(\alpha+1) \pm \sqrt{-4\alpha^3 + \alpha^2 + 2\alpha + 1}}{2}$$

and numerically

$$\begin{aligned} |\lambda_1| &= \left| \frac{-(\alpha+1) + \sqrt{-4\alpha^3 + \alpha^2 + 2\alpha + 1}}{2} \right| = 0, 11228 < 1\\ |\lambda_2| &= \left| \frac{-(\alpha+1) - \sqrt{-4\alpha^3 + \alpha^2 + 2\alpha + 1}}{2} \right| = 1, 4314 > 1. \end{aligned}$$

So, the equilibrium point α is a saddle point. This completes the proof.

6.1.2 The Difference Equation (6.2)

In this subsection, we study the Eq.(6.2), that is,

$$x_{n+1} = \frac{1}{x_n (x_{n-1} + 1) + 1}, \quad n = 0, 1, \dots$$

and examine the dynamical behavior of solutions of Eq.(6.2).

Theorem 6.3 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (6.2). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n\}_{n=-1}^{\infty}$ is given by

$$x_n = \frac{T_{n-1}x_{-1}x_0 + (T_{n+1} - T_n)x_0 + T_n}{T_n x_{-1}x_0 + (T_{n-1} + T_n)x_0 + T_{n+1}},$$
(6.8)

where T_n is the nth Tribonacci number and the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_2$, with F_2 is the forbidden set of Eq.(6.2) given by

$$F_2 = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : T_n x_{-1} x_0 + (T_{n-1} + T_n) x_0 + T_{n+1} = 0 \right\}.$$

Proof. (First proof) Now, we give the proof with an analytic approach. We make the substitution

$$x_n = \frac{t_{n-1}}{t_n} \tag{6.9}$$

in Eq.(6.2) to get the linear difference equation. Then, we have

$$t_{n+1} = t_n + t_{n-1} + t_{n-2}.$$

By using same operations in Theorem 2.1 in [86] such that a = b = c = 1, we obtain the initial values of three sequences are defined

$$a_n = aa_{n-1} + ba_{n-2} + ca_{n-3},$$

 $b_n = a_{n+1} - aa_n,$
 $c_n = ca_{n-1},$

such that

$$a_0 = 1, \quad a_{-1} = 0, \quad a_{-2} = 0,$$

 $b_0 = 0, \quad b_{-1} = 1, \quad b_{-2} = 0,$
 $c_0 = 0, \quad c_{-1} = 0, \quad c_{-2} = 1.$

$$(6.10)$$

Next, we get

$$t_n = a_n t_0 + (a_{n+1} - aa_n) t_{-1} + ca_{n-1} t_{-2}.$$

So a = b = c = 1 and from (6.9), we obtain

$$x_n = \frac{a_{n-2}x_{-1}x_0 + (a_n - a_{n-1})x_0 + a_{n-1}}{a_{n-1}x_{-1}x_0 + (a_{n+1} - a_n)x_0 + a_n}$$

or equivalently

$$x_n = \frac{a_{n-2}x_{-1}x_0 + (a_n - a_{n-1})x_0 + a_{n-1}}{a_{n-1}x_{-1}x_0 + (a_{n-1} + a_{n-2})x_0 + a_n}.$$

From initial values (6.10) and definitions of sequences a_n and T_n , we have

$$a_n = T_{n+1},$$

with the backward shifted initial values of the sequence a_n . Hence, we obtain

$$x_n = \frac{T_{n-1}x_{-1}x_0 + (T_{n+1} - T_n)x_0 + T_n}{T_n x_{-1}x_0 + (T_{n-1} + T_n)x_0 + T_{n+1}}$$

So, the proof is complete.

Proof. (Second proof) We will prove this theorem by induction on k. For k = 0, from Eq.(6.2),

$$x_{1} = \frac{1}{x_{0}(x_{-1}+1)+1} = \frac{1}{x_{-1}x_{0}+x_{0}+1} = \frac{T_{0}x_{-1}x_{0}+(T_{2}-T_{1})x_{0}+T_{1}}{T_{1}x_{-1}x_{0}+(T_{0}+T_{1})x_{0}+T_{2}}.$$

Now, we assume that

$$x_{k} = \frac{T_{k-1}x_{-1}x_{0} + (T_{k+1} - T_{k})x_{0} + T_{k}}{T_{k}x_{-1}x_{0} + (T_{k-1} + T_{k})x_{0} + T_{k+1}},$$
(6.11)

is true for all $1 \le n \le k$. Hence, we have to prove that it is true for k + 1. Taking into account (2.11) and (6.11), we have

$$\begin{aligned} x_{k+1} &= \frac{1}{x_k (x_{k-1}+1)+1} \\ &= \frac{1}{\left(\frac{T_{k-1}x_{-1}x_0 + (T_{k+1}-T_k)x_0 + T_k}{T_{k-1}x_0 + (T_{k-1}+T_k)x_0 + T_{k+1}}\right) \left(\frac{T_{k-2}x_{-1}x_0 + (T_k - T_{k-1})x_0 + T_{k-1}}{T_{k-1}x_{-1}x_0 + (T_{k-1}+T_k)x_0 + T_{k+1}} + 1\right) + 1} \\ &= \frac{T_k x_{-1}x_0 + (T_{k-1}+T_k)x_0 + T_{k+1}}{(T_{k-2}+T_{k-1}+T_k)x_{-1}x_0 + (T_k+T_{k+1})x_0 + T_{k-1}+T_k + T_{k+1}} \\ &= \frac{T_k x_{-1}x_0 + (T_{k+2}-T_{k+1})x_0 + T_{k+1}}{T_{k+1}x_{-1}x_0 + (T_k+T_{k+1})x_0 + T_{k+2}}, \end{aligned}$$

which ends the induction and the proof.

Proof. (Third proof) Consider Eq.(6.2) by taking n = 0, 1, 2, ... as follows:

 $n = 0 \implies x_1 = \frac{1}{x_{-1}x_0 + x_0 + 1},$ $n = 1 \implies x_2 = \frac{x_{-1}x_0 + x_0 + 1}{x_{-1}x_0 + 2x_0 + 2},$ $n = 2 \implies x_3 = \frac{x_{-1}x_0 + 2x_0 + 2}{2x_{-1}x_0 + 3x_0 + 4},$ $n = 3 \implies x_4 = \frac{2x_{-1}x_0 + 3x_0 + 4}{4x_{-1}x_0 + 6x_0 + 7},$ $n = 4 \implies x_5 = \frac{4x_{-1}x_0 + 6x_0 + 7}{7x_{-1}x_0 + 11x_0 + 13},$ $n = 5 \implies x_6 = \frac{7x_{-1}x_0 + 11x_0 + 13}{13x_{-1}x_0 + 20x_0 + 24},$

If we keep on this process and also regard (2.11), then the solution in (9.6) directly follows from a simple induction.

Theorem 6.4 Eq. (6.2) has unique positive equilibrium point $\overline{x} = a$ and a is locally asymptotically stable.

Proof. Equilibrium point of Eq.(6.2) is the real roots of the equation

$$\overline{x} = \frac{1}{\overline{x}\left(\overline{x}+1\right)+1}.$$

After simplification, we get the following cubic equation

$$\overline{x}^3 + \overline{x}^2 + \overline{x} - 1 = 0. \tag{6.12}$$

Then, the roots of the cubic equation (6.12) are given by

$$a = \frac{-1 + \sqrt[3]{3\sqrt{33} + 17} - \sqrt[3]{3\sqrt{33} - 17}}{3},$$

$$b = \frac{-1 + \omega\sqrt[3]{3\sqrt{33} + 17} - \omega^2\sqrt[3]{3\sqrt{33} - 17}}{3},$$

$$c = \frac{-1 + \omega^2\sqrt[3]{3\sqrt{33} + 17} - \omega\sqrt[3]{3\sqrt{33} - 17}}{3},$$

where

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. So, the root a is only real number. Therefore, the unique positive equilibrium point of Eq.(6.2) is $\overline{x} = a$.

Now, we show that the equilibrium point of Eq.(6.2) is locally asymptotically stable.
Let I be an interval of real numbers and consider the function

$$f: I^2 \to I$$

defined by

$$f(x,y) = \frac{1}{x(y+1)+1}.$$

The linearized equation of Eq.(6.2) about the equilibrium point $\overline{x} = a$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where

$$p = \frac{\partial f(\overline{x}, \overline{x})}{\partial x} = \frac{-(a+1)}{(a(a+1)+1)^2} = a-1$$
$$q = \frac{\partial f(\overline{x}, \overline{x})}{\partial y} = \frac{-a}{(a(a+1)+1)^2} = -a^3,$$

and the corresponding characteristic equation is

$$\lambda^{2} + (1 - a)\lambda + a^{3} = 0.$$

Therefore, from Theorem 2.1, it is easily seen that

$$\lambda_{1,2} = \frac{a - 1 \pm \sqrt{-4a^3 + a^2 - 2a + 1}}{2}$$

and numerically

$$|\lambda_1| = |\lambda_2| = 0,40089 < 1.$$

This completes the proof.

Theorem 6.5 The equilibrium point of Eq.(6.2) is globally asymptotically stable.

Proof. Let $\{x_n\}_{n\geq -1}$ be a solution of Eq.(6.2). By Theorem 6.4, we need only to prove that the equilibrium point *a* is global attractor, that is

 $\lim_{n \to \infty} x_n = a.$

From Theorem 6.3 and (2.13) and (2.14), it follows that

$$\lim_{n \to \infty} x_n = \lim_{n \to \infty} \frac{T_{n-1}x_{-1}x_0 + (T_{n+1} - T_n)x_0 + T_n}{T_n x_{-1}x_0 + (T_{n-1} + T_n)x_0 + T_{n+1}}$$

$$= \lim_{n \to \infty} \frac{T_{n-1}\left(x_{-1}x_0 + \left(\frac{T_{n+1}}{T_{n-1}} - \frac{T_n}{T_{n-1}}\right)x_0 + \frac{T_n}{T_{n-1}}\right)}{T_n\left(x_{-1}x_0 + \left(\frac{T_{n-1}}{T_n} + 1\right)x_0 + \frac{T_{n+1}}{T_n}\right)}$$

$$= \left(\frac{x_{-1}x_0 + (\alpha^2 - \alpha)x_0 + \alpha}{x_{-1}x_0 + \left(\frac{1}{\alpha} + 1\right)x_0 + \alpha}\right)\lim_{n \to \infty} \frac{T_{n-1}}{T_n}$$

$$= \lim_{n \to \infty} \frac{T_{n-1}}{T_n}$$

$$= a.$$

The proof is complete.

6.1.3 The Difference Equation (6.3)

In this subsection, we take into account the Eq.(6.3), that is,

$$x_{n+1} = \frac{-1}{x_n (x_{n-1} - 1) + 1}, \quad n = 0, 1, ...,$$

and analyze the dynamical behavior of solutions of Eq.(6.3).

Theorem 6.6 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (6.3). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n\}_{n=-1}^{\infty}$ is given by

$$x_n = \frac{-(T_{n-1}x_{-1}x_0 + (T_n - T_{n+1})x_0 + T_n)}{T_n x_{-1} x_0 - (T_{n-1} + T_n)x_0 + T_{n+1}},$$
(6.13)

where T_n is the nth Tribonacci number and the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_3$, with F_3 is the forbidden set of Eq.(6.3) given by

$$F_3 = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : T_n x_{-1} x_0 - (T_{n-1} + T_n) x_0 + T_{n+1} = 0 \right\}.$$

Proof. (First proof) We will prove this theorem by induction on k. For k = 0, from Eq.(6.3),

$$x_{1} = \frac{-1}{x_{0}(x_{-1}-1)+1} = \frac{-1}{x_{-1}x_{0}-x_{0}+1} = \frac{-(T_{0}x_{-1}x_{0}+(T_{1}-T_{2})x_{0}+T_{1})}{T_{1}x_{-1}x_{0}-(T_{0}+T_{1})x_{0}+T_{2}}$$

Now, we assume that

$$x_{k} = \frac{-(T_{k-1}x_{-1}x_{0} + (T_{k} - T_{k+1})x_{0} + T_{k})}{T_{k}x_{-1}x_{0} - (T_{k-1} + T_{k})x_{0} + T_{k+1}},$$
(6.14)

is true for all $1 \le n \le k$. Hence, we have to prove that it is true for k + 1. Taking into account (2.11) and (6.14), we have

$$\begin{aligned} x_{k+1} &= \frac{-1}{x_k \left(x_{k-1} - 1\right) + 1} \\ &= \frac{-1}{\left(\frac{-(T_{k-1}x_{-1}x_0 + (T_k - T_{k+1})x_0 + T_k)}{T_{kx-1}x_0 - (T_{k-1} + T_k)x_0 + T_{k+1}}\right) \left(\frac{-(T_{k-2}x_{-1}x_0 + (T_{k-1} - T_k)x_0 + T_{k-1})}{T_{k-1}x_{-1}x_0 - (T_{k-1} + T_k) x_0 + T_{k+1}} - 1\right) + 1} \\ &= \frac{-(T_k x_{-1}x_0 - (T_{k-1} + T_k) x_0 + T_{k+1})}{(T_{k-2} + T_{k-1} + T_k) x_{-1}x_0 - (T_k + T_{k+1}) x_0 + T_{k-1} + T_k + T_{k+1}} \\ &= \frac{-(T_k x_{-1}x_0 + (T_{k+1} - T_{k+2}) x_0 + T_{k+1})}{T_{k+1}x_{-1}x_0 - (T_k + T_{k+1}) x_0 + T_{k+2}}, \end{aligned}$$

which ends the induction and the proof.

Proof. (Second proof) Consider Eq.(6.3) by taking n = 0, 1, 2, ... as follows:

$$n = 0 \implies x_1 = \frac{-1}{x_{-1}x_0 - x_0 + 1},$$

$$n = 1 \implies x_2 = \frac{-(x_{-1}x_0 - x_0 + 1)}{x_{-1}x_0 - 2x_0 + 2},$$

$$n = 2 \implies x_3 = \frac{-(x_{-1}x_0 - 2x_0 + 2)}{2x_{-1}x_0 - 3x_0 + 4},$$

$$n = 3 \implies x_4 = \frac{-(2x_{-1}x_0 - 3x_0 + 4)}{4x_{-1}x_0 - 6x_0 + 7},$$

$$n = 4 \implies x_5 = \frac{-(4x_{-1}x_0 - 6x_0 + 7)}{7x_{-1}x_0 - 11x_0 + 13},$$

$$n = 5 \implies x_6 = \frac{-(7x_{-1}x_0 - 11x_0 + 13)}{13x_{-1}x_0 - 20x_0 + 24}$$

If we keep on this process and also regard (2.11), then the solution in (6.13) directly follows from a simple induction.

Theorem 6.7 Eq. (6.3) has unique negative equilibrium point $\overline{x} = d$ and d is locally asymptotically stable.

Proof. Equilibrium point of Eq.(6.3) is the real roots of the equation

$$\overline{x} = \frac{-1}{\overline{x}\left(\overline{x} - 1\right) + 1}.$$

After simplification, we get the following cubic equation

$$\overline{x}^3 - \overline{x}^2 + \overline{x} + 1 = 0. \tag{6.15}$$

Then, the roots of the cubic equation (6.15) are given by

$$d = \frac{1 + \sqrt[3]{3\sqrt{33} - 17} - \sqrt[3]{3\sqrt{33} + 17}}{3},$$

$$e = \frac{1 + \omega\sqrt[3]{3\sqrt{33} - 17} - \omega^2\sqrt[3]{3\sqrt{33} + 17}}{3},$$

$$f = \frac{1 + \omega^2\sqrt[3]{3\sqrt{33} - 17} - \omega\sqrt[3]{3\sqrt{33} + 17}}{3},$$

where

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. So, the root d is only real number. Therefore, the unique negative equilibrium point of Eq.(6.3) is $\overline{x} = d$.

Now, we show that the unique negative equilibrium point of Eq.(6.3) is locally asymptotically stable.

Let $I = (0, \infty)$ and consider the function

$$f: I^2 \to I$$

defined by

$$f(x,y) = \frac{-1}{x(y-1)+1}$$

The linearized equation of Eq.(6.3) about the equilibrium point $\overline{x} = d$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where, from (6.15),

$$p = \frac{\partial f(\overline{x}, \overline{x})}{\partial x} = \frac{d-1}{(d(d-1)+1)^2} = \frac{d-1}{(d^2-d+1)^2} = \frac{d-1}{(-\frac{1}{d})^2} = -(d+1),$$

$$q = \frac{\partial f(\overline{x}, \overline{x})}{\partial y} = \frac{d}{(d(d-1)+1)^2} = \frac{d}{(d^2-d+1)^2} = \frac{d}{(-\frac{1}{d})^2} = d^3,$$

and the corresponding characteristic equation is

$$\lambda^{2} + (d+1)\,\lambda - d^{3} = 0.$$

Therefore, from Theorem 2.1, it is easily seen that

$$\lambda_{1,2} = \frac{-(d+1) \pm \sqrt{4d^3 + d^2 + 2d + 1}}{2}$$

and numerically

$$|\lambda_1| = |\lambda_2| = 0,40089 < 1.$$

So, this completes the proof.

Theorem 6.8 The equilibrium point of Eq. (6.3) is globally asymptotically stable.

Proof. Let $\{x_n\}_{n\geq -1}$ be a solution of Eq.(6.3). By Theorem 6.7, we need only to prove that the equilibrium point d is global attractor, that is

 $\lim_{n \to \infty} x_n = d.$

From Theorem 6.6 and (2.13) and (2.14), it follows that

$$\lim_{n \to \infty} x_n = \lim_{n \to \infty} \frac{-(T_{n-1}x_{-1}x_0 + (T_n - T_{n+1})x_0 + T_n)}{T_n x_{-1} x_0 - (T_{n-1} + T_n)x_0 + T_{n+1}}$$

$$= \lim_{n \to \infty} \frac{-(T_{n-1}\left(x_{-1}x_0 + \left(\frac{T_n}{T_{n-1}} - \frac{T_{n+1}}{T_{n-1}}\right)x_0 + \frac{T_n}{T_{n-1}}\right))}{T_n\left(x_{-1}x_0 - \left(\frac{T_{n-1}}{T_n} + 1\right)x_0 + \frac{T_{n+1}}{T_n}\right)}$$

$$= \left(\frac{x_{-1}x_0 + (\alpha - \alpha^2)x_0 + \alpha}{x_{-1}x_0 - \left(\frac{1}{\alpha} + 1\right)x_0 + \alpha}\right)\lim_{n \to \infty} \frac{-T_{n-1}}{T_n}$$

$$= \lim_{n \to \infty} \frac{-T_{n-1}}{T_n}$$

$$= d.$$

The proof is complete.

6.1.4 The Difference Equation (6.4)

In this subsection, we take into account the Eq.(6.4), that is,

$$x_{n+1} = \frac{-1}{x_n \left(x_{n-1} + 1\right) - 1}, \quad n = 0, 1, \dots,$$

and analyze the dynamical behavior of solutions of Eq.(6.4).

Theorem 6.9 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (6.4). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n\}_{n=0-1}^{\infty}$ is given by

$$x_{n} = \frac{-\left(T_{-n}x_{-1}x_{0} - \left(T_{-(n+1)} + T_{-(n+2)}\right)x_{0} + T_{-(n+1)}\right)}{T_{-(n+1)}x_{-1}x_{0} + \left(T_{-(n+1)} - T_{-n}\right)x_{0} + T_{-(n+2)}},$$
(6.16)

where T_n is the nth Tribonacci number and the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_4$, with F_4 is the forbidden set of Eq.(6.4) given by

$$F_4 = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : T_{-(n+1)} x_{-1} x_0 + (T_{-(n+1)} - T_{-n}) x_0 + T_{-(n+2)} = 0 \right\}$$

Proof. We will prove this theorem by induction on k. For k = 0, from Eq.(6.4),

$$x_{1} = \frac{-1}{x_{0}(x_{-1}+1)-1} = \frac{-1}{x_{-1}x_{0}+x_{0}-1} = \frac{-(T_{-1}x_{-1}x_{0}(T_{-2}+T_{-3})x_{0}+T_{-2})}{T_{-2}x_{-1}x_{0}+(T_{-2}-T_{-1})x_{0}+T_{-3}}.$$

Now, we assume that

$$x_{k} = \frac{-\left(T_{-k}x_{-1}x_{0} - \left(T_{-(k+1)} + T_{-(k+2)}\right)x_{0} + T_{-(k+1)}\right)}{T_{-(k+1)}x_{-1}x_{0} + \left(T_{-(k+1)} - T_{-k}\right)x_{0} + T_{-(k+2)}}$$
(6.17)

is true for all $1 \le n \le k$. Hence, we have to prove that it is true for k + 1. Taking into account (2.12) and (6.17), we have

$$\begin{array}{ll} x_{k+1} \\ = & \frac{-1}{x_k \left(x_{k-1}+1\right)-1} \\ = & \frac{-1}{\left(\frac{-\left(T_{-k}x_{-1}x_0-\left(T_{-(k+1)}+T_{-(k+2)}\right)x_0+T_{-(k+1)}\right)}{T_{-(k+1)}x_{-1}x_0+\left(T_{-(k+1)}-T_{-k}\right)x_0+T_{-(k+2)}\right)} \left(\frac{-\left(T_{-(k-1)}x_{-1}x_0-\left(T_{-k}+T_{-(k+1)}\right)x_0+T_{-k}\right)}{T_{-k}x_{-1}x_0+\left(T_{-k}-T_{-(k-1)}\right)x_0+T_{-(k+1)}\right)}+1\right)-1 \\ = & \frac{-\left(T_{-(k+1)}x_{-1}x_0-\left(T_{-(k+2)}+T_{-(k+3)}\right)x_0+T_{-(k+2)}\right)}{\left(T_{-(k-1)}-T_{-k}-T_{-(k+1)}\right)x_{-1}x_0+\left(T_{-(k+2)}-T_{-(k+1)}\right)x_0+T_{-k}-T_{-(k+1)}-T_{-(k+2)}}{1} \\ = & \frac{-\left(T_{-(k+1)}x_{-1}x_0-\left(T_{-(k+2)}+T_{-(k+3)}\right)x_0+T_{-(k+2)}\right)}{T_{-(k+2)}x_{-1}x_0+\left(T_{-(k+2)}-T_{-(k+3)}\right)x_0+T_{-(k+3)}}, \end{array}$$

which ends the induction and the proof.

Theorem 6.10 Eq.(6.4) has unique negative equilibrium point $\overline{x} = a$ and a unstable.

Proof. Equilibrium point of Eq.(6.4) satisfy the equation

$$\overline{x} = \frac{-1}{\overline{x}\left(\overline{x}+1\right) - 1}.$$

After simplification, we get the following cubic equation

$$\overline{x}^3 + \overline{x}^2 - \overline{x} + 1 = 0. \tag{6.18}$$

Then, the roots of the cubic equation (6.18) are given by

$$g = \frac{-1 - \sqrt[3]{19 + 3\sqrt{33}} - \sqrt[3]{19 - 3\sqrt{33}}}{3},$$

$$h = \frac{-1 - \omega\sqrt[3]{19 + 3\sqrt{33}} - \omega^2\sqrt[3]{19 - 3\sqrt{33}}}{3},$$

$$k = \frac{-1 - \omega^2\sqrt[3]{19 + 3\sqrt{33}} - \omega\sqrt[3]{19 - 3\sqrt{33}}}{3},$$

where

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. So, the root g is only real number. Therefore, the unique negative equilibrium point of Eq.(6.4) is $\overline{x} = g$.

Now, we indicate that the negative equilibrium point of Eq.(6.4) is unstable.

Let I be an interval of real numbers and

$$f: I^2 \to I$$

be a continuous function defined by

$$f(x,y) = \frac{-1}{x(y+1) - 1}$$

Therefore, it follows that

$$\frac{\partial f(x,y)}{\partial x} = \frac{y+1}{(x(y+1)-1)^2},$$
$$\frac{\partial f(x,y)}{\partial y} = \frac{x}{(x(y+1)-1)^2}.$$

Then, from (6.18)

$$\frac{\partial f\left(\overline{x},\overline{x}\right)}{\partial x} = \frac{g+1}{\left(g\left(g+1\right)-1\right)^2}$$

$$= \frac{g+1}{\left(g^2+g-1\right)^2}$$

$$= \frac{g+1}{\left(-\frac{1}{g}\right)^2}$$

$$= g^3+g^2$$

$$= g-1,$$

$$\frac{\partial f\left(\overline{x},\overline{x}\right)}{\partial y} = \frac{g}{\left(g\left(g+1\right)-1\right)^2}$$

$$= \frac{g}{\left(g^2+g-1\right)^2}$$

$$= \frac{g}{\left(-\frac{1}{g}\right)^2}$$

$$= g^3,$$

and the linearized equation of Eq.(6.4) about $\overline{x} = g$ is

$$z_{n+1} = (g-1) z_n + g^3 z_{n-1}$$

or equivalently

$$z_{n+1} - (g-1) z_n - g^3 z_{n-1} = 0$$

Therefore, the corresponding characteristic polynomial is

$$\lambda^2 - (g - 1)\lambda - g^3 = 0.$$

Then, from Theorem 2.1, it is clearly seen that

$$\lambda_{1,2} = \frac{(g-1) \pm \sqrt{4g^3 + g^2 - 2g + 1}}{2}$$

and numerically

$$|\lambda_1| = |\lambda_2| = 2,4944 > 1$$

So, the equilibrium point g is unstable. This completes the proof.



CHAPTER 7

THE DYNAMICS OF EXACT SOLUTIONS OF SECOND ORDER NONLINEAR FOUR DIFFERENCE EQUATIONS

We mention that the results of this chapter are cited from [102] which has been published by us.

The purpose of this chapter is to determine the forms of solutions, the stability character of equilibrium points, the periodicity of solutions and global behavior of solutions of the following four difference equations

$$x_{n+1} = \frac{1}{x_n \left(x_{n-1} + 1\right) - 1}, \quad n = 0, 1, \dots,$$
(7.1)

$$x_{n+1} = \frac{-1}{x_n \left(x_{n-1} - 1\right) - 1}, \quad n = 0, 1, \dots,$$
(7.2)

$$x_{n+1} = \frac{1}{x_n \left(x_{n-1} - 1\right) + 1}, \quad n = 0, 1, \dots,$$
(7.3)

$$x_{n+1} = \frac{-1}{x_n \left(x_{n-1} + 1\right) + 1}, \quad n = 0, 1, \dots$$
(7.4)

7.1 MAIN RESULTS

In this section, we present our main results for the above mentioned difference equations. Our aim is to investigate the general solutions in explicit form of the above mentioned difference equations and the asymptotic behavior of solutions of these difference equations.

7.1.1 The Difference Equation (7.1)

Theorem 7.1 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (7.1). Then, for n = 0, 1, 2, ..., the forms of solutions $\{x_n\}_{n=-1}^{\infty}$ are given by

$$x_{2n-1} = \frac{(1-n)x_{-1}x_{0} + n}{nx_{-1}x_{0} + x_{0} - n}$$
(7.5)

$$x_{2n} = \frac{nx_{-1}x_{0} + x_{0}}{-nx_{-1}x_{0} + n + 1}$$
(7.6)

where the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_1$, with F_1 is the forbidden set of Eq.(7.1) given by

$$F_1 = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : nx_{-1}x_0 + x_0 - n = 0 \text{ or } -nx_{-1}x_0 + n + 1 = 0 \right\}.$$

Proof. For n = 0 the result holds. Assume that n > 0 and that our assumption holds for n - 1. That is,

$$x_{2n-3} = \frac{(2-n)x_{-1}x_0 + n - 1}{(n-1)x_{-1}x_0 + x_0 - (n-1)}$$

and

$$x_{2n-2} = \frac{(n-1)x_{-1}x_0 + x_0 - (n-1)}{(1-n)x_{-1}x_0 + n}$$

From this and from Eq.(7.1), it follows that

$$x_{2n-1} = \frac{1}{x_{2n-2} (x_{2n-3} + 1) - 1}$$

= $\frac{1}{\frac{(n-1)x_{-1}x_0 + x_0 - (n-1)}{(1-n)x_{-1}x_0 + n} \left(\frac{(2-n)x_{-1}x_0 + n-1}{(n-1)x_{-1}x_0 + x_0 - (n-1)} + 1\right) - 1}$
= $\frac{(1-n)x_{-1}x_0 + n}{nx_{-1}x_0 + x_0 - n}$.

Hence, similarly, we obtain

$$x_{2n} = \frac{1}{x_{2n-1}(x_{2n-2}+1)-1}$$

= $\frac{1}{\frac{(1-n)x_{-1}x_{0}+n}{nx_{-1}x_{0}+x_{0}-n}\left(\frac{(n-1)x_{-1}x_{0}+x_{0}-(n-1)}{(1-n)x_{-1}x_{0}+n}+1\right)-1}$
= $\frac{nx_{-1}x_{0}+x_{0}-n}{-nx_{-1}x_{0}+n+1}$.

Theorem 7.2 The following statements are true.

- (i) The equilibrium points of Eq.(7.1) are $\overline{x}_1 = 1$ and $\overline{x}_2 = -1$.
- (ii) The positive equilibrium point of Eq.(7.1), $\overline{x}_1 = 1$, is nonhyperbolic point.
- (iii) The negative equilibrium point of Eq. (7.1), $\overline{x}_2 = -1$, is nonhyperbolic point.

Proof.

(i) Equilibrium points of Eq.(7.1) satisfy the equation

$$\overline{x} = \frac{1}{\overline{x}\left(\overline{x}+1\right) - 1}.$$

After simplification, we have the following cubic equation

$$\overline{x}^3 + \overline{x}^2 - \overline{x} - 1 = 0. \tag{7.7}$$

The roots of the cubic equation (7.7) are -1, -1, 1. Therefore, Eq.(7.1) has two equilibra, one positive and one negative, such that

$$\overline{x}_1 = 1, \ \overline{x}_2 = -1.$$

(ii) Now, let $I = (0, \infty)$ and consider the function

$$f: I^2 \to I$$

defined by

$$f(x,y) = \frac{1}{x(y+1) - 1}.$$
(7.8)

Then, it follows that

$$\frac{\partial f(x,y)}{\partial x} = \frac{-(y+1)}{(x(y+1)-1)^2},$$
$$\frac{\partial f(x,y)}{\partial y} = \frac{-x}{(x(y+1)-1)^2}.$$

Therefore, the linearized equation of Eq.(7.1) about the equilibrium point $\overline{x}_1 = 1$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where

$$\begin{array}{rcl} p & = & \displaystyle \frac{\partial f\left(\overline{x}_{1},\overline{x}_{1}\right)}{\partial x} = -2, \\ q & = & \displaystyle \frac{\partial f\left(\overline{x}_{1},\overline{x}_{1}\right)}{\partial y} = -1, \end{array}$$

and the corresponding characteristic equation is

 $\lambda^2 + 2\lambda + 1 = 0.$

Therefore, from Theorem 2.1, it is clearly seen that

$$\lambda_{1,2} = -1$$

and

$$|\lambda_1| = |\lambda_2| = 1.$$

So, \overline{x}_1 is nonhyperbolic point.

(iii) Similarly, from (7.8), the linearized equation of Eq.(7.1) about the equilibrium point $\overline{x}_2 = -1$ is

 $z_{n+1} = pz_n + qz_{n-1},$

where

$$p = \frac{\partial f(\overline{x}_2, \overline{x}_2)}{\partial x} = 0,$$

$$q = \frac{\partial f(\overline{x}_2, \overline{x}_2)}{\partial y} = 1,$$

and its characteristic equation is

$$\lambda^2 - 1 = 0.$$

Thus, it follows that

 $\lambda_{1,2} = \pm 1$

and

$$|\lambda_1| = |\lambda_2| = 1.$$

So, \overline{x}_2 is nonhyperbolic point.

Theorem 7.3 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq.(7.1). Then, the negative equilibrium point of Eq.(7.1), \overline{x}_2 , is a global attractor.

Proof. From Theorem 7.1, we have

$$\lim_{n \to \infty} x_{2n-1} = \lim_{n \to \infty} \frac{(1-n)x_{-1}x_0 + n}{nx_{-1}x_0 + x_0 - n}$$
$$= \lim_{n \to \infty} \frac{(1-n)\left(x_{-1}x_0 + \frac{n}{1-n}\right)}{n\left(x_{-1}x_0 + \frac{x_0}{n} - 1\right)}$$
$$= \lim_{n \to \infty} \frac{(1-n)\left(x_{-1}x_0 - 1 + \frac{1}{1-n}\right)}{n\left(x_{-1}x_0 + \frac{x_0}{n} - 1\right)}$$
$$= -1,$$

and

$$\lim_{n \to \infty} x_{2n} = \lim_{n \to \infty} \frac{nx_{-1}x_0 + x_0 - n}{-nx_{-1}x_0 + n + 1}$$
$$= \lim_{n \to \infty} \frac{n\left(x_{-1}x_0 + \frac{x_0}{n+1} - 1\right)}{-n\left(x_{-1}x_0 - 1 - \frac{1}{n}\right)}$$
$$= -1.$$

Hereby, it implies

$$\lim_{n \to \infty} x_n = -1.$$

7.1.2 The Difference Equation (7.2)

Theorem 7.4 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (7.2). Then, for n = 0, 1, 2, ..., the forms of solutions $\{x_n\}_{n=-1}^{\infty}$ are given by

$$x_{2n-1} = \frac{-((1-n)x_{-1}x_0+n)}{nx_{-1}x_0-x_0-n}$$
(7.9)

$$x_{2n} = \frac{-(nx_{-1}x_0 - x_0 - n)}{-nx_{-1}x_0 + n + 1}$$
(7.10)

where the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_2$, with F_2 is the forbidden set of Eq.(7.2) given by

$$F_2 = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : nx_{-1}x_0 - x_0 - n = 0 \text{ or } -nx_{-1}x_0 + n + 1 = 0 \right\}.$$

Proof. For n = 0 the result holds. Assume that n > 0 and that our assumption holds for n - 1. That is,

$$x_{2n-3} = \frac{-((2-n)x_{-1}x_0 + n - 1)}{(n-1)x_{-1}x_0 - x_0 - (n-1)}$$

and

$$x_{2n-2} = \frac{-((n-1)x_{-1}x_0 - x_0 - (n-1))}{-(n-1)x_{-1}x_0 + n}.$$

From this and from Eq.(7.2), it follows that

$$\begin{aligned} x_{2n-1} &= \frac{-1}{x_{2n-2} (x_{2n-3} - 1) - 1} \\ &= \frac{-1}{\frac{-((n-1)x_{-1}x_0 - x_0 - (n-1))}{-(n-1)x_{-1}x_0 + n} \left(\frac{-((2-n)x_{-1}x_0 + n-1)}{(n-1)x_{-1}x_0 - x_0 - (n-1)} - 1\right) - 1} \\ &= \frac{-((1-n)x_{-1}x_0 + n)}{nx_{-1}x_0 - x_0 - n}. \end{aligned}$$

Hence, similarly, we obtain

$$\begin{aligned} x_{2n} &= \frac{-1}{x_{2n-1}(x_{2n-2}-1)-1} \\ &= \frac{-1}{\frac{-((1-n)x_{-1}x_{0}+n)}{nx_{-1}x_{0}-x_{0}-n} \left(\frac{-((n-1)x_{-1}x_{0}-x_{0}-(n-1))}{-(n-1)x_{-1}x_{0}+n}-1\right)-1} \\ &= \frac{-(nx_{-1}x_{0}-x_{0}-n)}{-nx_{-1}x_{0}+n+1}. \end{aligned}$$

Theorem 7.5 The following statements are true.

- (i) The equilibrium points of Eq. (7.2) are $\overline{x}_1 = 1$ and $\overline{x}_2 = -1$.
- (ii) The positive equilibrium point of Eq. (7.2), $\overline{x}_1 = 1$, is nonhyperbolic point.
- (iii) The negative equilibrium point of Eq. (7.2), $\overline{x}_2 = -1$, is nonhyperbolic point.

Proof.

(i) Equilibrium points of Eq.(7.2) satisfy the equation

$$\overline{x} = \frac{-1}{\overline{x}\left(\overline{x} - 1\right) - 1}.$$

After simplification, we have the following cubic equation

$$\overline{x}^3 - \overline{x}^2 - \overline{x} + 1 = 0. \tag{7.11}$$

The roots of the cubic equation (7.11) are -1, 1, 1. Therefore, Eq.(7.2) has two equilibra, one positive and one negative, such that

 $\overline{x}_1 = 1, \ \overline{x}_2 = -1.$

(ii) Now, let $I = (0, \infty)$ and consider the function

$$f: I^2 \to I$$

defined by

$$f(x,y) = \frac{-1}{x(y-1)-1}.$$
(7.12)

Then, it follows that

$$\frac{\partial f(x,y)}{\partial x} = \frac{(y-1)}{(x(y-1)-1)^2},$$
$$\frac{\partial f(x,y)}{\partial y} = \frac{x}{(x(y-1)-1)^2}.$$

Therefore, the linearized equation of Eq.(7.2) about the equilibrium point $\overline{x}_1 = 1$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where

$$p = \frac{\partial f(\overline{x}_1, \overline{x}_1)}{\partial x} = 0,$$
$$q = \frac{\partial f(\overline{x}_1, \overline{x}_1)}{\partial y} = 1,$$

and the corresponding characteristic equation is

$$\lambda^2 - 1 = 0.$$

Therefore, from Theorem 2.1, it is clearly seen that

 $\lambda_{1,2}=\pm 1$

and

$$|\lambda_1| = |\lambda_2| = 1.$$

So, \overline{x}_1 is nonhyperbolic point.

(iii) Similarly, from (7.12), the linearized equation of Eq.(7.2) about the equilibrium point $\overline{x}_2 = -1$ is

 $z_{n+1} = pz_n + qz_{n-1},$

where

$$p = \frac{\partial f(\overline{x}_2, \overline{x}_2)}{\partial x} = -2,$$

$$q = \frac{\partial f(\overline{x}_2, \overline{x}_2)}{\partial y} = -1,$$

and its characteristic equation is

$$\lambda^2 + 2\lambda + 1 = 0.$$

Thus, it follows that

$$\lambda_{1,2} = -1$$

and

$$|\lambda_1| = |\lambda_2| = 1.$$

So, \overline{x}_2 is nonhyperbolic point.

Theorem 7.6 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq.(7.2). Then, the positive equilibrium point of Eq.(7.2), \overline{x}_1 , is a global attractor.

Proof. From Theorem 7.4, we obtain

$$\lim_{n \to \infty} x_{2n-1} = \lim_{n \to \infty} \frac{-((1-n)x_{-1}x_0 + n)}{nx_{-1}x_0 - x_0 - n}$$
$$= \lim_{n \to \infty} \frac{(n-1)\left(x_{-1}x_0 + \frac{n}{1-n}\right)}{n\left(x_{-1}x_0 - \frac{x_0}{n} - 1\right)}$$
$$= \lim_{n \to \infty} \frac{(n-1)\left(x_{-1}x_0 - 1 + \frac{1}{1-n}\right)}{n\left(x_{-1}x_0 - \frac{x_0}{n} - 1\right)}$$
$$= 1,$$

and

$$\lim_{n \to \infty} x_{2n} = \lim_{n \to \infty} \frac{-(nx_{-1}x_0 - x_0 - n)}{-nx_{-1}x_0 + n + 1}$$
$$= \lim_{n \to \infty} \frac{-n(x_{-1}x_0 - \frac{x_0}{n} - 1)}{-n(x_{-1}x_0 - 1 - \frac{1}{n})}$$
$$= 1.$$

Herewith, it implies

$$\lim_{n \to \infty} x_n = 1.$$

So, the proof is complete.

7.1.3 The Difference Equation (7.3)

Lemma 7.7 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq.(7.3). Then, $\{x_n\}_{n=-1}^{\infty}$ is periodic with period four.

Proof. From Eq.(7.3),

$$x_{n+4} = \frac{1}{x_{n+3}(x_{n+2}-1)+1}$$

= $\frac{1}{\left(\frac{1}{x_{n+2}(x_{n+1}-1)+1}\right)\left(\frac{1}{x_{n+1}(x_n-1)+1}-1\right)+1}$

$$= \frac{1}{\left(\frac{1}{\left(\frac{1}{x_{n+1}(x_n-1)+1}\right)\left(\frac{1}{x_n(x_{n-1}-1)+1}-1\right)+1}\right)\left(\frac{1}{\left(\frac{1}{x_n(x_{n-1}-1)+1}\right)(x_n-1)+1}-1\right)+1}}{\left(\frac{1}{\left(\frac{1}{x_n(x_{n-1}-1)+1}\right)\left(\frac{1}{x_n(x_{n-1}-1)+1}\right)\left(\frac{1}{x_n(x_{n-1}-1)+1}\right)+1}\right)\left(\frac{1-x_n}{x_nx_{n-1}}\right)+1}}{\left(\frac{1}{x_n(x_{n-1}-1)+1}\right)\left(\frac{1}{x_nx_{n-1}}\right)+1}{x_{n-1}\left(\frac{1-x_n}{x_nx_{n-1}}\right)+1}}$$

= x_n .

Hence, the result holds.

Theorem 7.8 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq. (7.3). Then, for n = 1, 2, ...,

$$\begin{aligned}
x_{4n-3} &= \frac{1}{x_{-1}x_0 - x_0 + 1} \\
x_{4n-2} &= \frac{x_{-1}x_0 - x_0 + 1}{x_{-1}x_0} \\
x_{4n-1} &= x_{-1} \\
x_{4n} &= x_0
\end{aligned}$$
(7.13)

where the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_3$, with F_3 is the forbidden set of Eq.(7.3) given by

$$F_3 = \left\{ (x_{-1}, x_0) : x_{-1} x_0 = 0 \text{ or } x_{-1} = \frac{x_0 - 1}{x_0} \right\}.$$

Proof. From (7.3), for n = 0, the result holds. Suppose that n > 0 and that our assumption holds for n - 1. That is,

$$\begin{aligned}
x_{4n-7} &= \frac{1}{x_{-1}x_0 - x_0 + 1}, \\
x_{4n-6} &= \frac{x_{-1}x_0 - x_0 + 1}{x_{-1}x_0}, \\
x_{4n-5} &= x_{-1}, \\
x_{4n-4} &= x_0.
\end{aligned}$$

Now, from Eq.(7.3), it follows that

$$x_{4n-3} = \frac{1}{x_{4n-4} \left(x_{4n-5} - 1 \right) + 1} = \frac{1}{x_{-1} x_0 - x_0 + 1}$$

From this and from Eq.(7.3), it follows that

$$x_{4n-2} = \frac{1}{x_{4n-3}(x_{4n-4}-1)+1} = \frac{1}{\frac{1}{x_{-1}x_0 - x_0 + 1}(x_0 - 1) + 1} = \frac{x_{-1}x_0 - x_0 + 1}{x_{-1}x_0}$$

Again from Eq.(7.3), we get

$$x_{4n-1} = \frac{1}{x_{4n-2} \left(x_{4n-3} - 1 \right) + 1} = \frac{1}{\frac{x_{-1} x_0 - x_0 + 1}{x_{-1} x_0} \left(\frac{1}{x_{-1} x_0 - x_0 + 1} - 1 \right) + 1} = \frac{x_{-1} x_0}{x_0} = x_{-1}.$$

Similarly, from Eq.(7.3), we have

$$\begin{aligned} x_{4n-4} &= \frac{1}{x_{4n-1} \left(x_{4n-2} - 1 \right) + 1} = \frac{1}{x_{-1} \left(\frac{x_{-1} x_0 - x_0 + 1}{x_{-1} x_0} - 1 \right) + 1} \\ &= \frac{1}{x_{-1} - 1 + \frac{1}{x_0} - x_{-1} + 1} = x_0. \end{aligned}$$

Thus, the proof is complete.

Theorem 7.9 Eq.(7.3) has unique positive equilibrium point $\overline{x} = 1$ and 1 is nonhyperbolic point.

Proof. Equilibrium point of Eq.(7.3) satisfy the equation

$$\overline{x} = \frac{1}{\overline{x}\left(\overline{x} - 1\right) + 1}.$$

After simplification, we have the following cubic equation

$$\overline{x}^3 - \overline{x}^2 + \overline{x} - 1 = 0. \tag{7.14}$$

The roots of the cubic equation (7.14) are -i, i, 1. Therefore, the unique positive equilibrium point of Eq.(7.3) is $\overline{x} = 1$.

Now, we prove that the equilibrium point of Eq.(7.3) is nonhyperbolic.

Let $I = (0, \infty)$ and consider the function

$$f:I^2\to I$$

defined by

$$f(x,y) = \frac{1}{x(y-1)+1}.$$

The linearized equation of Eq.(7.3) about the equilibrium point $\overline{x} = 1$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where

$$p = \frac{\partial f(\overline{x}, \overline{x})}{\partial x} = 0,$$

$$q = \frac{\partial f(\overline{x}, \overline{x})}{\partial u} = -1,$$

and the corresponding characteristic equation is

$$\lambda^2 + 1 = 0.$$

Therefore, from Theorem 2.1, it is clearly seen that

$$\lambda_{1,2} = \pm i$$

and

$$|\lambda_1| = |\lambda_2| = 1.$$

So, this completes the proof.

7.1.4 The Difference Equation (7.4)

Lemma 7.10 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq.(7.4). Then, $\{x_n\}_{n=-1}^{\infty}$ is periodic with periods four.

$$\begin{aligned} x_{n+4} &= \frac{-1}{x_{n+3} (x_{n+2} + 1) + 1} \\ &= \frac{-1}{\left(\frac{-1}{x_{n+2} (x_{n+1} + 1) + 1}\right) \left(\frac{-1}{x_{n+1} (x_n + 1) + 1} + 1\right) + 1} \\ &= \frac{-1}{\left(\frac{-1}{\left(\frac{-1}{x_{n+1} (x_n + 1) + 1}\right) \left(\frac{-1}{x_n (x_{n-1} + 1) + 1}\right) \left(\frac{-1}{\left(\frac{-1}{x_n (x_{n-1} + 1) + 1}\right) (x_n + 1) + 1}\right) + 1} \right)} \\ &= \frac{-1}{\left(\frac{-1}{\left(\frac{-1}{x_n (x_{n-1} + 1) + 1} \left(\frac{x_n (x_{n-1} + 1)}{x_n x_{n-1} + 1}\right) + 1\right) \left(\frac{x_n (x_{n-1} + 1)}{x_n x_{n-1} + 1}\right) + 1}\right) \left(-\frac{x_n + 1}{x_n x_{n-1}}\right) + 1} \\ &= \frac{-1}{x_{n-1} \left(-\frac{x_n + 1}{x_n x_{n-1}}\right) + 1} \end{aligned}$$

Hence, the result holds.

Theorem 7.11 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq.(7.4). Then, for n = 1, 2, ...,

$$\begin{aligned}
x_{4n-3} &= \frac{-1}{x_{-1}x_0 + x_0 + 1} \\
x_{4n-2} &= \frac{-(x_{-1}x_0 + x_0 + 1)}{x_{-1}x_0} \\
x_{4n-1} &= x_{-1} \\
x_{4n} &= x_0
\end{aligned}$$
(7.15)

where the initial conditions x_{-1} , $x_0 \in \mathbb{R} - F_4$, with F_4 is the forbidden set of Eq.(7.4) given by

$$F_4 = \left\{ (x_{-1}, x_0) : x_{-1} x_0 = 0 \text{ or } x_{-1} = \frac{-(x_0 + 1)}{x_0} \right\}.$$

Proof. From (7.4), for n = 0, the result holds. Suppose that n > 0 and that our assumption holds for n - 1. That is,

$$\begin{aligned}
x_{4n-7} &= \frac{-1}{x_{-1}x_0 + x_0 + 1}, \\
x_{4n-6} &= \frac{-(x_{-1}x_0 + x_0 + 1)}{x_{-1}x_0}, \\
x_{4n-5} &= x_{-1}, \\
x_{4n-4} &= x_0.
\end{aligned}$$

Now, from Eq.(7.4), it follows that

$$x_{4n-3} = \frac{-1}{x_{4n-4} \left(x_{4n-5} + 1 \right) + 1} = \frac{-1}{x_{-1} x_0 + x_0 + 1}$$

From this and from Eq.(7.4), it follows that

$$x_{4n-2} = \frac{-1}{x_{4n-3}\left(x_{4n-4}+1\right)+1} = \frac{-1}{\frac{-1}{x_{-1}x_{0}+x_{0}+1}\left(x_{0}+1\right)+1} = \frac{-\left(x_{-1}x_{0}+x_{0}+1\right)}{x_{-1}x_{0}}.$$

Again from Eq.(7.4), we get

$$x_{4n-1} = \frac{-1}{x_{4n-2}\left(x_{4n-3}+1\right)+1} = \frac{-1}{\frac{-(x_{-1}x_0+x_0+1)}{x_{-1}x_0}\left(\frac{-1}{x_{-1}x_0+x_0+1}+1\right)+1} = \frac{-x_{-1}x_0}{-x_0} = x_{-1}.$$

Similarly, from Eq.(7.4), we have

$$x_{4n} = \frac{-1}{x_{4n-1}(x_{4n-2}+1)+1} = \frac{-1}{x_{-1}\left(\frac{-(x_{-1}x_0+x_0+1)}{x_{-1}x_0}+1\right)+1}$$
$$= \frac{-1}{-x_{-1}-1-\frac{1}{x_0}+x_{-1}+1} = x_0.$$

Thus, the proof is complete.

Theorem 7.12 Eq.(7.4) has unique positive equilibrium point $\overline{x} = 1$ and the equilibrium point 1 is locally assymptotically stable.

Proof. Equilibrium point of Eq.(7.4) satisfy the equation

$$\overline{x} = \frac{-1}{\overline{x}\left(\overline{x}+1\right)+1}.$$

After simplification, we have the following cubic equation

$$\overline{x}^3 + \overline{x}^2 + \overline{x} + 1 = 0. \tag{7.16}$$

The roots of the cubic equation (7.16) are -i, i, 1. Therefore, the unique positive equilibrium point of Eq.(7.4) is $\overline{x} = 1$.

Now, we demonstrate that the equilibrium point of Eq.(7.4) is locally asymptotically stable.

Let $I = (0, \infty)$ and consider the function

$$f: I^2 \to I$$

defined by

$$f(x,y) = \frac{-1}{x(y+1)+1}.$$

The linearized equation of Eq.(7.4) about the equilibrium point $\overline{x} = 1$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where

$$p = \frac{\partial f(\overline{x}, \overline{x})}{\partial x} = \frac{2}{9},$$
$$q = \frac{\partial f(\overline{x}, \overline{x})}{\partial y} = \frac{1}{9},$$

and the corresponding characteristic equation is

$$\lambda^2 - \frac{2}{9}\lambda - \frac{1}{9} = 0.$$

Therefore, from Theorem 2.1, it is clearly seen that

$$\lambda_{1,2} = \frac{1}{9} \pm \frac{1}{9}\sqrt{10}$$

and

$$|\lambda_{1,2}| < 1.$$

Thus, the proof is complete.

CHAPTER 8

GLOBAL ANALYSIS OF SOLUTIONS OF SYSTEMS OF DIFFERENCE EQUATIONS VIA TRIBONACCI NUMBERS

Primarily, we mean that the results of this chapter are cited from [103] which has been published by us.

The main objective of this chapter is to investigate the explicit form, stability character and global behavior of solutions of the following two systems of rational difference equations

$$x_{n+1} = \frac{1}{y_n \left(x_{n-1} + 1\right) + 1}, \quad y_{n+1} = \frac{1}{x_n \left(y_{n-1} + 1\right) + 1}, \quad n = 0, 1, \dots,$$
(8.1)

$$x_{n+1} = \frac{-1}{y_n \left(x_{n-1} - 1\right) + 1}, \quad y_{n+1} = \frac{-1}{x_n \left(y_{n-1} - 1\right) + 1}, \quad n = 0, 1, ...,$$
(8.2)

such that their solutions are associated with Tribonacci numbers.

Our aim in this chapter is to determine some relationships both between Tribonacci numbers and and solutions of the aforementioned systems of difference equations and between the Tribonacci constant and the equilibrium points of these systems of difference equations.

8.1 MAIN RESULTS

In this section, we introduce our results.

8.1.1 The System (8.1)

In this subsection, we present our main results related to the system (8.1). Our aim is to investigate the general solution in exact form of system (8.1) and the asymptotic behavior of solutions of system (8.1).

Theorem 8.1 Let $\{x_n, y_n\}_{n=-1}^{\infty}$ be a solution system (8.1). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n, y_n\}_{n=-1}^{\infty}$ is given by

$$x_{2n-1} = \frac{T_{2n-2}x_{-1}y_0 + (T_{2n} - T_{2n-1})y_0 + T_{2n-1}}{T_{2n-1}x_{-1}y_0 + (T_{2n-2} + T_{2n-1})y_0 + T_{2n}},$$

$$x_{2n} = \frac{T_{2n-1}y_{-1}x_0 + (T_{2n+1} - T_{2n})x_0 + T_{2n}}{T_{2n}y_{-1}x_0 + (T_{2n-1} + T_{2n})x_0 + T_{2n+1}},$$

$$y_{2n-1} = \frac{T_{2n}y_{-1}x_0 + (T_{2n-1} + T_{2n})x_0 + T_{2n+1}}{T_{2n-2}y_{-1}x_0 + (T_{2n} - T_{2n-1})x_0 + T_{2n-1}}$$
$$\frac{T_{2n-1}y_{-1}x_0 + (T_{2n-2} + T_{2n-1})x_0 + T_{2n}}{T_{2n-1}x_{-1}y_0 + (T_{2n+1} - T_{2n})y_0 + T_{2n}}$$

$$y_{2n} = \frac{1}{T_{2n}x_{-1}y_0 + (T_{2n-1} + T_{2n})y_0 + T_{2n+1}},$$

where T_n is the nth Tribonacci number and the initial conditions x_{-1} , y_{-1} , x_0 , $y_0 \in \mathbb{R} - F_1$, with F_1 is the forbidden set of system (8.1) given by

$$F_1 = \bigcup_{n=-1}^{\infty} \{ (x_{-1}, y_{-1}, x_0, y_0) : A_n = 0 \text{ or } B_n = 0 \text{ or } C_n = 0 \text{ or } D_n = 0 \}$$

where

$$A_{n} = T_{2n-1}x_{-1}y_{0} + (T_{2n-2} + T_{2n-1})y_{0} + T_{2n},$$

$$B_{n} = T_{2n}y_{-1}x_{0} + (T_{2n-1} + T_{2n})x_{0} + T_{2n+1},$$

$$C_{n} = T_{2n-1}y_{-1}x_{0} + (T_{2n-2} + T_{2n-1})x_{0} + T_{2n},$$

$$D_{n} = T_{2n}x_{-1}y_{0} + (T_{2n-1} + T_{2n})y_{0} + T_{2n+1}.$$

Proof. We use the induction on k. For k = 0, the result holds. Suppose that k > 0 and that our assumption holds for k - 1. That is,

$$\begin{aligned} x_{2k-3} &= \frac{T_{2k-4}x_{-1}y_0 + (T_{2k-2} - T_{2k-3})y_0 + T_{2k-3}}{T_{2k-3}x_{-1}y_0 + (T_{2k-4} + T_{2k-3})y_0 + T_{2k-2}}, \\ x_{k-2} &= \frac{T_{2k-3}y_{-1}x_0 + (T_{2k-1} - T_{2k-2})x_0 + T_{2k-2}}{T_{2k-2}y_{-1}x_0 + (T_{2k-3} + T_{2k-2})x_0 + T_{2k-1}}, \\ y_{2k-3} &= \frac{T_{2k-4}y_{-1}x_0 + (T_{2k-2} - T_{2k-3})x_0 + T_{2k-3}}{T_{2k-3}y_{-1}x_0 + (T_{2k-4} + T_{2k-3})x_0 + T_{2k-2}}, \\ y_{2k-2} &= \frac{T_{2k-3}x_{-1}y_0 + (T_{2k-1} - T_{2k-2})y_0 + T_{2k-2}}{T_{2k-2}x_{-1}y_0 + (T_{2k-3} + T_{2k-2})y_0 + T_{2k-1}}. \end{aligned}$$

From system (8.1) and (2.11), it follows that

$$\begin{aligned} x_{2k-1} &= \frac{1}{y_{2k-2} \left(x_{2k-3} + 1 \right) + 1} \\ &= \frac{1}{\frac{T_{2k-3} x_{-1} y_0 + (T_{2k-1} - T_{2k-2}) y_0 + T_{2k-2}}{T_{2k-2} x_{-1} y_0 + (T_{2k-3} - T_{2k-3}) y_0 + T_{2k-3}} + 1 \right) + 1} \\ &= \frac{1}{\frac{T_{2k-2} x_{-1} y_0 + (T_{2k-3} + T_{2k-2}) y_0 + T_{2k-1}}{(T_{2k-4} + T_{2k-3} + T_{2k-2}) x_0 + (T_{2k-3} + T_{2k-2}) y_0 + T_{2k-1}}} \end{aligned}$$

Therefore, we have

$$x_{2k-1} = \frac{T_{2k-2}x_{-1}y_0 + (T_{2k} - T_{2k-1})y_0 + T_{2k-1}}{T_{2k-1}x_{-1}y_0 + (T_{2k-2} + T_{2k-1})y_0 + T_{2k}}$$

And also, it follows that

$$y_{2k-1} = \frac{1}{x_{2k-2} (y_{2k-3} + 1) + 1}$$

$$= \frac{1}{\frac{T_{2k-3}y_{-1}x_{0} + (T_{2k-1} - T_{2k-2})x_{0} + T_{2k-2}}{T_{2k-2}y_{-1}x_{0} + (T_{2k-3} + T_{2k-2})x_{0} + T_{2k-1}} \left(\frac{T_{2k-4}y_{-1}x_{0} + (T_{2k-2} - T_{2k-3})x_{0} + T_{2k-3}}{T_{2k-2}y_{-1}x_{0} + (T_{2k-3} + T_{2k-3})x_{0} + T_{2k-2}} + 1\right) + 1}$$

$$= \frac{T_{2k-2}y_{-1}x_{0} + (T_{2k-3} + T_{2k-2})x_{0} + T_{2k-1}}{(T_{2k-4} + T_{2k-3} + T_{2k-2})y_{-1}x_{0} + (T_{2k-2} + T_{2k-1})x_{0} + T_{2k-3} + T_{2k-2} + T_{2k-1}}$$

So, we obtain

$$y_{2k-1} = \frac{T_{2k-2}y_{-1}x_0 + (T_{2k} - T_{2k-1})x_0 + T_{2k-1}}{T_{2k-1}y_{-1}x_0 + (T_{2k-2} + T_{2k-1})x_0 + T_{2k}}$$

Similarly, from system (8.1) and (2.11), it follows that

$$\begin{aligned} x_{2k} &= \frac{1}{y_{2k-1}(x_{2k-2}+1)+1} \\ &= \frac{1}{\frac{T_{2k-2}y_{-1}x_0 + (T_{2k}-T_{2k-1})x_0 + T_{2k-1}}{T_{2k-1}y_{-1}x_0 + (T_{2k-2})x_0 + T_{2k-2}x_{-1}} + 1 + 1} \\ &= \frac{1}{\frac{T_{2k-2}y_{-1}x_0 + (T_{2k-2}+T_{2k-1})x_0 + T_{2k}}{(T_{2k-3}+T_{2k-2}+T_{2k-1})y_{-1}x_0 + (T_{2k-2}+T_{2k-1})x_0 + T_{2k}}}. \end{aligned}$$

Thus, we get

$$x_{2k} = \frac{T_{2k-1}y_{-1}x_0 + (T_{2k-2} + T_{2k-1})x_0 + T_{2k}}{T_{2k}y_{-1}x_0 + (T_{2k-1} + T_{2k})x_0 + T_{2k+1}}.$$

And also, it follows that

$$y_{2k} = \frac{1}{x_{2k-1}(y_{2k-2}+1)+1}$$

$$= \frac{1}{\frac{T_{2k-2}x_{-1}y_{0}+(T_{2k}-T_{2k-1})y_{0}+T_{2k-1}}{T_{2k-1}x_{-1}y_{0}+(T_{2k-2}+T_{2k-1})y_{0}+T_{2k}}\left(\frac{T_{2k-3}x_{-1}y_{0}+(T_{2k-3}-T_{2k-2})y_{0}+T_{2k-2}}{T_{2k-2}x_{-1}y_{0}+(T_{2k-2}+T_{2k-1})y_{0}+T_{2k}}}\right)$$

$$= \frac{T_{2k-1}x_{-1}y_{0}+(T_{2k-2}+T_{2k-1})y_{0}+T_{2k}}{(T_{2k-3}+T_{2k-2}+T_{2k-1})x_{-1}y_{0}+(T_{2k-1}+T_{2k})y_{0}+T_{2k-2}+T_{2k-1}+T_{2k}}$$

Herefrom, we have

$$y_{2k} = \frac{T_{2k-1}x_{-1}y_0 + (T_{2k-2} + T_{2k-1})y_0 + T_{2k}}{T_{2k}x_{-1}y_0 + (T_{2k-1} + T_{2k})y_0 + T_{2k+1}}$$

Theorem 8.2 The system (8.1) has unique positive equilibrium point $(\overline{x}, \overline{y}) = (a, a)$ and (a, a) is locally asymptotically stable.

Proof. Clearly, equilibrium point of system (8.1) is the real roots of the equations

$$\overline{x} = \frac{1}{\overline{x}(\overline{y}+1)+1}, \ \overline{y} = \frac{1}{\overline{y}(\overline{x}+1)+1}.$$
(8.3)

In (8.3), after some operations, we obtain

$$\overline{x} = \overline{y}.$$

As a result, we obtain the following equation

$$\overline{x}^3 + \overline{x}^2 + \overline{x} - 1 = 0. \tag{8.4}$$

Then, the roots of the cubic equation (8.4) are given by

$$a = \frac{-1 + \sqrt[3]{3\sqrt{33} + 17} - \sqrt[3]{3\sqrt{33} - 17}}{3},$$

$$b = \frac{-1 + \omega\sqrt[3]{3\sqrt{33} + 17} - \omega^2\sqrt[3]{3\sqrt{33} - 17}}{3},$$

$$c = \frac{-1 + \omega^2\sqrt[3]{3\sqrt{33} + 17} - \omega\sqrt[3]{3\sqrt{33} - 17}}{3}$$

where

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. So, the root a is only real number. Therefore, the unique positive equilibrium point of system (8.1) is $(\overline{x}, \overline{y}) = (a, a)$.

Now, we show that the unique positive equilibrium point of system (8.1) is locally asymptotically stable.

Let I and J are some intervals of real numbers and consider the functions

$$f: I^2 \times J^2 \to I \text{ and } g: I^2 \times J^2 \to J$$

defined by

$$f(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{1}{y_n(x_{n-1}+1)+1}, \ g(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{1}{x_n(y_{n-1}+1)+1}.$$

We consider the following transformation to build corresponding linearized form of system (8.1)

$$(x_n, x_{n-1}, y_n, y_{n-1}) \to (f, f_1, g, g_1),$$

where

$$f(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{1}{y_n(x_{n-1}+1)+1},$$

$$f_1(x_n, x_{n-1}, y_n, y_{n-1}) = x_n,$$

$$g(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{1}{x_n(y_{n-1}+1)+1},$$

$$g_1(x_n, x_{n-1}, y_n, y_{n-1}) = y_n.$$

Then, the linearized system of system (8.1) about the equilibrium point (a, a) under the above transformation is given as

$$X_{n+1} = BX_n,$$

where $X_n = (x_n, x_{n-1}, y_n, y_{n-1})^T$ and B is a Jacobian matrix of system (8.1) about the equilibrium point (a, a) and given by

$$B = \begin{pmatrix} 0 & \frac{-a}{(a(a+1)+1)^2} & \frac{-(1+a)}{(a(a+1)+1)^2} & 0\\ 1 & 0 & 0 & 0\\ \frac{-(1+a)}{(a(a+1)+1)^2} & 0 & 0 & \frac{-a}{(a(a+1)+1)^2}\\ 0 & 0 & 1 & 0 \end{pmatrix}$$
$$= \begin{pmatrix} 0 & -a^3 & a - 1 & 0\\ 1 & 0 & 0 & 0\\ a - 1 & 0 & 0 & -a^3\\ 0 & 0 & 1 & 0 \end{pmatrix}.$$

Thus, we obtain the characteristic equation of the Jacobian matrix B as

$$(a^3 + \lambda^2)^2 - (a-1)^2 \lambda^2 = 0,$$

or

$$(\lambda^2 + (a-1)\lambda + a^3)(\lambda^2 - (a-1)\lambda + a^3) = 0.$$

Hence, it is clearly seen that numerically

$$|\lambda_1| = |\lambda_2| = |\lambda_3| = |\lambda_4| = 0.40089 < 1.$$

Consequently, the equilibrium point (a, a) is locally asymptotically stable. So, this completes the proof.

Theorem 8.3 The equilibrium point of system (8.1) is globally asymptotically stable.

Proof. Let $\{x_n, y_n\}_{n \ge -1}$ be a solution system (8.1). By Theorem 8.2, we need only to prove that the equilibrium point (a, a) is global attractor, that is

 $\lim_{n \to \infty} \left(x_n, y_n \right) = \left(a, a \right).$

From Theorem 8.1 and (2.13) and (2.14), it follows that

$$\lim_{n \to \infty} x_{2n-1} = \lim_{n \to \infty} \frac{T_{2n-2}x_{-1}y_0 + (T_{2n} - T_{2n-1})y_0 + T_{2n-1}}{T_{2n-1}x_{-1}y_0 + (T_{2n-2} + T_{2n-1})y_0 + T_{2n}}$$

$$= \lim_{n \to \infty} \frac{T_{2n-2}\left(x_{-1}y_0 + \left(\frac{T_{2n}}{T_{2n-2}} - \frac{T_{2n-1}}{T_{2n-2}}\right)y_0 + \frac{T_{2n-1}}{T_{2n-2}}\right)}{T_{2n-1}\left(x_{-1}y_0 + \left(\frac{T_{2n-2}}{T_{2n-1}} + 1\right)y_0 + \frac{T_{2n}}{T_{2n-1}}\right)}$$

$$= \left(\frac{x_{-1}y_0 + (\alpha^2 - \alpha)y_0 + \alpha}{x_{-1}y_0 + \left(\frac{1}{\alpha} + 1\right)y_0 + \alpha}\right)\lim_{n \to \infty} \frac{T_{2n-2}}{T_{2n-1}}$$

$$= \lim_{n \to \infty} \frac{T_{2n-2}}{T_{2n-1}}$$

$$= \frac{1}{\alpha}$$

$$= a.$$

and

$$\lim_{n \to \infty} x_{2n} = \lim_{n \to \infty} \frac{T_{2n-1}y_{-1}x_0 + (T_{2n+1} - T_{2n})x_0 + T_{2n}}{T_{2n}y_{-1}x_0 + (T_{2n-1} + T_{2n})x_0 + T_{2n+1}}$$

$$= \lim_{n \to \infty} \frac{T_{2n-1}\left(y_{-1}x_0 + \left(\frac{T_{2n+1}}{T_{2n-1}} - \frac{T_{2n}}{T_{2n-1}}\right)x_0 + \frac{T_{2n}}{T_{2n-1}}\right)}{T_{2n}\left(y_{-1}x_0 + \left(\frac{T_{2n-1}}{T_{2n}} + 1\right)x_0 + \frac{T_{2n+1}}{T_{2n}}\right)}$$

$$= \left(\frac{y_{-1}x_0 + (\alpha^2 - \alpha)x_0 + \alpha}{y_{-1}x_0 + \left(\frac{1}{\alpha} + 1\right)x_0 + \alpha}\right)\lim_{n \to \infty} \frac{T_{2n-1}}{T_{2n}}$$

$$= \lim_{n \to \infty} \frac{T_{2n-1}}{T_{2n}}$$

$$= \frac{1}{\alpha}$$

$$= a.$$

Then, we have

 $\lim_{n \to \infty} x_n = a.$

Similarly, we obtain

 $\lim_{n \to \infty} y_n = a.$

Therefore, we get

$$\lim_{n \to \infty} \left(x_n, y_n \right) = \left(a, a \right).$$

The proof is completed.

8.1.2 The System (8.2)

In this subsection, we introduce our main results related to the system (8.2). Our aim is to investigate the general solution in explicit form of system (8.2) and the asymptotic behavior of solutions of system (8.2).

Theorem 8.4 Let $\{x_n, y_n\}_{n=-1}^{\infty}$ be a solution system (8.2). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n, y_n\}_{n=-1}^{\infty}$ is given by

$$\begin{aligned} x_{2n-1} &= \frac{-\left(T_{2n-2}x_{-1}y_0 + \left(T_{2n-1} - T_{2n}\right)y_0 + T_{2n-1}\right)\right)}{T_{2n-1}x_{-1}y_0 - \left(T_{2n-2} + T_{2n-1}\right)y_0 + T_{2n}}, \\ x_{2n} &= \frac{-\left(T_{2n-1}y_{-1}x_0 + \left(T_{2n} - T_{2n+1}\right)x_0 + T_{2n}\right)\right)}{T_{2n}y_{-1}x_0 - \left(T_{2n-1} + T_{2n}\right)x_0 + T_{2n+1}}, \\ y_{2n-1} &= \frac{-\left(T_{2n-2}y_{-1}x_0 + \left(T_{2n-1} - T_{2n}\right)x_0 + T_{2n-1}\right)\right)}{T_{2n-1}y_{-1}x_0 - \left(T_{2n-2} + T_{2n-1}\right)x_0 + T_{2n}}, \\ y_{2n-1} &= \frac{-\left(T_{2n-1}x_{-1}y_0 + \left(T_{2n} - T_{2n+1}\right)y_0 + T_{2n}\right)\right)}{T_{2n}x_{-1}y_0 - \left(T_{2n-1} + T_{2n}\right)y_0 + T_{2n+1}}, \end{aligned}$$

where initial conditions x_{-1} , y_{-1} , x_0 , $y_0 \in \mathbb{R} - F_2$, with F_2 is the forbidden set of system (8.2) given by

$$F_2 = \bigcup_{n=-1}^{\infty} \{ (x_{-1}, y_{-1}, x_0, y_0) : A_n = 0 \text{ or } B_n = 0 \text{ or } C_n = 0 \text{ or } D_n = 0 \}$$

where

$$A_{n} = T_{2n-1}x_{-1}y_{0} - (T_{2n-2} + T_{2n-1})y_{0} + T_{2n},$$

$$B_{n} = T_{2n}y_{-1}x_{0} - (T_{2n-1} + T_{2n})x_{0} + T_{2n+1},$$

$$C_{n} = T_{2n-1}y_{-1}x_{0} - (T_{2n-2} + T_{2n-1})x_{0} + T_{2n},$$

$$D_{n} = T_{2n}x_{-1}y_{0} - (T_{2n-1} + T_{2n})y_{0} + T_{2n+1}.$$

Proof. Consider system (8.2) by taking n = 0, 1, 2, ... as follows:

$$\begin{array}{lll} n=0 & \Rightarrow & x_1=\frac{-1}{x_{-1}y_0-y_0+1}, & y_1=\frac{-1}{y_{-1}x_0-x_0+1}, \\ n=1 & \Rightarrow & x_2=\frac{-(y_{-1}x_0-x_0+1)}{y_{-1}x_0-2x_0+2}, & y_2=\frac{-(x_{-1}y_0-y_0+1)}{x_{-1}y_0-2y_0+2}, \\ n=2 & \Rightarrow & x_3=\frac{-(x_{-1}y_0-2y_0+2)}{2x_{-1}y_0-3y_0+4}, & y_3=\frac{-(y_{-1}x_0-2x_0+2)}{2y_{-1}x_0-3x_0+4}, \\ n=3 & \Rightarrow & x_4=\frac{-(2y_{-1}x_0-3x_0+4)}{4y_{-1}x_0-6x_0+7}, & y_4=\frac{-(2x_{-1}y_0-3y_0+4)}{4x_{-1}y_0-6y_0+7}, \\ n=4 & \Rightarrow & x_5=\frac{-(4x_{-1}y_0-y_0+7)}{7x_{-1}y_0-11y_0+13}, & y_5=\frac{-(4y_{-1}x_0-6x_0+7)}{7y_{-1}x_0-11x_0+13}, \\ n=5 & \Rightarrow & x_6=\frac{-(7y_{-1}x_0-11x_0+13)}{13y_{-1}x_0-20x_0+24}, & x_6=\frac{-(7x_{-1}y_0-11y_0+13)}{13x_{-1}y_0-20y_0+24}, \end{array}$$

If we keep on this process and also regard (2.11), then the result directly follows from a simple induction.

Theorem 8.5 The system (8.2) has unique negative equilibrium point $(\overline{x}, \overline{y}) = (d, d)$ and (d, d) is locally asymptotically stable.

Proof. Clearly, equilibrium point of system (8.2) is the real roots of the equations

$$\overline{x} = \frac{-1}{\overline{x}(\overline{y}-1)+1}, \quad \overline{y} = \frac{-1}{\overline{y}(\overline{x}-1)+1}.$$
(8.5)

In (8.5), after some operations, we get

$$\overline{x} = \overline{y}.$$

As a result, we obtain the following equation

$$\overline{x}^3 - \overline{x}^2 + \overline{x} + 1 = 0. \tag{8.6}$$

Then, the roots of the cubic equation (8.6) are given by

$$d = \frac{1 + \sqrt[3]{3\sqrt{33} - 17} - \sqrt[3]{3\sqrt{33} + 17}}{3},$$

$$e = \frac{1 + \omega\sqrt[3]{3\sqrt{33} - 17} - \omega^2\sqrt[3]{3\sqrt{33} + 17}}{3},$$

$$f = \frac{1 + \omega^2\sqrt[3]{3\sqrt{33} - 17} - \omega\sqrt[3]{3\sqrt{33} + 17}}{3},$$

where

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. So, the root d is only real number. Therefore, the unique negative equilibrium point of system (8.2) is $(\overline{x}, \overline{y}) = (d, d)$.

Now, we show that the unique negative equilibrium point of system (8.2) is locally asymptotically stable.

Let I and J are some intervals of real numbers and consider the functions

 $f: I^2 \times J^2 \to I$ and $g: I^2 \times J^2 \to J$

defined by

$$f(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{-1}{y_n(x_{n-1}-1)+1}, \ g(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{-1}{x_n(y_{n-1}-1)+1}$$

We consider the following transformation to build corresponding linearized form of system (8.2)

$$(x_n, x_{n-1}, y_n, y_{n-1}) \to (f, f_1, g, g_1),$$

where

$$f(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{-1}{y_n (x_{n-1} - 1) + 1},$$

$$f_1(x_n, x_{n-1}, y_n, y_{n-1}) = x_n,$$

$$g(x_n, x_{n-1}, y_n, y_{n-1}) = \frac{-1}{x_n (y_{n-1} - 1) + 1},$$

$$g_1(x_n, x_{n-1}, y_n, y_{n-1}) = y_n.$$

The linearized system of system (8.2) about the equilibrium point (d, d) under the above transformation is given as

$$X_{n+1} = BX_n,$$

where $X_n = (x_n, x_{n-1}, y_n, y_{n-1})^T$ and B is a Jacobian matrix of system (8.2) about the equilibrium point (d, d) and given by

$$B = \begin{pmatrix} 0 & \frac{d}{(d(d-1)+1)^2} & \frac{d-1}{(d(d-1)+1)^2} & 0 \\ 1 & 0 & 0 & 0 \\ \frac{d-1}{(d(d-1)+1)^2} & 0 & 0 & \frac{d}{(d(d-1)+1)^2} \\ 0 & 0 & 1 & 0 \end{pmatrix}$$
$$= \begin{pmatrix} 0 & d^3 - (1+d) & 0 \\ 1 & 0 & 0 & 0 \\ -(1+d) & 0 & 0 & d^3 \\ 0 & 0 & 1 & 0 \end{pmatrix}.$$

Thus, we obtain the characteristic equation of the Jacobian matrix B as

$$(d^3 - \lambda^2)^2 - (1+d)^2 \lambda^2 = 0,$$

or

$$(\lambda^2 - (1+d)\lambda - d^3)(\lambda^2 + (1+d)\lambda - d^3) = 0.$$

Hence, it is clearly seen that numerically

$$|\lambda_1| = |\lambda_2| = |\lambda_3| = |\lambda_4| = 0.40089 < 1.$$

Consequently, the equilibrium point (d, d) is locally asymptotically stable.

Theorem 8.6 The equilibrium point of system (8.2) is globally asymptotically stable.

Proof. Let $\{x_n, y_n\}_{n \ge -1}$ be a solution system (8.2). By Theorem 8.5, we need only to prove that the equilibrium point (d, d) is global attractor, that is

 $\lim_{n \to \infty} (x_n, y_n) = (d, d) \,.$

From Theorem 8.4 and (2.13) and (2.14), it follows that

$$\lim_{n \to \infty} x_{2n-1} = \lim_{n \to \infty} \frac{-(T_{2n-2}x_{-1}y_0 + (T_{2n-1} - T_{2n})y_0 + T_{2n-1})}{T_{2n-1}x_{-1}y_0 - (T_{2n-2} + T_{2n-1})y_0 + T_{2n}}$$

$$= \lim_{n \to \infty} \frac{-T_{2n-2}\left(x_{-1}y_0 + \left(\frac{T_{2n-1}}{T_{2n-2}} - \frac{T_{2n}}{T_{2n-2}}\right)y_0 + \frac{T_{2n-1}}{T_{2n-2}}\right)}{T_{2n-1}\left(x_{-1}y_0 - \left(\frac{T_{2n-2}}{T_{2n-1}} + 1\right)y_0 + \frac{T_{2n}}{T_{2n-1}}\right)}$$

$$= \left(\frac{x_{-1}y_0 + (\alpha - \alpha^2)y_0 + \alpha}{x_{-1}y_0 - \left(\frac{1}{\alpha} + 1\right)y_0 + \alpha}\right)\lim_{n \to \infty} \frac{-T_{2n-2}}{T_{2n-1}}$$

$$= \lim_{n \to \infty} \frac{-T_{2n-2}}{T_{2n-1}}$$

$$= -\frac{1}{\alpha}$$

$$= d,$$

and

$$\lim_{n \to \infty} x_{2n} = \lim_{n \to \infty} \frac{-(T_{2n-1}y_{-1}x_0 + (T_{2n} - T_{2n+1})x_0 + T_{2n})}{T_{2n}y_{-1}x_0 - (T_{2n-1} + T_{2n})x_0 + T_{2n+1}}$$

$$= \lim_{n \to \infty} \frac{-T_{2n-1}\left(y_{-1}x_0 + \left(\frac{T_{2n}}{T_{2n-1}} - \frac{T_{2n+1}}{T_{2n-1}}\right)x_0 + \frac{T_{2n}}{T_{2n-1}}\right)}{T_{2n}\left(y_{-1}x_0 - \left(\frac{T_{2n-1}}{T_{2n}} + 1\right)x_0 + \frac{T_{2n+1}}{T_{2n}}\right)}$$

$$= \left(\frac{y_{-1}x_0 + (\alpha - \alpha^2)x_0 + \alpha}{y_{-1}x_0 - \left(\frac{1}{\alpha} + 1\right)x_0 + \alpha}\right)\lim_{n \to \infty} \frac{-T_{2n-1}}{T_{2n}}$$

$$= \lim_{n \to \infty} \frac{-T_{2n-1}}{T_{2n}}$$

$$= -\frac{1}{\alpha}$$

$$= d.$$

Then, we have

 $\lim_{n \to \infty} x_n = d.$

Similarly, we obtain

 $\lim_{n \to \infty} y_n = d.$

Therefore, we get

 $\lim_{n \to \infty} \left(x_n, y_n \right) = \left(d, d \right),$

which completes the proof.



CHAPTER 9

THE DYNAMICS OF SOLUTIONS OF A RATIONAL DIFFERENCE EQUATION VIA GENERALIZED TRIBONACCI NUMBERS

Initially, we state that the results of this chapter are cited from [104] which has been published by us.

In this chapter, in the light of Chapter 6 and 7, we study the following difference equation

$$x_{n+1} = \frac{\gamma}{x_n \left(x_{n-1} + \alpha\right) + \beta}, \quad n = 0, 1, ...,$$
(9.1)

where the initial values x_{-1} and x_0 are arbitrary nonzero real and the parameters α , β and γ are nonnegative real numbers with $\gamma \neq 0$.

9.1 Introduction

First, from [121], consider the generalized Tribonacci sequence $\{V_n\}_{n=0}^{\infty}$ defined by the recurrent relation

$$V_{n+3} = rV_{n+2} + sV_{n+1} + tV_n, \ n \in \mathbb{N},$$
(9.2)

where the constant coefficients r, s, t are real numbers and the special initial conditions

$$V_0 = 0, V_1 = 1, V_2 = r.$$

The sequence $\{V_n\}_{n=0}^{\infty}$ can be extended to negative subscripts by defining

$$V_{-n} = -\frac{s}{t}V_{-(n-1)} - \frac{r}{t}V_{-(n-2)} + \frac{1}{t}V_{-(n-3)}$$

for n = 1, 2, 3, ... when $t \neq 0$. Hereby, recurrence (9.2) holds for all integer n.

As $\{V_n\}_{n=0}^{\infty}$ is a third order recurrence sequence (difference equation), it's characteristic equation is

$$x^3 - rx^2 - sx - t = 0, (9.3)$$
whose roots are

$$\varphi = \varphi(r, s, t) = \frac{r}{3} + A + B$$

$$\chi = \chi(r, s, t) = \frac{r}{3} + \omega A + \omega^2 B$$

$$\psi = \psi(r, s, t) = \frac{r}{3} + \omega^2 A + \omega B$$

where

$$A = \left(\frac{r^3}{27} + \frac{rs}{6} + \frac{t}{2} + \sqrt{\Delta}\right)^{1/3}, \ B = \left(\frac{r^3}{27} + \frac{rs}{6} + \frac{t}{2} - \sqrt{\Delta}\right)^{1/3}$$
$$\Delta = \Delta(r, s, t) = \frac{r^3 t}{27} - \frac{r^2 s^2}{108} + \frac{rst}{6} - \frac{s^3}{27} + \frac{t^2}{4},$$
$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp(2\pi i/3).$$

Notice that we get the following identities

$$\begin{aligned} \varphi + \chi + \psi &= r, \\ \varphi \chi + \varphi \psi + \chi \psi &= -s, \\ \varphi \chi \psi &= t. \end{aligned}$$

From now on, we assume that $\Delta(r, s, t) > 0$, so that the Eq.(9.2) has one real φ and two non-real solutions with the latter being conjugate complex. Therefore, in this case, it is widely known that generalized Tribonacci numbers can be stated, for all integers n, using Binet's formula

$$V_n = \frac{\varphi^{n+1}}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi^{n+1}}{(\chi - \varphi)(\chi - \psi)} + \frac{\psi^{n+1}}{(\psi - \varphi)(\psi - \chi)}.$$
(9.4)

We can present Binet's formula of the generalized Tribonacci numbers for the negative subscripts: for n = 1, 2, 3, ... we have

$$V_{-n} = \frac{\varphi^2 - r\varphi - s}{t} \frac{\varphi^{2-n}}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi^2 - r\chi - s}{t} \frac{\chi^{2-n}}{(\chi - \varphi)(\chi - \psi)} + \frac{\psi^2 - r\psi - s}{t} \frac{\psi^{2-n}}{(\psi - \varphi)(\psi - \chi)}.$$

Lemma 9.1 Let φ , χ and ψ be the roots of Eq.(9.3), suppose that φ is a real root with $\max\{|\varphi|; |\chi|; |\psi|\} = |\varphi|$. Then,

$$\lim_{n \to \infty} \frac{V_{n+1}}{V_n} = \varphi.$$
(9.5)

Proof. Note that there are three cases of the roots, that is when the roots are all real and distinct, all roots are equal or two roots are equal, complex conjugate. We will only proof the first case. The proof of the other two cases of the roots is similar to first one, so it will be omitted.

If φ , χ and ψ are real and distinct then, from Binet's formula

$$\lim_{n \to \infty} \frac{V_{n+1}}{V_n} = \lim_{n \to \infty} \frac{\frac{\varphi^{n+2}}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi^{n+2}}{(\chi - \varphi)(\chi - \psi)} + \frac{\psi^{n+2}}{(\psi - \varphi)(\psi - \chi)}}{\frac{\varphi^{n+1}}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi^{n+1}}{(\chi - \varphi)(\chi - \psi)} + \frac{\psi^{n+1}}{(\psi - \varphi)(\psi - \chi)}}$$

$$= \lim_{n \to \infty} \frac{\varphi^{n+1} \left(\frac{\varphi}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi}{(\chi - \varphi)(\chi - \psi)} \frac{\chi^{n+1}}{\varphi^{n+1}} + \frac{\psi}{(\psi - \varphi)(\psi - \chi)} \frac{\psi^{n+1}}{\varphi^{n}} \right)}{\varphi^n \left(\frac{\varphi}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi}{(\chi - \varphi)(\chi - \psi)} \frac{\chi^n}{\varphi^n} + \frac{\psi}{(\psi - \varphi)(\psi - \chi)} \frac{\psi^n}{\varphi^n} \right)}$$

$$= \lim_{n \to \infty} \frac{\varphi^{n+1}}{\varphi^n} \frac{\left(\frac{\varphi}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi}{(\chi - \varphi)(\chi - \psi)} \left(\frac{\chi}{\varphi} \right)^{n+1} + \frac{\psi}{(\psi - \varphi)(\psi - \chi)} \left(\frac{\psi}{\varphi} \right)^{n+1} \right)}{\left(\frac{\varphi}{(\varphi - \chi)(\varphi - \psi)} + \frac{\chi}{(\chi - \varphi)(\chi - \psi)} \left(\frac{\chi}{\varphi} \right)^n + \frac{\psi}{(\psi - \varphi)(\psi - \chi)} \left(\frac{\psi}{\varphi} \right)^n \right)}$$

$$= \varphi.$$

9.2 Main Results

In this section, we present our main results related to the difference equation (9.1). Our aim is to investigate the general solution in explicit form of Eq.(9.1) and the asymptotic behavior of solutions of Eq.(9.1).

Theorem 9.2 Let $\{x_n\}_{n=-1}^{\infty}$ be a solution of Eq.(9.1). Then, for n = 0, 1, 2, ..., the form of solutions $\{x_n\}_{n=-1}^{\infty}$ is given by

$$x_n = \frac{tV_{n-1}x_{-1}x_0 + (V_{n+1} - rV_n)x_0 + V_n}{tV_n x_{-1}x_0 + (V_{n+2} - rV_{n+1})x_0 + V_{n+1}},$$
(9.6)

where V_n is the nth generalized-Tribonacci number and the initial conditions $x_{-1}, x_0 \in \mathbb{R} - F$, with F is the forbidden set of Eq.(9.1) given by

$$F = \bigcup_{n=-1}^{\infty} \left\{ (x_{-1}, x_0) : tV_n x_{-1} x_0 + (V_{n+2} - rV_{n+1}) x_0 + V_{n+1} = 0 \right\}.$$

Proof. First, by using the change of variables

$$x_n = \frac{w_{n-1}}{w_n},\tag{9.7}$$

Eq.(9.1) is reduced to linear third order difference equation

$$w_{n+1} = \frac{\beta}{\gamma}w_n + \frac{\alpha}{\gamma}w_{n-1} + \frac{1}{\gamma}w_{n-2}.$$

 Set

$$r:=\frac{\beta}{\gamma},\,s:=\frac{\alpha}{\gamma},\ t:=\frac{1}{\gamma},$$

so we have

 $w_{n+1} = rw_n + sw_{n-1} + tw_{n-2}.$

Now, as done in [86], we describe initial values of three sequences which will be repetitively defined and used in the rest of the proof. Let

$$a_1 := r, \ b_1 := s, \ c_1 := t.$$

We use an recurrent (iterative) method. Thus, we get

$$w_{n} = a_{1}w_{n-1} + b_{1}w_{n-2} + c_{1}w_{n-3}$$

$$= a_{1}(rw_{n-2} + sw_{n-3} + tw_{n-4}) + b_{1}w_{n-2} + c_{1}w_{n-3}$$

$$= (ra_{1} + b_{1})w_{n-2} + (sa_{1} + c_{1})w_{n-3} + ta_{1}w_{n-4}$$

$$= a_{2}w_{n-2} + b_{2}w_{n-3} + c_{2}w_{n-4},$$
(9.8)

where

$$a_2 := ra_1 + b_1, \ b_2 := sa_1 + c_1, \ c_2 := ta_1.$$
 (9.9)

By continuing iteration, it implies that

$$w_{n} = a_{2}w_{n-2} + b_{2}w_{n-3} + c_{2}w_{n-4}$$

$$= a_{2}(rw_{n-3} + sw_{n-4} + tw_{n-5}) + b_{2}w_{n-3} + c_{2}w_{n-4}$$

$$= (ra_{2} + b_{2})w_{n-3} + (sa_{2} + c_{2})w_{n-4} + ta_{2}w_{n-5}$$

$$= a_{3}w_{n-3} + b_{3}w_{n-4} + c_{3}w_{n-5},$$
(9.10)

where

$$a_3 := ra_2 + b_2, \ b_3 := sa_2 + c_2, \ c_3 := ta_2.$$
 (9.11)

Based on relations (9.8)-(9.11), we suppose that for some $k \in \mathbb{N}$ such that $2 \leq k \leq n-1$, we have

$$w_n = a_k w_{n-k} + b_k w_{n-k-1} + c_k w_{n-k-2}, (9.12)$$

and

$$a_k := ra_{k-1} + b_{k-1}, \ b_k := sa_{k-1} + c_{k-1}, \ c_k := ta_{k-1}.$$

$$(9.13)$$

Next, by continuing iteration, it follows that

$$w_{n} = a_{k}w_{n-k} + b_{k}w_{n-k-1} + c_{k}w_{n-k-2}$$

$$= a_{k}(rw_{n-k-1} + sw_{n-k-2} + tw_{n-k-3}) + b_{k}w_{n-k-1} + c_{k}w_{n-k-2}$$

$$= (ra_{k} + b_{k})w_{n-k-1} + (sa_{k} + c_{k})w_{n-k-2} + ta_{k}w_{n-k-3}$$

$$= a_{k+1}w_{n-k-1} + b_{k+1}w_{n-k-2} + c_{k+1}w_{n-k-3},$$

where

$$a_{k+1} := ra_k + b_k, \ b_{k+1} := sa_k + c_k, \ c_{k+1} := ta_k.$$

Now, we maintain sequences a_k , b_k and c_k for some nonpositive values of index k. Notice that since $\gamma \neq 0$, the recurrent relations in (9.13) can be really used for computing values of sequences a_k , b_k and c_k for every $k \leq 0$.

Using the recurrent relations with the indices k = 1, k = 0 and k = -1, respectively, after some computations, it implies that

$$a_{0} = \frac{c_{1}}{c} = 1$$

$$b_{0} = a_{1} - aa_{0} = a - a.1 = 0$$

$$c_{0} = b_{1} - ba_{0} = b - b.1 = 0$$

$$a_{-1} = \frac{c_{0}}{c} = 0$$

$$b_{-1} = a_{0} - aa_{-1} = 1 - a.0 = 1$$

$$c_{-1} = b_{0} - ba_{-1} = 0 - b.0 = 0$$

$$a_{-2} = \frac{c_{-1}}{c} = 0$$

$$b_{-2} = a_{-1} - aa_{-2} = 0 - a.0 = 0$$

$$c_{-2} = b_{-1} - ba_{-2} = 1 - b.0 = 1.$$

Thus, we obtain

$$a_{0} = 1 \quad a_{-1} = 0 \quad a_{-2} = 0$$

$$b_{0} = 0 \quad b_{-1} = 1 \quad b_{-2} = 0$$

$$c_{0} = 0 \quad c_{-1} = 0 \quad c_{-2} = 1.$$

(9.14)

From (9.13), we get

 $a_n = ra_{n-1} + sa_{n-2} + ta_{n-3},\tag{9.15}$

 $b_n = a_{n+1} - ra_n, (9.16)$

$$c_n = ta_{n-1},\tag{9.17}$$

for $n \in \mathbb{N}$.

If we get k = n in (9.12), we have

$$w_n = a_n w_0 + b_n w_{-1} + c_n w_{-2},$$

for $n \in \mathbb{N}_0$.

From (9.15)-(9.17), we obtain

$$w_n = a_n w_0 + (a_{n+1} - ra_n) w_{-1} + ta_{n-1} w_{-2},$$

for $n \in \mathbb{N}_0$.

Using (9.18) in (9.7), we get

$$x_n = \frac{a_{n-1}w_0 + (a_n - ra_{n-1})w_{-1} + ta_{n-2}w_{-2}}{a_nw_0 + (a_{n+1} - ra_n)w_{-1} + ta_{n-1}w_{-2}}$$

it follows that

$$x_n = \frac{ta_{n-2}x_{-1}x_0 + (a_n - ra_{n-1})x_0 + a_{n-1}}{ta_{n-1}x_{-1}x_0 + (a_{n+1} - ra_n)x_0 + a_n}$$

or equivalently

$$x_{n} = \frac{ta_{n-2}x_{-1}x_{0} + (a_{n} - ra_{n-1})x_{0} + a_{n-1}}{ta_{n-1}x_{-1}x_{0} + (sa_{n-1} + ta_{n-2})x_{0} + a_{n}}$$

From initial values (9.14) and definitions of sequences a_n and V_n , we have

$$a_n = V_{n+1},$$

with the backward shifted initial values of the sequence a_n . Then, it follows

$$x_n = \frac{tV_{n-1}x_{-1}x_0 + (V_{n+1} - rV_n)x_0 + V_n}{tV_n x_{-1}x_0 + (V_{n+2} - rV_{n+1})x_0 + V_{n+1}}$$

or

$$x_n = \frac{tV_{n-1}x_{-1}x_0 + (V_{n+1} - rV_n)x_0 + V_n}{tV_n x_{-1}x_0 + (sV_n + tV_{n-1})x_0 + V_{n+1}}.$$

(9.18)

The proof is complete.

Now, we will analyze five special cases of the above theorem according to the states of r, s, t.

Case 1: r = s = t = 1

In this case, the (a_n) sequence has the following recurrence relation

$$a_n = a_{n-1} + a_{n-2} + a_{n-3},$$

such that a few terms of this sequence are

$$a_{-2} = 0, \ a_{-1} = 0, \ a_0 = 1, \ a_1 = 1, \ a_2 = 2, \ a_3 = 4.$$
 (9.19)

Then, from initial values (9.19) and definitions of sequences a_n and T_n which is Tribonacci numbers, we have

$$a_n = T_{n+1},$$

with the backward shifted initial values of the sequence a_n .

Hence, we obtain

$$x_n = \frac{T_{n-1}x_{-1}x_0 + (T_{n+1} - T_n)x_0 + T_n}{T_n x_{-1}x_0 + (T_n + T_{n-1})x_0 + T_{n+1}}.$$

Case 2:
$$r = 0, s = t = 1$$

In this case, the (a_n) sequence has the following recurrence relation

$$a_n = a_{n-2} + a_{n-3},$$

such that a few terms of this sequence are

$$a_{-2} = 0, \ a_{-1} = 0, \ a_0 = 1, \ a_1 = 0, \ a_2 = 1, \ a_3 = 1, \ a_4 = 1.$$
 (9.20)

Then, from initial values (9.20) and definitions of sequences a_n and P_n which is Padovan numbers, we get

$$a_{n+2} = P_n,$$

with the forward shifted initial values of the sequence a_n .

Therefore, we have

$$x_n = \frac{P_{n-4}x_{-1}x_0 + P_{n-2}x_0 + P_{n-3}}{P_{n-3}x_{-1}x_0 + P_{n-1}x_0 + P_{n-2}}.$$

Case 3: r = 0, s = t = 1

In this case, the (a_n) sequence has the following recurrence relation

$$a_n = a_{n-2} + a_{n-3},$$

such that a few terms of this sequence are

$$a_{-2} = 0, \ a_{-1} = 0, \ a_0 = 1, \ a_1 = 0, \ a_2 = 1, \ a_3 = 1, \ a_4 = 1.$$
 (9.21)

Then, from initial values (9.21) and definitions of sequences a_n and S_n which is Padovan-Perrin numbers, we have

$$a_n = S_{n+2},$$

with the backward shifted initial values of the sequence a_n .

Thus, we obtain

$$x_n = \frac{S_n x_{-1} x_0 + S_{n+2} x_0 + S_{n+1}}{S_{n+1} x_{-1} x_0 + S_{n+3} x_0 + S_{n+2}}$$

Case 4: r = 1, s = 0, t = 1

In this case, the (a_n) sequence has the following recurrence relation

$$a_n = a_{n-1} + a_{n-3}$$

such that a few terms of this sequence are

$$a_{-2} = 0, \ a_{-1} = 0, \ a_0 = 1, \ a_1 = 1, \ a_2 = 1, \ a_3 = 2, \ a_4 = 3.$$
 (9.22)

Then, from initial values (9.22) and definitions of sequences a_n and N_n which is Narayana numbers, we have

$$a_n = N_{n+1},$$

with the backward shifted initial values of the sequence a_n .

Fromhere, we have

$$x_n = \frac{N_{n-1}x_{-1}x_0 + N_{n-2}x_0 + N_n}{N_n x_{-1}x_0 + N_{n-1}x_0 + N_{n+1}}$$

Case 5: r = s = 1, t = 2

In this case, the (a_n) sequence has the following recurrence relation

$$a_n = a_{n-1} + a_{n-2} + 2a_{n-3},$$

such that a few terms of this sequence are

$$a_{-2} = 0, \ a_{-1} = 0, \ a_0 = 1, \ a_1 = 1, \ a_2 = 2, \ a_3 = 5, \ a_4 = 9.$$
 (9.23)

Next, from initial values (9.23) and definitions of sequences a_n and J_n which is third order Jacobsthal numbers, we have

 $a_n = J_{n+1},$

with the backward shifted initial values of the sequence a_n .

Herefrom, we get

$$x_n = \frac{2J_{n-1}x_{-1}x_0 + (J_{n+1} - J_n)x_0 + J_n}{2J_n x_{-1}x_0 + (J_{n+2} - J_{n+1})x_0 + J_{n+1}}$$

Theorem 9.3 Eq.(9.1) has unique equilibrium point $\overline{x} = \mu$ and μ is locally asymptotically stable.

Proof. Equilibrium point of Eq.(9.1) is the real roots of the equation

$$\overline{x} = \frac{\gamma}{\overline{x}\left(\overline{x} + \alpha\right) + \beta}.$$

After simplification, we get the following cubic equation

$$\overline{x}^3 + \alpha \overline{x}^2 + \beta \overline{x} - \gamma = 0. \tag{9.24}$$

Then, the roots of the cubic equation (9.24) are given by

$$\mu = \mu(\alpha, \beta, \gamma) = -\frac{\alpha}{3} + C + D,$$

$$\sigma = \sigma(\alpha, \beta, \gamma) = -\frac{\alpha}{3} + \omega C + \omega^2 D,$$

$$\phi = \phi(\alpha, \beta, \gamma) = -\frac{\alpha}{3} + \omega^2 C + \omega D,$$

where

$$C = \left(\frac{-\alpha^3}{27} + \frac{\alpha\beta}{6} + \frac{\gamma}{2} + \sqrt{\Delta}\right)^{1/3}, D = \left(\frac{-\alpha^3}{27} + \frac{\alpha\beta}{6} + \frac{\gamma}{2} - \sqrt{\Delta}\right)^{1/3}$$
$$\Delta = \Delta(r, s, t) = -\frac{\alpha^3\gamma}{27} - \frac{\alpha^2\beta^2}{108} + \frac{\alpha\beta\gamma}{6} + \frac{\beta^3}{27} + \frac{\gamma^2}{4}$$

and

$$\omega = \frac{-1 + i\sqrt{3}}{2} = \exp\left(2\pi i/3\right)$$

is a primitive cube root of unity. So, the root μ is only real number. So, the unique equilibrium point of Eq.(9.1) is $\overline{x} = \mu$.

Now, we demonstrate that the equilibrium point of Eq.(9.1) is locally asymptotically stable.

Let I be an interval of real numbers and consider the function

 $f:I^2\to I$

defined by

$$f(x,y) = \frac{\gamma}{x(y+\alpha)+\beta}.$$

The linearized equation of Eq.(9.1) about the equilibrium point $\overline{x} = \mu$ is

$$z_{n+1} = pz_n + qz_{n-1},$$

where

$$p = \frac{\partial f(\overline{x}, \overline{x})}{\partial x} = \frac{-\gamma (\mu + \alpha)}{(\mu (\mu + \alpha) + \beta)^2}$$
$$= \frac{-\gamma (\mu + \alpha)}{(\mu^2 + \alpha \mu + \beta)^2}$$
$$= \frac{-\gamma (\mu + \alpha)}{\left(\frac{\gamma}{\mu}\right)^2}$$
$$= \frac{-(\mu^3 + \alpha \mu^2)}{\gamma}$$
$$= \frac{\beta \mu - \gamma}{\gamma},$$
$$q = \frac{\partial f(\overline{x}, \overline{x})}{\partial y} = \frac{-\gamma \mu}{(\mu (\mu + \alpha) + \beta)^2}$$
$$= \frac{-\gamma \mu}{(\mu^2 + \alpha \mu + \beta)^2}$$
$$= \frac{-\gamma \mu}{\left(\frac{\gamma}{\mu}\right)^2}$$
$$= -\frac{\mu^3}{\gamma},$$

and the corresponding characteristic equation is

$$\lambda^2 - \left(\frac{\beta\mu - \gamma}{\gamma}\right)\lambda + \frac{\mu^3}{\gamma} = 0.$$

Consider two functions defined by

$$a(\lambda) = \lambda^2, \ b(\lambda) = \left(\frac{\beta\mu - \gamma}{\gamma}\right)\lambda - \frac{\mu^3}{\gamma}.$$

We have

$$\left|\frac{\beta\mu-\gamma}{\gamma}-\frac{\mu^3}{\gamma}\right|<1.$$

Then, it follows that

 $|b(\lambda)| < |a(\lambda)|$, for all $\lambda : |\lambda| = 1$.

Therefore, by Rouche's Theorem, all zeros of $P(\lambda) = a(\lambda) - b(\lambda) = 0$ lie in $|\lambda| < 1$. Hereby, by Theorem 2.4, we have that the unique equilibrium point of Eq.(9.1) $\overline{x} = \mu$ is locally asymptotically stable.

Theorem 9.4 Assume that $\mu \varphi = 1$. Then, the equilibrium point of Eq.(9.1) is globally asymptotically stable.

Proof. Let $\{x_n\}_{n\geq -1}$ be a solution of Eq.(9.1). By Theorem 9.3, we need only to prove that the equilibrium point μ is global attractor, that is

$$\lim_{n \to \infty} x_n = \mu.$$

From Theorem 9.2 and (9.3) and (9.5), it follows that

$$\lim_{n \to \infty} x_n = \lim_{n \to \infty} \frac{tV_{n-1}x_{-1}x_0 + (V_{n+1} - rV_n)x_0 + V_n}{tV_n x_{-1}x_0 + (V_{n+2} - rV_{n+1})x_0 + V_{n+1}}$$

$$= \lim_{n \to \infty} \frac{tV_{n-1}\left(x_{-1}x_0 + \left(\frac{1}{t}\frac{V_{n+1}}{V_{n-1}} - \frac{r}{t}\frac{V_n}{V_{n-1}}\right)x_0 + \frac{V_n}{V_{n-1}}\right)}{tV_n\left(x_{-1}x_0 + \left(\frac{1}{t}\frac{V_{n+2}}{V_n} - \frac{r}{t}\frac{V_{n+1}}{V_n}\right)x_0 + \frac{V_{n+1}}{V_n}\right)}$$

$$= \left(\frac{x_{-1}x_0 + \left(\frac{1}{t}\varphi^2 - \frac{r}{t}\varphi\right)x_0 + \varphi}{x_{-1}x_0 + \left(\frac{1}{t}\varphi^2 - \frac{r}{t}\varphi\right)x_0 + \varphi}\right)\lim_{n \to \infty} \frac{V_{n-1}}{V_n}$$

$$= \lim_{n \to \infty} \frac{V_{n-1}}{V_n}$$

$$= \frac{1}{\varphi}$$

This completes the proof.

Note that when $\alpha = \beta = \gamma = 1$, our assumption in Theorem 9.4 is immediately seen. Indeed,

$$\begin{split} \mu\varphi &= \frac{1}{3\gamma} \left(\frac{1}{3\sqrt[3]{2}} \sqrt[3]{27\gamma - 2\alpha^3 + 9\alpha\beta - 3\sqrt{3}S} - \frac{1}{3}\alpha + \frac{1}{3\sqrt[3]{2}} \sqrt[3]{27\gamma - 2\alpha^3 + 9\alpha\beta + 3\sqrt{3}S} \right) \\ \left(\beta + \frac{1}{\sqrt[3]{2}} \sqrt[3]{2\beta^3 + 27\gamma^2 - 3\sqrt{3}\gamma S + 9\alpha\beta\gamma} + \frac{1}{\sqrt[3]{2}} \sqrt[3]{2\beta^3 + 27\gamma^2 + 3\sqrt{3}\gamma S + 9\alpha\beta\gamma} \right) \\ S &= \sqrt{-4\alpha^3\gamma - \alpha^2\beta^2 + 18\alpha\beta\gamma + 4\beta^3 + 27\gamma^2}. \end{split}$$

Then, in the case $\alpha = \beta = \gamma = 1$, it follows that $\mu \varphi = 1$.



REFERENCES

- Camouzis E and Ladas G (2007) Dynamics of Third-Order Rational Difference Equations with Open Problems and Conjectures. Vol. 5 Chapman & Hall/CRC, Boca Raton, Fla, USA.
- [2] Kocić V and Ladas G (1993) Global Behavior of Nonlinear Difference Equations of Higher Order with Applications. Kluwer Academic Publishers, Dordrecht.
- [3] Kulenović M R S and Ladas G (2001) Dynamics of Second Order Rational Difference Equations. Chapman & Hall/CRC, Boca Raton, London.
- [4] Sedaghat H (2003) Nonlinear Difference Equations Theory with Applications to Social Science Models. Vol. 15. Springer Science & Business Media.
- [5] Agarwal R (1992) Difference Equations and Inequalities, Theory, Methods and Applications. Marcel Dekker Inc., New York.
- [6] Alligood K T, Sauer T and Yorke J A (1997) Chaos An Introduction to Dynamical Systems. Springer-Verlag, New York, Berlin, Heidelberg, Tokyo.
- [7] Devaney R (1992) A First Course in Chaotic Dynamical Systems: Theory and Experiment. Addison-Wesley, Reading, MA.
- [8] Elaydi S (2005) An Introduction to Difference Equations. 3rd ed., Springer-Verlag, New York.
- [9] Elaydi S (2000) Discrete Chaos. CRC Press, Boca Raton, FL.
- [10] Kelley W G and Peterson A C (1991) Difference Equations. Academic Press, New York.
- [11] Martelli M (1999) Introduction to Discrete Dynamical Systems and Chaos. John Wiley & Sons, New York.

- [12] Robinson C (1995) Dynamical Systems, Stability, Symbolic Dynamics and Chaos.
 CRC Press, Boca Raton, FL.
- [13] Grove E A and Ladas G (2004) Periodicities in Nonlinear Difference Equations.
 Chapman & Hall/CRC Press, Boca Raton, London.
- [14] Kulenović M R S and Merino O (2002) Discrete Dynamical Systems and Difference Equations with Mathematica. Chapman & Hall/CRC, Boca Raton, London.
- [15] Papaschinopoulos G and Schinas C J (1998) On a System of Two Difference Equations. J. Mathematical Analysis and Applications, 219 (2): 415-426.
- [16] Papaschinopoulos G and Schinas C J (2000) On the System of Two Nonlinear Difference Equations $x_{n+1} = A + \frac{x_{n-1}}{y_n}, y_{n+1} = A + \frac{y_{n-1}}{x_n}$. International Journal Mathematics and Mathematical Sciences, 23 (12): 839-848.
- [17] **Camouzis E and Papaschinopoulos G** (2004) Global Asymptotic Behavior of Positive Solutions on the System of Rational Difference Equations $x_{n+1} = 1 + \frac{x_n}{y_{n-m}}$, $y_{n+1} = 1 + \frac{y_n}{x_{n-m}}$. Applied Mathematics Letters, 17 (6): 733-737.
- [18] **Yang X** (2005) On the System of Rational Difference Equations $x_{n+1} = A + \frac{y_{n-1}}{x_{n-p}y_{n-q}}$, $y_{n+1} = A + \frac{x_{n-1}}{x_{n-r}y_{n-s}}$. J. Math. Anal. Appl., 307 (1): 305-311.
- [19] Zhang Y, Yang X, Megson G M and Evans D J (2006) On the System of Rational Difference Equations $x_{n+1} = A + \frac{1}{y_{n-p}}, y_{n+1} = A + \frac{1}{x_{n-r}y_{n-s}}$. Applied Mathematics and Computation, 176 (2): 403-408.
- [20] Zhang Y, Yang X, Evans D J and Zhu C (2007) On the Nonlinear Difference Equation System $x_{n+1} = A + \frac{y_{n-m}}{x_n}$, $y_{n+1} = A + \frac{x_{n-m}}{y_n}$. Computers and Mathematics with Applications, 53 (10): 1561-1566.
- [21] Zhang D, Ji W, Wang L and Li X (2013) On the Symmetrical System of Rational Difference Equation x_{n+1} = A + ^{y_{n-k}}/_{y_n}, y_{n+1} = A + ^{x_{n-k}}/_{x_n}. Applied Mathematics, 4 (05): 834-837.

- [22] Kurbanlı A S, Çınar C and Yalçınkaya İ (2011) On the Behavior of Positive Solutions of the System of Rational Difference Equations $x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1}+1}$, $y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1}+1}$. Mathematical and Computer Modelling, 53 (5-6): 1261-1267.
- [23] Kurbanlı A S, Çınar C and Şimsek D (2011) On the Periodicity of Solutions of the System of Rational Difference Equations $x_{n+1} = \frac{x_{n-1}+y_n}{y_n x_{n-1}-1}, \quad y_{n+1} = \frac{y_{n-1}+x_n}{x_n y_{n-1}-1}.$ Applied Mathematics, 2: 410-413.
- [24] Wang Q, Zhang G and Fu L (2013) On the Behavior of the Positive Solutions of the System of the Two Higher-Order Rational Difference Equations. Applied Mathematics, 4 (08): 1220-1225.
- [25] Zhang Q, Zhang W, Shao Y and Liu J (2014) On the System of High Order Rational Difference Equations. *International Scholarly Research Notices*, Article ID 760502, 5 p.
- [26] Zhang Q and Zhang W (2014) On a System of Two High Order Nonlinear Difference Equations. Advances in Mathematical Physics, Article ID 729273, 8 p.
- [27] Zhang Q, Liu J and Luo Z (2015) Dynamical Behavior of a System of Third-Order Rational Difference Equation. Discrete Dynamic in Nature and Society, Article ID 530453, 6 p.
- [28] Stevic S, Alghamdi M A, Alotaibi A and Shahzad N (2013) On a Nonlinear Second Order System of Difference Equations. Applied Mathematics and Computation, 219 (24): 11388-11394.
- [29] Bao H (2015) Dynamical Behavior of a System of Second-Order Nonlinear Difference Equations. International Journal of Differential Equations, Article ID 679017, 7 p.
- [30] Gümüş M and Soykan Y (2016) Global Character of a Six-Dimensional Nonlinear System of Difference Equations. Discrete Dynamics in Nature and Society, Article ID 6842521, 7 p.

- [31] Din Q (2016) Asymptotic Behavior of an Anti-Competitive System of Second-Order Difference Equations. Journal of the Egyptian Mathematical Society, 24: 37-43.
- [32] Mansour M, El-Dessoky M M and Elsayed E M (2012) The Form of the Solutions and Periodicity of Some Systems of Difference Equations. *Discrete Dynamics* in Nature and Society, Article ID 406821, 17 p.
- [33] Elsayed E M and El-Metwally H (2013) On the Solutions of Some Nonlinear Systems of Difference Equations. Advances in Difference Equations, 2013 (1): 161.
- [34] Elsayed E M (2014) Solution for Systems of Difference Equations of Rational Form of Order Two. Computational and Applied Mathematics, 33 (3): 751-765.
- [35] Clark D and Kulenović M R S (2002) A Coupled System of Rational Difference Equations. Computers and Mathematics with Applications, 43 (6): 849-867.
- [36] Clark D, Kulenović M R S and Selgrade J F (2003) Global Asymptotic Behavior of a Two-Dimensional Difference Equation Modelling Competition. Nonlinear Analysis, 52 (7): 1765-1776.
- [37] Kulenović M R S and Nurkanovic M (2003) Global Asymptotic Behavior of a Two-dimensional System of Difference Equations Modeling Cooperation. *Journal* of Difference Equations and Applications, 9 (1): 149-159.
- [38] Grove E A, Hadley D, Lapierre E and Schultz S W (2016) On the Global Behavior of the Rational System $x_{n+1} = \frac{\alpha_1}{x_n+y_n}$, $y_{n+1} = \frac{\alpha_2+\beta_2x_n+y_n}{y_n}$. Sarajova J. Mathematics, 8 (21): 283-292.
- [39] Dekkar I, Touafek N and Yazlik Y (2017) Global Stability of a Third-Order Nonlinear System of Difference Equations with Period-Two Coefficients. *RACSAM*, 111: 325-347.
- [40] Elsayed E M, Alotaibi A and Almaylabi H A (2017) On a Solutions of Fourth Order Rational Systems of Difference Equations. J. Comp. Anal. Appl., 22 (7): 1298-1308.

- [41] Haddad N, Touafek N and Rabago J F T (2018) Well-Defined Solutions of a Systems of Difference Equations. J. Appl. Math. Comput., 56 (1-2): 439-458.
- [42] Zhang Q, Yang L and Liu J (2013) On the Recursive System $x_{n+1} = A + \frac{x_{n-m}}{y_n}$, $y_{n+1} = B + \frac{y_{n-m}}{x_n}$. Acta Math. Univ. Comenianae, 2: 201-208.
- [43] Papaschinopoulos G, Radin M A and Schinas C J (2011) On the System of Two Difference Equations of Exponential Form: $x_{n+1} = a + bx_{n-1}e^{-y_n}$, $y_{n+1} = c + dy_{n-1}e^{-x_n}$: $x_{n+1} = a + bx_{n-1}e^{-y_n}$, $y_{n+1} = c + dy_{n-1}e^{-x_n}$. Mathematical and Computer Modelling, 54 (11): 2969-2977.
- [44] Papaschinopoulos G and Schinas C J (2012) On the Dynamics of Two Exponential Type Systems of Difference Equations. Computers and Mathematics with Application, 64 (7): 2326-2334.
- [45] Papaschinopoulos G, Radin M A and Schinas C J (2012) Study of the Asymptotic Behavior of the Solutions of Three Systems of Difference Equations of Exponential Form. Applied Mathematics and Computation, 218 (9): 5310-5318.
- [46] Elettreby M F and El-Metwally H (2013) On a System of Difference Equations of an Economic Model. Discrete Dynamics in Nature and Society, Article ID 405628, 6 p.
- [47] Papaschinopoulos G, Fotiades N and Schinas C J (2014) On a System of Difference Equations Including Negative Exponential Terms. Journal of Difference Equations and Applications, 20 (5-6): 717-732.
- [48] Khan A.Q (2014) Global Dynamics of Two Systems of Exponential Difference Equations by Lyapunov Function. Advances in Difference Equations, 2014 (1): 297.
- [49] **Şimsek D, Demir B and Çınar C** (2009) On the Solutions of the System of Difference Equation $x_{n+1} = \max\left\{\frac{A}{x_n}, \frac{y_n}{x_n}\right\}, y_{n+1} = \max\left\{\frac{A}{y_n}, \frac{x_n}{y_n}\right\}$. Discrete Dynamics in Nature and Society, Article ID 325296, 11 p.
- [50] Stevic S (2012) Solutions of a Max-Type System of Difference Equations. Applied Mathematics and Computation, 218 (19): 9825-9830.

- [51] Fotiades N and Papaschinopoulos G (2013) On a System of Difference Equations with Maximum. Applid Mathematics and Computation, 221: 684-690.
- [52] Stevic S (2013) On a Symmetric System of Max-Type Difference Equations. Applied Mathematics and Computation, 219 (15): 8407-8412.
- [53] Stevic S, Alghamdi M A, Alotaibi A and Shahzad N (2014) Boundedness Character of a Max-Type System of Difference Equations Second Order. *Electronic J. Qualitative Theory Differential Equ.*, 2015, (45): 1-12.
- [54] Kulenović M R S and Nurkanovic M (2005) Global Behavior of a Three-Dimensional Linear Fractional System of Difference Equations. J. Mathematical Analysis and Applications, 310 (2): 673-689.
- [55] **Kurbanlı A S** (2011) On the Behavior of Solutions of the System of Rational Difference Equations: $x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} - 1}, \quad y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} - 1}, \quad z_{n+1} = \frac{z_{n-1}}{y_n z_{n-1} - 1}.$ Discrete Dynamics in Nature and Society, Article ID 932362, 12 p.
- [56] **Kurbanlı A S** (2011) On the Behavior of Solutions of the System of Rational Difference Equations $x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} - 1}$, $y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} - 1}$, $z_{n+1} = \frac{1}{y_n z_n}$. Advances in Difference Equations, 2011 (1): 40.
- [57] Kurbanlı A S, Çınar C and Erdoğan M E (2011) On the Behavior of Solutions of the System of Rational Difference Equations: $x_{n+1} = \frac{x_{n-1}}{y_n x_{n-1} - 1}, \quad y_{n+1} = \frac{y_{n-1}}{x_n y_{n-1} - 1},$ $z_{n+1} = \frac{x_n}{y_n z_{n-1}}.$ Applied Mathematics, 2 (08): 1031-1038.
- [58] Özkan O and Kurbanlı A S (2013) On a System of Difference Equations. Discrete Dynamics in Nature and Society, Article ID 970316, 7 p.
- [59] Stevic S (2012) On a Third-Order System of Difference Equations. Applied Mathematics and Computation, 218 (14): 7649-7654.
- [60] Stevic S (2012) On Some Periodic Systems of Max-Type Difference Equations. Applied Mathematics and Computation, 218 (23): 11483-11487.

- [61] Stevic T and Iricanin B (2015) Long-Term Behavior of a Cyclic Max-Type System of Difference Equations. *Electronic Journal of Differential Equations*, 234: 1-12.
- [62] Göcen M and Cebeci A (2018) On the Periodic Solutions of Some Systems of Higher Order Difference Equations. Rocky Mountain J. Math., 48 (3): 845-858.
- [63] Göcen M and Güneysu M (2018) The Global Attractivity of Some Rational Difference Equations. J. Comput. Anal. Appl., 25 (7): 1233-1243.
- [64] Okumuş İ and Soykan Y (2017) On the Stability of a Nonlinear Difference Equations. Asian Journal of Mathematics and Computer Research, 17 (2): 88-110.
- [65] Okumuş İ and Soykan Y (2018) Dynamical Behavior of a System of Three-Dimensional Nonlinear Difference Equations. Advance in Difference Equations, 2018 (1): 223.
- [66] Okumuş İ and Soykan Y (2018) On the Dynamics of a Higher Order Nonlinear System of Difference Equations. arXiv: 1810.07986v1, 2018.
- [67] Okumuş İ and Soykan Y (2018) Some Technique to Show the Boundedness of Rational Difference Equations. Journal of Progressive Research in Mathematics, 13 (2): 2246-2258.
- [68] Okumuş I and Soykan Y (2018) On the Nature of Solutions of a System of Second Order Nonlinear Difference Equations. Journal of Progressive Research in Mathematics, 14 (2): 2399-2407.
- [69] Taşdemir E and Soykan Y (2016) On the Periodicies of the Difference Equation $x_{n+1} = x_n x_{n-1} + \alpha$. Karaelmas Sci. Eng. J., 6 (2): 329-333.
- [70] Taşdemir E and Soykan Y (2017) Long-Term Behavior of Solutions of the Non-Linear Difference Equation $x_{n+1} = x_{n-1}x_{n-3} - 1$. Gen. Math. Notes, 38 (1): 13-31.
- [71] Taşdemir E and Soykan Y (2018) Stability of Negative Equilibrium of a Non-Linear Difference Equation. J. Math. Sci. Adv. Appl., 49 (1): 51-57.

- [72] Taşdemir E and Soykan Y (2019) Dynamical Analysis of a Non-Linear Difference Equation. J. Comput. Anal. Appl., 26 (2): 288-301.
- [73] Halim Y (2016) Form and Periodicity of Solutions of Some Systems of Difference Equations. Mathematical Sciences Letters, 5 (1): 79-84.
- [74] Tollu D T, Yazlık Y and Taşkara N (2013) On the solutions of two special types of Riccati difference equation via Fibonacci numbers. Advances in Difference Equations, 2013:174.
- [75] Rabago J F T (2016) On the Closed-Form Solution of a Nonlinear Difference Equation and Another Proof to Sroysang's Conjecture. arXiv:1604.06659v1 [math.NT].
- [76] Yazlık Y, Tollu D T and Taşkara N (2013) On the Solutions of Difference Equation Systems with Padovan Numbers. Applied Mathematics, 4: 15-20.
- [77] Tollu D T, Yazlık Y and Taşkara N (2014) The Solutions of Four Riccati Difference Equations Associated with Fibonacci numbers. *Balkan Journal of Mathematics*, 2: 163-172.
- [78] Tollu D T, Yazlık Y and Taşkara N (2014) On Fourteen Solvable Systems of Difference Equations. Applied Mathematics and Computation, 233: 310-319.
- [79] Halim Y (2015) Global Character of Systems of Rational Difference Equations. Electronic Journal of Mathematical Analysis and Applications, 3 (1): 204-214.
- [80] Bacani J B and Rabago J F T (2015) On Two Nonlinear Difference Equations. Dynamics of Continuous, Discrete and Impulsive Systems, (Serias A) to appear.
- [81] Halim Y and Bayram M (2016) On the Solutions of a Higher-Order Difference Equation in terms of Generalized Fibonacci Sequences. *Mathematical Methods in* the Applied Sciences, 39: 2974-2982.
- [82] Halim Y (2016) A System of Difference Equations with Solutions Associated to Fibonacci Numbers. International Journal of Difference Equations, 11 (1): 65-77.

- [83] **El-Dessoky M M** (2017) On the Dynamics of Higher Order Difference Equations $x_{n+1} = ax_n + \frac{\alpha x_n x_{n-l}}{\beta x_n + \gamma x_{n-k}}$. J. Computational Analysis and Applications, 22 (7): 1309-1322.
- [84] Halim Y and Rabago J F T (2017) On Some Solvable Systems of Difference Equations with Solutions Associated to Fibonacci Numbers. *Electronic Journal of Mathematical Analysis and Applications*, 5 (1): 166-178.
- [85] Halim Y and Rabago J F T (2018) On the Solutions of a Second-Order Difference Equation in terms of Generalized Padovan Sequences. *Mathematica Slovaca*, 68 (3): 625-638.
- [86] Stevic S, Iricanin B, Kosmala W and Smarda Z (2018) Representation of Solutions of a Solvable Nonlinear Difference Equation of Second Order. *Electronic Journal of Qualitative Theory of Differential Equations*, 95: 1-18.
- [87] Alotaibi A M, Noorani M S M and El-Moneam M A (2018) On the Solutions of a System of Third-Order Rational Difference Equations. Discrete Dynamics in Nature and Society, Article ID 1743540, 11 pages.
- [88] El-Dessoky M M, Elabbasy E M and Asiri A (2018) Dynamics and Solutions of a Fifth-Order Nonlinear Difference Equations. *Discrete Dynamics in Nature and Society*, Article ID 9129354, 21 pages.
- [89] Matsunaga H and Suzuki R (2018) Classification of Global Behavior of a System of Rational Difference Equations. Applied Mathematics Letters, 85: 57-63.
- [90] Öcalan Ö and Duman O (2019) On Solutions of the Recursive Equations $x_{n+1} = x_{n-1}^p/x_n^p \ (p > 0)$ via Fibonacci-Type Sequences. *Electronic Journal of Mathematical Analysis and Applications*, 7 (1): 102-115.
- [91] Akrour Y, Touafek N and Halim Y (2019) On a System of Difference Equations of Second Order Solved in a Closed Form. arXiv:1904.04476v1, [math.DS].

- [92] Asiri A and Elsayed E M (2019) Dynamics and Solutions of Some Recursive Sequences of Higher Order. J. Computational Analysis and Applications, 27 (4): 656-670.
- [93] Elsayed E M (2015) Periodicity and Solutions for Some Systems of Nonlinear Rational Difference Equations. *Hacettepe Journal of Mathematics and Statistics*, 44 (6): 1361-1390.
- [94] Halim Y, Touafek N and Yazlık Y (2015) Dynamics Behavior of a Second-Order Nonlinear Rational Difference Equation. *Turkish Journal of Mathematics*, 39: 1004-1018.
- [95] Rabago J F T (2015) Effective Methods on Determining the Periodicity and Form of Solutions of Some Systems of Nonlinear Difference Equations. arXiv:1512.01605v1 [math.DS].
- [96] Stevic S (2004) More on a Rational Recurrence Relation. Applied Mathematics E-Notes, 4: 80-84.
- [97] Stevic S (2012) On Some Solvable Systems of Difference Equations. Applied Mathematics and Computation, 218: 5010-5018.
- [98] Stevic S (2014) Representation of Solutions of Bilinear Difference Equations in terms of Generalized Fibonacci Sequences. *Electronic Journal of Qualitative Theory* of Differential Equations, 67: 1-15.
- [99] Tollu D T, Yazlık, Y and Taşkara N (2018) On a Solvable Nonlinear Difference Equation of Higher Order. Turkish Journal of Mathematics, 42: 1765-1778.
- [100] Yılmazyıldırım B and Tollu D T (2018) Explicit Solutions of a Three-Dimensional System of Nonlinear Difference Equations. *Hittite Journal of Science* and Engineering, 5 (2): 119-123.
- [101] Okumuş İ and Soykan Y (2019) On the Solutions of Four Rational Difference Equations Associated to Tribonacci Numbers. *Preprints*, preprints201906.0266.v1.

- [102] Okumuş İ and Soykan Y (2019) On the Solutions of Four Second-Order Nonlinear Difference Equations. *Preprints*, preprints201906.0265.v1.
- [103] Okumuş İ and Soykan Y (2019) On the Solutions of Systems of Difference Equations via Tribonacci Numbers. arXiv preprint, arXiv:1906.09987v1 [math.DS].
- [104] Okumuş İ and Soykan Y (2019) On the Dynamics of Solutions of a Rational Difference Equation via Generalized Tribonacci Numbers. arXiv preprint, arXiv:1906.11629v1 [math.DS].
- [105] Zill D G and Shanahan P D (2009). A First Course in Complex Analysis with Applications. Jones & Bartlett Learning.
- [106] Psarros N, Papaschinopoulos G and Schinas C J (2017) Study of the Stability of a 3 × 3 System of Difference Equations Using Centre Manifold Theory. Applied Mathematics Letters, 64: 185-192.
- [107] Marsden J and McCracken M (1976) The Hopf Bifurcation and Its Application. Springer, New York.
- [108] Kuznetsov Y A (1998) Elements of Applied Bifurcation Theory. Second Edition, Applied Mathematical Sciences, vol. 112, Springer, New York.
- [109] Karydas N and Schinas J (1992) The Center Manifold Theorem for a Discrete System. Applicable Analysis, 44: 267-284.
- [110] Luis R, Elaydi S and Oliveira H (2011) Stability of a Ricker-Type Competition Model and the Competitive Exclusion Principle. Journal of Biological Dynamics, 5 (6): 636-660.
- [111] Kelley A (2017) The Stable, Center-Stable, Center-Unstable and Unstable Manifolds. Journal of Differential Equations, 3: 546-570.
- [112] Bektesević J, Mehuljić M and Hadziabdić V (2017) Global Asymptotic Behavior of Some Quadratic Rational Second-Order Difference Equations. International Journal of Difference Equations, 12 (2): 169-183.

- [113] Psarros N, Papaschinopoulos G and Schinas C J (2018) On the Stability of Some Systems of Exponential Difference Equations. Opuscula Mathematica, 38 (1): 95-115.
- [114] Koshy T (2001) Fibonacci and Lucas Numbers with Applications Vol. 5 John Wiley & Sons, New York, USA.
- [115] Mc Carty C P (1981) A formula for Tribonacci numbers, Fibonacci Quarterly, 19 (5): 391-393.
- [116] Horadam A F (1965) Basic Properties of Certain Generalized Sequence of Numbers. Fibonacci Quarterly, 3: 161-176.
- [117] Shannon A G and Horadam A F (1972). Some Properties of Third-Order Recurrence Relations. *The Fibonacci Quarterly*, 10 (2): 135-145.
- [118] Bacani J B and Rabago J F T (2015) On Generalized Fibonacci Numbers. Applied Mathematics Sciences, 9 (25): 3611-3622.
- [119] Trojovsky P (2016) On Some Combinations of terms of a Recurrence Sequence. Chaos, Solitons and Fractals, 82: 34-37.
- [120] Soykan Y (2019) Simson Identity of Generalized *m*-step Fibonacci Numbers. arXiv preprint, arXiv:1903.01313.
- [121] Soykan Y, Okumuş İ and Taşdemir E (2019). On Generalized Tribonacci Sedenions. arXiv preprint, arXiv:1901.05312.
- [122] Howard F T and Saidak F (2010) Zhou's Theory of Constructing Identities. In: Congress Numer. 200: 225-237.

CURRICULUM VITAE

PERSONAL INFORMATION

Name, Surname: İNCİ OKUMUŞ Nationality: Turkish (T.C.) Date of Birth: 08.01.1990 Date of Place: GEBZE E-mail: <u>inci_okumus_90@hotmail.com</u>

Address: Kazım Karabekir Mah. Vefa Sok. Okumuşlar Apt. No: 6/6 Darıca/ KOCAELİ Tel: (0 535) 613 73 85

EDUCATION

Degree	Institution	Year of Graduation
MSc	Bülent Ecevit University, Mathematics	2012-2014
BS	Bülent Ecevit University, Mathematics	2008-2012
High School	Darıca Yabancı Dil Ağırlıklı Lisesi	2003-2007

FOREIGN LANGUAGE

English

FIELD OF STUDY

Major Field: Difference and Functional Equations: (Stability and Asymptotics of Difference Equations; Oscillatory and Periodic Solutions, etc.)